

# This is to certify that the

dissertation entitled

THE EFFECT OF GENDER-ROLE ATTITUDES AND RAPE MYTH BELIEFS ON THE DECISION-MAKING PROCESS OF COLLEGE JUDICIAL BOARDS.

presented by

ROGER L. SERR

has been accepted towards fulfillment of the requirements for

Ph.D. degree in EDUCATION

Major professor

DR. LOUIS F. HEKHUIS

Date OCTOBER 6, 1993



LIBRARY Michigan State University

PLACE IN RETURN BOX to remove this checkout from your record. TO AVOID FINES return on or before date due.

DATE DUE	DATE DUE	DATE DUE
THE PARTY OF THE P		
- MAGIC 2		
MAR 1 5-1999		
JUL 8 6 2900		

MSU is An Affirmative Action/Equal Opportunity Institution

# THE EFFECT OF GENDER-ROLE ATTITUDES AND RAPE MYTH BELIEFS ON THE DECISION-MAKING PROCESS OF COLLEGE JUDICIAL BOARDS

By

Roger L. Serr

# A DISSERTATION

Submitted to
Michigan State University
in partial fulfillment of the requirements
for the degree of

DOCTOR OF PHILOSOPHY

Department of Educational Administration

1993

		;;
		;
		:
		3
		:
		:
		:

#### **ABSTRACT**

THE EFFECT OF GENDER-ROLE ATTITUDES AND RAPE MYTH BELIEFS ON THE DECISION-MAKING PROCESS OF COLLEGE JUDICIAL BOARDS

By

## Roger L. Serr

Heterosexual acquaintance rape is a social problem on college campuses. When reported, colleges often respond by adjudicating alleged perpetrators via institutional judicial processes. These processes often include the option of having judicial boards judge the case and render disciplinary sanctions. Critics charge that judicial boards are ill-prepared to adjudicate acquaintance rape cases because members are more influenced by rape myth beliefs and gender-role stereotypes than by the facts of the incident.

No studies were found that addressed the criticisms concerning judicial boards adjudicating acquaintance rape cases. Thus, the purpose of this study was to determine what effect judicial board members' rape myth belief and gender-role attitudes had on the adjudication of a campus rape case. This investigation focused on the influence of the aforementioned perceptions as they related to the determination of a verdict and subsequent sanctioning (when the assailant was found guilty).

The eighty-two subjects represented in this study were drawn from a sample of one-hundred-thirteen student judicial board members representing six independent and nine state-assisted colleges and universities located in Michigan. All the subjects served on primary judicial boards empowered to suspend or recommend suspension for students found guilty of violating conduct standards.

All the subjects completed scales measuring gender-role attitudes and rape myth beliefs. Additionally, one-half of the subjects judged an acquaintance rape scenario. The remaining forty-one subjects judged a stranger rape scenario. After reading the scenarios, subjects determined if the male perpetrator was guilty or innocent of rape.

Subjects also assigned a judicial sanction when they found the perpetrator guilty.

T-tests determined that significant differences existed between males and females related to rape myth belief and traditional gender-role attitudes. Females were more feminist and believed less in rape myths. The sample demonstrated a full spectrum of rape myth belief and gender-role attitudes. However, all subjects selected the guilty verdict regardless of the rape scenario judged, their social belief structure, or their gender. Subjects generally assigned harsher judicial sanctions as well, with only six people assigning sanctions less than suspension or expulsion from the institution.

This dissertation is dedicated to my best friend and wife,

Konnie Haugen Serr,

whose love and support made this journey possible.

#### **ACKNOWLEDGMENTS**

The completion of this dissertation is, in great part, due to the support, encouragement, and contributions of many people.

Dr. Louis Hekhuis, my faculty advisor, was very supportive, patient, and encouraging. Dr. Marvin Grandstaff provided insightful commentary, especially in the early stages of this research. Dr. Lee June was very supportive and also provided helpful feedback throughout the process. Dr. Linda Forrest, an excellent scholar, was supportive, provided excellent insight and feedback, and challenged me to scholarly excellence.

Dr. Philip Gardner was instrumental in the completion of this dissertation. He provided outstanding research guidance and also made many helpful suggestions concerning the conclusions and implications of this research. He also motivated my thinking and challenged me to scholarly excellence. Above all, he gave unselfishly of his time.

Dr. Louis Stamatakos also deserves special acknowledgement. He served as my advisor up until his retirement. I fondly remember the many long discussions we engaged in over the years. He truly is a credit to the academy and remains influential in my professional life.

My colleagues (Cindy Helman, Leonard Jones, Dr. Kate Murphy, Kathy Stepanovich, Ina Kimball, Charis Linklater, Shirleyetta Sernick, Dr. Nancy Lange, Dr. Rudy Altamirano) in the Department of Residence Life at Michigan State University were supportive throughout this process.

Karen Podolski deserves special acknowledgment for providing invaluable word-processing technical support. The professional/personal support and encouragement of Mary Haas, the Director of Residence Life, was a critical ingredient in the completion of this dissertation --- I am deeply grateful to her!

The people most deserving of acknowledgment are my family. Konnie, my wife, was very supportive and assumed a disproportionate share of the domestic duties and child-rearing responsibilities. Her sacrifices were substantial and words fail to capture the depth of my appreciation for all that she contributed to this journey.

Leah, Jared, and Ethan, each in their own way, were very understanding of my frequent absences and busy schedule. Their excitement about my completing the "book" motivated me. I, too, look forward to more family recreational time together, especially fishing. They are great kids!

# TABLE OF CONTENTS

LIST OF TA	BLES	×
CHAPTER 1	INTRODUCTION	1
A.	Background and Theoretical Foundation	1
В.	Statement of the Problem	11
c.	Purpose of the Study	12
D.	Need for the Study	14
E.	Research Questions	15
F.	Hypotheses	16
G.	Methodology	17
	1. Subjects	17
	2. Instrumentation	17
	3. Data Collection Procedures	19
	4. Scoring the Data	20
	5. Data Analysis	20
н.	Definition of Terms	21
	Limitations of the Study	23
I.	Organization of the Study	
J.	Organization of the Study	24
CHAPTER 2	LITERATURE REVIEW	26
λ.	Introduction	26
В.	Definitions	27
٥.	1. General	27
	2. Legal	29
	3. Social	30
c.		
C.	Theoretical Perspective	32
	1. Evolutionary Theory	33
	2. Social Learning Theory	33
_	3. Feminist Theory	33
D.	Rape Law	35
E.	Prevalence of Rape	39
	1. General	39
	2. Higher Education	42
F.	Influence of Alcohol and Other Drugs	50
G.	Gender-Role Perception	52
	1. Stereotypes	52
	2. Dating Relationships	54
	3. Rape Myth Acceptance	57
	4. Gender-Role Attitudes	61
	5. Likelihood of Raping	63
н.	Attribution of Responsibility	63
•••	1. Theoretical Overview	63
	2. Attributional Factors	65
		66
	a. Force and Resistance	
		67
	c. Respectability	67
	d. Acquaintanceship	68
_	3. Gender Differences	69
Į.	Jury Decision-Making	70
J.	Campus Disciplinary Procedures	71
<b>K</b> .	Summa xv	74

CHAPTER 3	METHODOLOGY	77
λ.	Introduction	77
	1. Statement of the Problem	77
	2. Purpose of the Study	77
	3. Research Questions	79
	4. Hypotheses	79
В.	Subjects	80
c.	Instrumentation	83
C.		83
		85
	3. Rape Scenarios	87
	4. Adjudication Questionnaire	88
	5. Demographic Information Form	89
D.	Data Collection Procedures	89
E.	Scoring the Data	91
F.	Data Analysis	92
- •		
CHAPTER 4	PRESENTATION AND ANALYSIS OF THE DATA	95
λ.	Introduction	95
•••	1. Subjects	95
	2. Instrumentation	96
	3. Summary	103
		104
В.	General Profile of the Sample	
		104
		105
		106
		106
	5. Academic Class Standing	108
	6. Academic Major	108
		109
		109
		109
		111
		111
		113
c.		113
		114
	2. ATR Questionnaire (Unified Factor)	115
	3. Hypothesis 1	117
	4. Hypothesis 2	
		118
	6. Hypothesis 4	
		110
	8. Hypothesis 6	118
	9. Hypothesis 7	119
	10. Hypothesis 8	119
	11. Hypothesis 9	119
		120
	13. Hypothesis 11	
D.	Descriptive Analysis	120
E.	Chapter Summary	
E.	Chapter Summary	123
CHAPTER 5	SUMMARY, DISCUSSION, AND RECOMMENDATIONS	125
•	Introduction	10-
A.		
В.	Summary	125
	1. Theoretical Foundation	
	2. Statement of the Problem	
	3. Purpose of the Study	127
	4. Need for the Study	129
	5. Research Questions	
	6. Research Methodology	
	a. Subjects	
	A. DUDIELLO AAAAAAAAAAAA	

	b. Instrumentation	131
	c. Research Design	132
	d. Data Analysis	133
c.	Major Findings	134
	1. Rape Myth Belief	134
	2. Gender-Role Attitudes	134
	3. Verdict	135
	4. Sanctions	135
	5. Hypotheses	137
	6. Hypothesis 1	137
	7. Hypothesis 2	137
	8. Hypothesis 3	138
	9. Hypothesis 4	138
	10. Hypothesis 5	138
	11. Hypothesis 6	138
	12. Hypothesis 7	138
	13. Hypothesis 8	139
	14. Hypothesis 9	139
	15. Hypothesis 10	139
	16. Hypothesis 11	139
D.		140
	<ol> <li>Limitations Associated With Adjudication</li> </ol>	
	Research	140
	2. Limitations Associated With Social	
	Phenomenon Research	140
	3. Limitations Associated With Judicial	
	Descriptors	141
	4. Limitations Associated With Case Studies	142
E	Generalizability of the Results	
F	F. Discussion of the Major Findings	143
	1. Pluralistic Ignorance	144
	2. Standard of Evidence	148
(	G. Conclusions	149
	H. Theoretical Implications	150
		151
	J. Recommendations for Further Research	
	K. Concluding Commentary	153
	DICES	
PPBNL		1
	B. Survey Instrument With Stranger Rape Scenario C. Survey Instructions	
		1/5
	D. Varimax Rotated Factor Structure of ATR Questionnaire	
	Anamitouugila	T \8
THR E	RTRI.TOCDADHY	101

# LIST OF TABLES

Table	<u>1</u>	Page
1.	Reliability Coefficients for the Revised ATR Factors	102
2.	Correlation Between the Revised ATR Factors	102
3.	Comparison of Rape Scenarios By Gender	105
4.	Mean Age of Respondents By Rape Scenarios	105
5.	Comparison of Rape Scenarios By Ethnicity	106
6.	Comparison of Rape Scenarios By Religious Practice	107
7.	Comparison of Rape Scenarios By Religious Devoutness	108
8.	Comparison of Rape Scenarios By Academic Class	108
9.	Number Hearings Participated In By Rape Scenarios	110
10.	Participation in Rape Hearings By Rape Scenarios	111
11.	. Received Training About Rape By Rape Scenario	111
12	. Knowledge of Rape Victims By Rape Scenarios	112
13	. Relationship With Rape Victim By Rape Scenario	112
14	Judicial Verdict By Rape Scenario	114
1	5. Judicial Sanctions By Rape Scenario	114
1	6. Significance Test of FEM and Gender	115
1	17. Significance Test of FEM and Knowing a Rape Victim	116
	18. Significance Test of FEM and Ethnicity	116
	19. Significance Test of FEM and Rape Hearing Participation	116
	20. Significance Test of ATR Unified Factor and Gender	117
	21. Significance Test of ATR Unified Factor and Victim Knowledge	117

22.	Correlation Between FEM Scale and ATR Factors	120
23.	Gender Influence On FEM Scale	121
24.	Gender Influence On Unified Factor	121
25.	Ethnic Influence On FEM Scale	122
	Varimax Rotated Factor Structure of ATR Questionnaire	178

#### CHAPTER ONE

#### INTRODUCTION

# Background and Theoretical Foundation

Approximately 80 to 90 percent of female rape victims are assaulted by acquaintances (Ageton, 1983; Aizenman and Kelley, 1988; Bailey, 1986; Copenhaver and Grauerholz, 1991; Koss, 1988; National Victim Center, 1992), yet many people still cling to outmoded definitions of rape. For example, the classic definition of rape involves a stranger violently attacking a woman with a weapon outside at night (Burt, 1980). Only recently have people begun to recognize acquaintance rape as a very real phenomenon in American society, especially among women ages 16 to 24 (Bureau of Justice Statistics, 1985b, 1986, 1987, 1991a; Harlow, 1991).

This phenomenon is especially apparent on college campuses where researchers have established prevalence rates by using survey techniques designed to overcome the subjective bias associated with rape.

Recognizing that rape is one of the most underreported crimes (Federal Bureau of Investigation, 1984), researchers adopted the survey approach used by early victimization studies (Bureau of Justice Statistics, 1984; Ross, Gidycz, and Wisniewski, 1987; Russell, 1982) to better estimate the occurrence of rape. Secondly, researchers used questionnaires that asked respondents to describe actual behaviors, that were then compared to criterion definitions of rape and sexual assault to establish the levels of sexual aggression occurring (Burkhart and Stanton, 1988; Koss and Leonard, 1984).

Kirkpatrick and Kanin (1957) conducted a pioneering study involving 291 female college students that found 21 percent of the women had been offended by a male's forceful attempts at intercourse, and 6 percent were offended by aggressively forceful attempts. Kanin and Parcell (1977) replicated and extended Kirkpatrick and Kanin's (1957) study, finding that 51 percent of the 282 female respondents were offended at some level of erotic intimacy during the academic year. Approximately 32 percent of the females who experienced male sexual aggression, where

intercourse was the goal, reported that rape had occurred.

Bailey (1986) reported that nearly 8 percent of 362 female college students surveyed had been raped, and 17 percent were victims of attempted rape. Approximately 93 percent of these rapes were committed by acquaintances. A sample involving 447 college females found that 15 percent had been raped and 18 percent had experienced an attempted rape while on a date (Wilson and Durrenberger, 1982). Twenty-one percent of 247 female college students sampled reported that they had been physically forced by a dating partner to have sexual intercourse (Sandberg, Jackson, and Petretic-Jackson, 1987).

At a large Midwestern university, Copenhaver and Grauerholz (1991) found that almost 25 percent of the 140 females sampled were victims of an attempted rape, and 17 percent of the women had experienced an act of rape. Ninety-five percent of these women were acquainted with their attacker. Miller and Marshall (1987) determined that 27 percent of 323 women had been victims of a date rape while attending college. Based on the results of a deviance survey administered to 563 female college students over a 4-year period, Eskridge (1989) discovered that 21 percent of the women had been raped. From a sample of 204 college females, Aizenman and Kelley (1988) found that 22 percent had been raped by an acquaintance.

Koss and Oros (1982) determined from a representative sample of 2,016 female college students that 6 percent had been raped, 6 percent had been forced to participate in anal or oral intercourse because physical force or threats had been used, and 8 percent had sexual intercourse because the man used some degree of physical force. Koss, Gidycz, and Wisniewski (1987) conducted a national survey of 32 diverse institutions of higher education involving 3,187 women, finding that 12 percent of the females had experienced attempted rape and 15 percent had been raped.

An examination of surveys has determined men's level of involvement in acquaintance rape situations. For example, Eskridge (1989) found

that 7 percent of 836 male college students had forced a female to have sexual intercourse against her will at least once. Ross, et al. (1987) discovered that 4 percent of 2,972 male students had engaged in rape. Six percent of 300 males reported forcing a female date to engage in sexual activity in the prior year (Yegidis, 1986). Kanin (1967) determined that 26 percent of 341 male undergraduate students used aggressive tactics to obtain intercourse while at college. Twenty percent of 175 (Mosher and Anderson, 1986) and 15 percent of 201 (Rapaport and Burkhart (1984) male college students admitted using force to have sexual intercourse with a female.

The use of alcohol and other drugs has been associated with the occurrence of sexual assaults (Cage, 1992). Sixty-six percent of 71 self-disclosed date rapists implicated excessive alcohol consumption as a causal factor in the rape (Kanin, 1984). Mosher and Anderson (1986) found that 19 percent of their sample of male college students had gotten drunk and forced a woman to have sexual intercourse, and 66 percent had used alcohol to get a woman drunk to have sex with her.

Koss (1988) in a national sample of over 6,100 college students, found that 73 percent of the offenders and 55 percent of the victims admitted using drugs or alcohol at the time of the sexual victimization.

Research involving 155 sexually victimized female college students found that 65 percent of their offenders were moderately to very intoxicated at the time of the assault (Amick and Calhoun, 1987).

Ward, Chapman, Cohn, White, and Williams (1991) noted that "a wide range of estimates of the incidence of sexual assault is found in the literature" (p. 65). They state "this may be due in part to varying methodologies and in part to variations in the definition of the behavior covered by the estimates" (p. 65). Ellis (1989) agreed, writing that the "...estimates are far too varied to be dismissed as resulting from sampling error ... [thus] they appear to reflect major differences in criteria used to identify rape" (p. 4). Ward et al. concluded that rape "... is an important social problem on college

).
.

er :

.

campuses, even if one uses the 'low' ... " (p. 65) estimates of sexual victimization found by researchers.

Ellis (1989) identified 3 contemporary theories to help explain the cause of acquaintance rape: evolutionary theory, social learning theory, and feminist theory. Evolutionary theory emphasizes natural selection processes where conflict occurs between the male biological drive and the female desire to control reproduction. Social learning theory, while very similar to feminist theory, focuses more on cultural mores than on political and socioeconomic exploitation.

Feminist theory disagrees with the traditional psychiatric view that rapists are sexual deviants (Check and Malamuth, 1985). Instead, feminist theory "... considers rape to be the result of long and deep-rooted social traditions in which males have dominated nearly all important political and economic activities" (Ellis, 1989, p. 10). This results in an inferior sociopolitical status for women who gain worth and recognition only in relation to men (Morrison, 1980). Thus, "rape is a ritual of power ... [and] ... the symbolic enactment of social and cultural attitudes" (Metzger, 1976, p. 405).

Feminist writers (Clark and Lewis, 1977; Griffin, 1975; MacKinnon, 1983; Morrison, 1980; Sanday, 1981; Weis and Borges, 1973) stated that we live in a "rape culture" where rape is viewed as natural because of men's superior position in society. This theory stated that rape is "... a pseudosexual act, a pattern of sexual behavior that is concerned much more with status, hostility, control, and dominance than with sexual pleasure or sexual satisfaction" (Groth and Birnbaum, 1980, p. 21). Sanday (1981) studied rape cross-culturally and concluded that "rape is not an integral part of male nature, but the means by which men programmed [by the culture] for violence express their sexual selves" (pp. 25-26).

A visible symbol of the "rape culture" are rape laws that have been heavily criticized as representative of the sexism that exists in American society (Brownmiller, 1975; Clark and Lewis, 1977; Estrich,

1987; Griffin, 1979; MacKinnon, 1983; Medea and Thompson, 1974; Russell, 1975; Wood, 1973). Many of these laws were based on common law that "... is judge-made law rather than law that originates from constitutions or from legislatures or administrative agencies" (Kaplin, 1989, p. 13). The common law presumed that "unchaste" women are more likely to consent to sexual intercourse and to lie about the event than "chaste" women, that women invite sexual attacks via their behavior, and that a rape victim's credibility must be substantiated to ensure she is telling the truth (Harris, 1976; Tong, 1984). In essence, rape laws act "... to punish men who do not rape appropriately" (Reynolds, 1974, p. 63).

Social-political movements, including the passage of the equal rights amendment and crime control reform efforts, aided the passage of rape reform laws by many state legislatures in the 1970s (Loh, 1981).

Most notable was the Michigan Criminal Sexual Conduct Statute, adopted in 1974, that was considered a model rape law (Bessmer, 1984; Caringella-MacDonald, 1985; Cobb and Schauer, 1974; Praeger Publishers, 1977; Virginia Law Review, 1975). Although criticized for not being totally sufficient (Cobb and Schauer, 1974; Estrich, 1987; Marsh, Geist, and Caplan, 1982), "the Michigan Sexual Assault Act reflects a major rethinking of the common assumptions about rape" (Cobb and Schauer, 1974, p. 235).

The Michigan Statute was considered a model because it codified definitions (Cobb and Schauer, 1974), used sex-neutral terminology (Caringella-MacDonald, 1985; Cobb and Schauer, 1974; Virginia Law Review, 1975), and incorporated 4 degrees of criminal sexual contact (Cobb and Schauer, 1974). The degrees were distinguished by variables related to the occurrence of sexual contact versus sexual penetration and to the amount of force and coercion used. The model law also does not require the victim to resist the attacker, to have testimony corroborated, to prove nonconsent, or to reveal past sexual conduct except in severely limited circumstances. The law presumes nonconsent

when the victim is underage or physically or mentally helpless (Cobb and Schauer, 1974).

Rape laws define the criteria used to judge guilt and innocence in acquaintance rape situations. While laws attempt to provide objective criteria, attribution theory postulates that judgments concerning blame and responsibility are influenced by the dispositional qualities of an actor and by environmental factors associated with the situation. Thus, objectivity is suspect because both first-hand observation and secondary reports concerning an event are subject to error given the complexity of the social stimuli surrounding the event (Shaver, 1975).

For example, Heider (1958) theorized that people make attributions based on a combination of ability, exertion, and task difficulty that represents an action sequence used to determine whether behavior was the result of dispositional or environmental causes. Jones and Davis (1965) speculated that actors make choices about courses of action, allowing a perceiver to diagnose an actor's intentions. Kelley (1973) stated that attributions are made according to a 3-dimensional model that looks at the context, persons, and stimulus associated with an event.

Additionally, Walster (1966, 1967) theorized that people have a need to live in an orderly world that is predictable and controllable. As such, they assign causal responsibility for an incident to something other than chance, believing that they will avert a similar situation from happening to them. Shaver (1970) suggests that a defensive attribution process operates where less responsibility is assigned to people perceived as similar to the perceiver, with conversely more responsibility assigned to circumstances. Lerner (1965, 1970, 1980) postulated that people believe in a "just world" where positive outcomes happen to good people and negative outcomes happen to bad people.

Attribution theory is not concerned with the true cause of an event, rather it focuses on understanding how people attribute causation. When judging incidents, people are influenced by cultural assumptions or social stereotypes that obscure the objective facts of a

p:::13

10724

**12.**10

:9**8**£)

Schwe

r.e:

crea

250

:=6

111

r.e

ķ

**4**.

·c.

:

. :

situation (Jones and Davis, 1965). This is especially true in acquaintance rape situations where people rely on biased or inaccurate beliefs or assumptions about the nature of rape (Hueston and Burkhart, 1986). Commonly called rape myths (Burt, 1980; Pineau, 1989; Schwendinger and Schwendinger, 1974), these beliefs rely on false stereotypes concerning the true dynamics of acquaintance rape, often creating a climate hostile to the victim (Burt, 1980).

Research conducted by Wilson, Faison, and Britton (1983) involving 250 college students suggests that the level of belief in rape myths influences the definitional boundaries between rape and normal sexual intercourse. As a result, when a situation is not congruent with rape stereotypes people look for other interpretations, often blaming the women for her own rape (Klemmack and Klemmack, 1976). For example, Krulewitz and Payne (1978) found that subjects generally blamed the male assailant more and attributed less responsibility to the female victim as the level of force used by the male increased. Rape victims were found to have more credibility as the level of violence used by the male increased (LaFree, 1980) and when they physically resisted the attack (Howard, 1984a, 1984b; Krulewitz and Nash, 1979; Shotland and Goodstein, 1983).

A victim's physical attractiveness also influences attribution of responsibility. Physically attractive women were perceived as more sexually alluring (Dermer and Thiel, 1975), promiscuous (Thornton, 1977), and naive to the affect they have on men (Mazelan, 1980). Women perceived as physically unattractive were believed to provoke their victimization more than attractive women (Seligman, Brickman, and Koulack, 1977; Tieger, 1981).

Victims perceived as more respectable according to characteristics such as occupation, marital status, and sexual history received less attribution of blame for their victimization than less respectable victims (Cann, Calhoun, and Selby, 1979; Feldman-Summers and Lindner, 1976; Kanekar and Kolsawalla, 1977; Kerr and Kurtz, 1977; McGraw, 1987;

Smith, Keating, Hester, and Mitchell, 1976). McGraw (1987) found that people attributed significantly greater responsibility to the rapist when the assailant and victim were strangers. Acquaintance rape situations are often viewed less seriously than stranger rape cases (Bohmer, 1974; Chappell and Singer, 1977; Kerstetter, 1990; LaFree, 1980; Wilson, 1978).

Thornton, Ryckman, and Robbins (1982) found that gender and genderrole attitudes were the 2 primary contributors concerning the amount of perceived victim culpability. Males, as compared to females, viewed rape less seriously (Tieger, 1981), perceived it more as a sexual act (Barnett and Feild, 1977; Bridges and McGrail, 1989), emphasized the personal characteristics of rape victims (Calhoun, Selby, and Warring, 1976), were less punitive toward the assailant (Feldman-Summers and Lindner, 1976; Scroggs, 1976; Smith, Keating, Hester, and Mitchell, 1976; Stephan, 1974; Tolor, 1978), and believed that women precipitate their own rape (Cann, Calhoun, and Selby, 1979; Kanekar and Vaz, 1983; Selby, Calhoun, and Brock, 1977; Smith, et al, 1976). Males, especially sexually aggressive men, also had greater belief in general rape myths and held more traditional gender-role attitudes than females (Barnett and Feild, 1977; Dull and Giacopassi, 1987; Garrett-Gooding and Senter, 1987; Horsley, 1988; Jenkins and Dambrot, 1987; Koss and Dinero, 1988; Ross, Leonard, Breezley, and Oros, 1985; McDonald, 1988).

Williams (1979) found significant support for a link between gender-role attitudes and assessment of female blame for rape. This conclusion has also been supported by the work of Acock and Ireland (1983), Feild (1978a, 1978b), Feild and Bienen (1980), Howard (1984a), Klemmack and Klemmack (1976), and Shotland and Goodstein (1983). For example, Howard (1984a) found that people with traditional gender-role attitudes attributed more global and characterological blame to female assault victims than to male assault victims. Furthermore, those with more egalitarian gender-role attitudes tended to not differ in their attributions between male and female victims.

; •• •• • . . 4. Ç .

14

Additionally, gender-role ideology and general belief in rape myths have been found to be related (Briere and Malamuth, 1983; Bunting and Reeves, 1983; Check and Malamuth, 1983; Coller and Resick, 1987; Costin, 1985; Fisher, 1986; Kanin, 1984; Malamuth, 1981; Malamuth and Check, 1980; Muehlenhard and Hollabaugh, 1988; Quackenbush, 1987; Seiverd, 1989; Tieger, 1981). Specifically, Check and Malamuth (1983) found that individuals with more stereotyped gender-role beliefs perceived a rape victim as reacting more favorably to her rape than did people subscribing to low gender-role stereotypes. An important finding was that no gender differences emerged from this study, indicating that both males and females judged acquaintance rape victims from the perspective of their gender-role beliefs.

People with traditional gender-role orientations were more likely to believe that forced sexual intercourse was normal (Check and Malamuth, 1985; Fisher, 1986; Goodchilds, Zellman, Johnson, and Giarrusso, 1988; Muehlenhard, 1988; Muehlenhard, Friedman, and Thomas, 1985; Muehlenhard and Hollabaugh, 1988; Sandberg, Jackson, and Petretic-Jackson, 1987). Muehlenhard (1988) consistently found that rape justification ratings, across a variety of dating scenarios, were influenced by a person's level of gender-role traditionality.

Generally, traditional persons thought rape was more justifiable than did nontraditional persons.

Langer, Blank, and Chanowitz (1978) demonstrated that people ignore relevant information when a situation is consistent with their prior expectations. As such, juries often ignore the strict application of the law and respond to the circumstances of each case according to the general stereotypes of society (Brooks and Doob, 1975; DiPerna, 1984; Feild and Bienen, 1980; Frazier and Borgida, 1988; Guinther, 1988; Kalven and Zeisel, 1966; L'Armand and Pepitone, 1982; Williams, 1976). The result may be an arbitrary hearing (Griffitt and Jackson, 1973; Frank, 1949; Simon, 1980; Stephan, 1975), where the victim's gender-role behavior influenced the verdict more than the evidence of the rape case

(LaFree, Reskin, and Visher, 1985).

As has been previously documented, (a) acquaintance rape is a frequent occurrence on college campuses; (b) gender-role ideology and rape myth beliefs influence the definitional boundaries of rape; and (c) various attributional factors influence judgements of blame and responsibility in acquaintance rape situations, including those of criminal juries. It is these aforementioned points that cause critics (Parrot, 1991a, 1991b) to question the effectiveness of college judicial systems to properly adjudicate acquaintance rape cases.

Acquaintance rape incidents that occur on college campuses can be adjudicated in both the criminal court and college disciplinary systems (Adams and Abarbanel, 1988; Parrot, 1991b). While criminal proceedings utilize the "beyond a reasonable doubt" standard, campus disciplinary systems use the less stringent "by a preponderance" or "clear and convincing" standards of proof to judge violations of college rules (Long, 1985). College proceedings are generally less formal and often take place in front of faculty and student representatives who serve as regular members of a judicial hearing board (Lancaster, Cooper, and Harmen, 1993). The judicial board listens to both sides of the case, decides if campus rules have been violated, and determines a penalty for the guilty party (Adams and Abarbanel, 1988).

stoner and Cerminara (1990) stated that college judicial boards are an important option for judging students who violate campus rules and regulations. However, most college judicial boards are only prepared to adjudicate standard cases like vandalism, alcohol possession, and plagiarism (Collison, 1991; Julian, 1992; Kalette, 1988). As a result, complex cases like acquaintance rape are often poorly adjudicated because judicial board members don't receive proper training about: (a) the laws and policies related to acquaintance rape; (b) the psychological needs of victims and defendants; and (c) the typical patterns associated with acquaintance rapes (Parrot, 1991a).

In fact, research conducted by Dannells (1991) concluded that

a. **~** , . . . ŧ t

judicial affairs professionals have made few changes in the last 10 years in the methods used to handle student misconduct cases. Thus, in acquaintance rape cases, Parrot (1991b) stated that judicial boards should have a legal expert appointed as the chairperson to ensure that the proceeding are properly conducted consistent with state laws and institutional policies. Furthermore, Parrot (1991b) stated that assailants, if found guilty of acquaintance rape, must be dealt with to the fullest extent permitted by campus policy. This was consistent with recent legal developments in higher education law that suggest colleges must thoroughly investigate and respond to acquaintance rape incidents (Associated Press, 1991; Batson, 1986; Freeman, 1990; Gibbs, 1991; Helms, 1987; Pavela, 1992b; Steiner, 1989; Wagner, 1991).

## Statement of the Problem

Heterosexual acquaintance rape is a social problem on college campuses, with approximately 27 percent of college women experiencing rape while on a date (Miller and Marshall, 1987). When reported, colleges often respond by adjudicating alleged perpetrators in acquaintance rape cases via institutional student judicial processes. These processes often include the option of having peer judicial boards hear the facts of the case, determine a verdict, and render discipline sanctions when applicable.

Critics charge that judicial boards are ill-prepared to adjudicate acquaintance rape cases because of inadequate training and lack of sensitivity concerning the dynamics of acquaintance rape situations.

Furthermore, research involving both actual and mock juries demonstrated that verdicts in acquaintance rape cases were often influenced more by the individual member's levels of belief in rape myths and gender-role stereotypes than by actual facts.

### Purpose of the Study

The review of the literature clearly documented that college students, both male and female, believed in traditional gender-role stereotypes (Rosenkrantz, et al., 1968) that influenced dating relationships (Muchlenhard, Friedman, and Thomas, 1985) and perceptions concerning a willingness to engage in sexual intercourse (Bostwick, 1990). College students were also found to believe in rape myths (Check and Malamuth, 1985) that influenced the definitional boundaries between rape and normal sexual activity (Wilson, Faison, and Britton, 1983). Furthermore, rape myths were found to be related to views concerning the roles and rights of women in society (Costin, 1985), with students holding traditional views more likely to believe in rape myths and to accept sexual aggression (Greendlinger and Byrne, 1987). Men with strong gender-role stereotyping beliefs reported a greater belief in rape myths and more likelihood of using aggression in sexual situations (Malamuth, 1981).

A review of the literature on attribution demonstrated that people defined a situation as rape according to their stereotypes of rape (Klemmack and Klemmack, 1976). For example, rape was more often defined when the woman resisted (Krulewitz and Nash, 1979) and when she was not acquainted with the rapist (McGraw, 1987). Further research determined that gender and gender-role attitudes were the 2 primary contributors concerning the definition of rape (Thornton, Ryckman, and Robbins, 1982), with males viewing the rape less seriously than females (Calhoun, Selby, and Warring, 1976; Tieger, 1981).

The literature clearly and consistently identified the prevalence of rape myth beliefs and gender-role attitudes in the general college population. Also defined was the influence of these beliefs and attitudes on attributions of blame and responsibility in acquaintance rape situations. Of special note was the consistent results of the studies, over time, from a variety of colleges and universities.

Research involving both actual and mock juries demonstrated that verdicts in acquaintance rape cases were often influenced more by the individual member's level of belief in rape myths and gender-role attitudes than by actual facts (LaFree, Reskin, and Visher, 1985). In essence, jury members reflected the stereotypes of society. This was especially relevant when related to the effectiveness of peer judicial boards adjudicating acquaintance rape cases on college campuses. Specifically, given the documented level of rape myth belief and traditional gender-role attitudes in the general college population, were college judicial board members more aware of the influence of rape myths and gender-role attitudes on their judgement of blame and responsibility in rape cases?

This was an important question given that judicial board members are trained to impartially judge all cases heard before them. If individual members subscribe to the same beliefs as their peers in terms of rape myth beliefs, Parrot (1991a, 1991b) and other critics may be correct in questioning the competence of judicial boards to adjudicate acquaintance rape cases on campus. Thus, this study was designed to answer the aforementioned question. Specifically, the purpose of this study was to determine what effect college judicial board members' levels of rape myth belief and gender-role attitudes had on the judgement of a male perpetrator's guilt or innocence for committing an act of rape. The rape was depicted by two written scenarios featuring an acquaintance assault and a stranger assault. The study specifically examined the differences between the acquaintance rape and stranger rape situations consistent with the purpose of this study.

Each of the two rape scenarios were judged by 50 percent of the subjects. Thus, 50 percent of the judicial board members judged the acquaintance rape scenario. The remaining 50 percent from the sample judged the stranger rape scenario. All subjects completed the FEM Scale that measured attitudes toward feminism and sexism, the Attitudes Toward Rape Questionnaire, the Adjudication Questionnaire, and the Demographic

•

•

•

**F** 

r •,

• .

. [

•-

3.

Information Form.

Subjects were enrolled students selected from 15 Michigan colleges and universities, representing 6 independent and 9 state-assisted institutions. All subjects were members of judicial boards empowered by an institution to levy or recommend disciplinary sanctions, including suspension, against other students found guilty, after a hearing, of violating conduct standards.

#### Need for the Study

Pavela (1992a) stated that colleges have a responsibility to resolve acquaintance rape cases based on their obligation to: (a) promote the moral development of students; and (b) protect the safety of people using their premises. Furthermore, Pavela stated that college judicial boards play an important role in fulfilling the college's obligation to adjudicate acquaintance rape cases. Pavela also stated that college judicial boards are composed of qualified individuals, that when guided by a competent advisor are capable of rendering sound judgements in even the most complex cases.

Critics, such as Parrot (1991a, 1991b), however question the competence of judicial boards to adjudicate acquaintance rape cases. They question whether judicial boards are adequately trained and advised about acquaintance rape situations to make sound judgements. This question is especially important when one looks at attribution research (Howard, 1984a, 1984b; LaFree, 1980; Smith, et al, 1976; Wilson, et al, 1983) that indicated acquaintance rape cases are often decided on extralegal factors based on rape myth belief and gender-role attitudes. Furthermore, much of the attribution research was conducted with college students, thus providing a baseline for inference about how college students attribute blame and responsibility in acquaintance rape situations.

Even given the current debate concerning the effectiveness of



judicial boards, this author was unable to find published research that determined how actual college judicial boards adjudicate acquaintance rape cases. No research was found that measured the level of rape myth belief or the gender-role attitudes of actual judicial board members. Finally, no research was discovered regarding the impact of attitudes and beliefs on either the verdicts or discipline sanctions levied in rape cases.

The literature that has been published on this subject generally critiques college judicial programs and/or provided recommendations for policy implementation (Adams and Abarbanel, 1988; Parrot, 1991a, 1991b; Pavela, 1992a). Thus, this research appears to substantially contribute to the literature by providing empirical data concerning the effect of gender-role attitudes and rape myth beliefs on the adjudication of a hypothetical rape case. Depending on the outcome, this research could contribute to the effective advising, selection, and training of judicial board members.

#### Research Questions

This investigation was guided by the following questions:

- What relationship, if any, exists between the verdicts of guilt or innocence for the male perpetrator and rape myth beliefs across acquaintance rape and stranger rape modes?
- 2. What relationship, if any, exists between the verdicts of guilt or innocence for the male perpetrator and gender-role attitudes across acquaintance rape and stranger rape modes?
- 3. What relationship, if any, exists between gender-role attitudes and rape myth belief on verdicts of guilt or innocence for the male perpetrator across acquaintance rape and stranger rape modes?

#### Hypotheses

The following directional hypotheses were tested in this study:

- As a group, subjects with high level belief in rape myths will most frequently select the not guilty verdict in the acquaintance rape mode.
- As a group, subjects with low level belief in rape myths will most frequently select the guilty verdict in the acquaintance rape mode.
- 3. As a group, subjects with high level belief in rape myths will most frequently select the guilty verdict in the stranger rape mode.
- 4. As a group, subjects with low level belief in rape myths will most frequently select the guilty verdict in the stranger rape mode.
- 5. As a group, subjects with sexist beliefs will most frequently select the not guilty verdict in the acquaintance rape mode.
- 6. As a group, subjects with feminist beliefs will most frequently select the guilty verdict in the acquaintance rape mode.
- 7. As a group, subjects with sexist beliefs will most frequently select the guilty verdict in the stranger rape mode.
- 8. As a group, subjects with feminist beliefs will most frequently select the guilty verdict in the stranger rape mode.
- Overall, there will be a positive relationship between genderrole attitudes and rape myth belief.
- 10. Overall, gender-role attitudes will be more influential than rape myth belief in determining the verdict in the acquaintance rape mode.
- 11. Overall, gender-role attitudes will be more influential than rape myth belief in determining the verdict in the stranger rape mode.

.

,

¥\*

,

,

•

.

# Methodology

#### Subjects

The subjects of this investigation included 113 judicial board members enrolled as undergraduates for the spring semester of 1993 at 6 independent and 9 state-assisted colleges within the state of Michigan. Eighty-five judicial board members were selected from the state-assisted colleges, with the remaining 28 board members representing the independent institutions. All subjects served on campus judicial boards empowered to suspend or recommend suspension for students found guilty of violating college conduct standards.

#### Instrumentation

The instrument used in this study was comprised of 5 components. Two of the components, the FEM Scale (Smith, Ferree, and Miller, 1975) and the Attitudes Toward Rape Questionnaire (Feild and Bienen, 1980) were used as originally developed. The Acquaintance Rape and Stranger Rape Scenarios were modified, based on the expert advise of the Sexual Assault Safety Program Coordinator at Michigan State University, from the original versions developed by Quackenbush (1987). The Adjudication Questionnaire and the Demographic Information Form were developed by this investigator (See Chapter 3 for additional information on the instrumentation).

The FEM Scale is a 20-item scale measuring attitudes toward feminism and traditional gender-role norms. The scale may be more generally conceived of as a measure of sexist attitudes toward women. Respondents answered the questions by using a 5-point Likert scale ranging from strongly agree to strongly disagree. Independent validation of the scale by Singleton and Christiansen (1977, p. 301) concluded that "... it is the shortest, most reliable measure of sexist or feminist attitudes yet available..." (See Appendix A or B, Section D, for the complete instrument).

.•

**--**'

L.

٠.,

•

\*i

.

;

1

Ĺ

. .

31.

1.

• :

..

•

E

•

. • . .

The Attitudes Toward Rape Questionnaire is a 32-item questionnaire composed of 8 latent factors that measures people's multidimensional attitudes about rape. The questionnaire used both positively and negatively phrased questions that respondents answered by using a 6-point Likert scale ranging from strongly agree to strongly disagree. Originally Factor 1, titled "Women's Responsibility In Rape Prevention", was to be used as a measure of rape myth belief. Koss and Leonard (1984, p. 220) stated that "... this factor could be interpreted as a general rape myth factor" (See Appendix A or B, Section E, for the complete ATR Questionnaire).

However, subsequent investigation prompted this researcher to conduct a new principal components factor analysis with varimax rotations (Babbie, 1990) for the ATR Questionnaire. The results identified 6 latent factors. As a result, a new Unified Rape Myth Pactor was created for use as the measure of rape myth belief specific to this investigation (See Appendix D for factor presentation; Chapter 4 also provides more detailed information on the creation of the Unified Factor).

The Acquaintance Rape and Stranger Rape Scenarios provided identical demographic and situational information through the first 7 paragraphs. The remaining paragraphs of the scenarios were identical in terms of the female's response and the male's actions, the only difference being the stranger or acquainted condition of the alleged rapist. Both scenarios met the legal requirements within the State of Michigan concerning the commission of rape by the male (See Appendix A, Section B, for the acquaintance rape scenario. See Appendix B, Section B, for the stranger rape scenario).

The Adjudication Questionnairs was utilized to determine whether the respondents judged the alleged male rapist as guilty or innocent of rape. When respondents judged the alleged rapist as guilty, the questionnaire also instructed them to select a disciplinary sanction that most closely approximated their view of appropriate punishment for

_		
s		
. •		
•		
• •		
.•		
•		
•		
• •		
•		
<del>-</del> .		
•		
•		
u*		

the offense. The descriptors explaining the disciplinary sanctions were closely matched with those utilized at 5 of the colleges in the study and in Stoner and Cerminara's (1990) model code (See Appendix A or B, Section C).

The Demographic Information Form collected personal data (e.g. gender, race, type of college attended, etc.) from each subject.

Additional questions were asked that discerned each subject's level of exposure to the topic of rape (See Appendix A or B, Section F).

# Data Collection Procedures

This investigation utilized a descriptive survey research design (Babbie, 1990; Turney and Robb, 1971) that determines the distribution of certain characteristics, traits, attributes, and/or opinions within a sample at a given point in time. The sample for this investigation was students that served on judicial boards empowered to suspend or recommend suspension for students found guilty of violating college rules.

A student affairs staff member with judicial responsibilities served as the contact person at each of the participating institutions. The contact person was responsible for administering the survey to the appropriate judicial board members. The subjects' names or addresses were not known by this researcher.

Each contact person was directly mailed in March 1993 the requisite number of survey booklets (See Appendix A and B), along with a letter explaining the instructions for administering the surveys (See Appendix C). Each survey booklet contained the same cover page explaining the purpose of the survey. Section A of the booklet contained the general instructions for completing the survey. The remaining sections contained the rape scenarios (one-half of the booklets contained the stranger scenario, one-half contained the acquaintance scenario), the Adjudication Questionnaire, the FEM Scale, the Attitudes Toward Rape Questionnaire, and the Demographic Information Form.

# Scoring the Data

This study utilized 1 instrument comprised of 5 components. The Demographic Information Form and Adjudication Questionnaire provided nominal data that were used as independent variables, requiring only classification. The Rape Scenarios required no scoring.

The FEM Scale was a 20-item instrument designed to measure sexist or feminist attitudes. No sub-factors were identified in the scale. The instrument utilized a 5-point Likert Scale, scored according to a range from strongly agree (1) to strongly disagree (5). Questions 15, 16, 19, and 20 were scored in reverse order, where strongly agree = 5 and strongly disagree = 1. A high score was consistent with feminist or less traditional gender-role attitudes.

The Attitudes Toward Rape Questionnaire was composed of 8 latent factors. The instrument used a 6-point Likert Scale where 1 = strongly agree and 6 = strongly disagree. As originally developed by Feild (1978a), a high score was indicative of substantial rape myth belief, as well as possessing anti-victim, pro-rape sentiments.

Subsequent modification of the ATR Questionnaire, because a new principal components factor analysis with varimax rotations (Babbie, 1990) was conducted, changed the scoring structure of the ATR Questionnaire. Specifically, items 2, 3, 20, and 22 were recoded so that strongly agree = 6 and strongly disagree = 1. These coding changes altered the direction of the scale. For the purpose of this investigation, a high score on the ATR Unified Rape Myth Factor was suggestive of low rape myth belief (See Chapter 4 for more specific information on the scoring changes associated with this instrument).

# Data Analysis

All respondents completing and returning the survey booklet were grouped according to whether their survey booklet included the stranger rape or acquaintance rape. The groups were initially compared on the basis of discrete demographic characteristics using chi-square analysis.

••

÷

·
. .

• •

· ·

The groups were also compared according to age, academic grade point average, and months of judicial experience by using a t-test.

ANOVA was originally selected to determine what relationship, if any, existed between the judicial verdict, rape mode, and rape myth belief or gender-role attitudes. MANOVA was to be used to determine what relationship, if any, existed between gender-role attitudes and rape myth belief on the judicial verdict. MANOVA was also to be used to determine what effect, if any, demographic variables had on the relationships between judicial verdict, rape mode, rape myth belief, and gender-role attitudes.

Regression was originally selected to determine the relationship, if any, between gender-role attitudes and rape myth belief. Multiple regression procedures were to be utilized to gain greater understanding concerning what variables had the greatest impact on the judicial verdicts.

The aforementioned procedures were selected with the idea that the subjects participating in this investigation would choose both the guilty and not guilty verdicts. However, the unexpected finding that all subjects judged the male to be guilty of rape limited the statistical analysis to t-tests, chi-squares, and correlation coefficients.

A level of significance of (p) .05 (Borg and Gall, 1983) was established to determine whether a hypothesis was tenable or must be rejected as false.

# <u>Definition of Terms</u>

Acquaintance rape: A type of rape that involves parties who knew each other prior to the assault. The level of acquaintanceship may range from slight recognition to knowing each other very well, including dating relationships involving some level of romantic interest.

٠, • • • <u>College</u>: Any postsecondary institution of higher education offering courses of study leading to at least a bachelor's degree.

<u>Conduct Standards</u>: The rules and regulations of the college that govern the behavior of enrolled students.

<u>Discipline Sanction</u>: The punishment or penalty given to an enrolled student found guilty by authorized agents of the college, including judicial boards, of violating conduct standards.

<u>Gender Role</u>: A pattern of behaviors that society both expects and encourages. Gender roles are based on widely shared cultural beliefs about how males and females should behave.

<u>Gender-Role Attitudes</u>: Affective, cognitive, and behavioral responses to the gender-role behaviors of males and females.

<u>Gender-Role Stereotypes</u>: Generalizations, based on an attitude structure, concerning the expected gender-role behaviors of males and females that influences the processing of information about individual members of each gender.

<u>Independent colleges</u>: Often referred to as private colleges, these institutions are autonomous from state control and derive their budgets primarily from private donations and tuition.

<u>Judicial Board</u>: A group authorized by the college to hear cases and levy discipline sanctions when enrolled students allegedly violate conduct standards. Judicial boards typically are selected, trained, and advised by student affairs staff.

, ٠. • : ..\* -• . **:** . ; Judicial Process: The due process requirements and administrative procedures followed by college staff when implementing discipline procedures against enrolled students alleged to have violated college conduct standards.

Legal Requirements for Rape: Within the state of Michigan, it is considered rape when force is used by the rapist and sexual penetration, defined as the insertion of any object into any body cavity of the victim, occurs. The victim is not required to resist the rapist, nor prove nonconsent.

Rape Myths: Biased, stereotypical, and inaccurate beliefs about the nature of rape, rape victims, and the rapist that create a hostile climate for rape victims.

<u>State-Assisted Colleges</u>: Often referred to as public colleges, these institutions derive the majority of their budget from state governments and are viewed as state entities.

<u>Stranger Rape</u>: Considered the classic rape, it involves a victim and rapist who had no relationship to each other prior to the assault.

<u>Suspension</u>: A discipline sanction requiring an enrolled student found guilty of violating college conduct standards to deactivate his/her enrollment for a pre-determined period of time.

# Limitations of the Study

This investigation had the following limitations inherent in the design of this study:

a) Studies demonstrate that real juries are less likely to find guilt than mock juries (Diamond and Zeisel, 1974) and are

more lenient when judging a college discipline case (Kerr, Nerenz, and Herrick as cited in Gerbasi, Euckerman, and Reis, 1977).

- b) Complex social phenomenon are difficult to define, resulting in an oversimplification that make it difficult to gather accurate measures (Babbie, 1990).
- c) Researchers using questionnaires must be aware that subjects may be careless, have faulty perceptions, lack interest, and/or are untruthful when completing the instrument (Turney and Robb, 1971). It is also possible that respondents are expressing attitudes that they have thought little about (Babbie, 1990).
- that makes it difficult to generalize beyond the case study being analyzed (Babbie, 1990; LaFree, 1989). Case studies also limit subject exposure to extra-legal variables (defendant/victim characteristics) that influence outcomes of hearings (Gerbasi, Zuckerman, and Reis, 1977).
- e) All institutions participating in this study used slightly different descriptors to explain their conduct rules and judicial sanctions. As such, the descriptors explaining the judicial sanctions in Section C (Adjudication Questionnaire) of the survey were a compilation. Interpretation of the results assumed that all respondents understood the descriptors before determining a punishment.

### Organization of the Study

This manuscript is organized into 5 chapters. The first chapter delineates the research questions and hypothesis of the study. Chapter 1 also provides a statement of the problem, the purpose of the study, and the value of researching the stated problem.

chapter 2 provides a literature review that is relevant to the stated problem and purpose of the study. This chapter explains rape according to legal, social, and general definitions. A theoretical perspective, with emphasis on feminist theory is provided. A review of rape laws, highlighting the law in Michigan, is presented. The prevalence of rape on college campuses is established. An overview of gender-role stereotypes, attitudes, and rape myth beliefs outlines their prevalence and effect on the attribution of blame and responsibility in acquaintance rape cases. Finally, a review of jury decision-making and college disciplinary procedures is provided.

Chapter 3 contains a description of the research methodology. Included is a description of the research subjects, instrumentation, data collection procedures, scoring the data, and data analysis techniques.

Chapter 4 provides a presentation and analysis of the research data. This chapter examines the demographics of the samples and provides empirical and descriptive analysis of the data.

Chapter 5 presents a summary of the findings, examines the limitations of the study, and discusses the major findings. Also included are the major conclusions of the study, including the implications and recommendations for further research.

#### CHAPTER TWO

#### LITERATURE REVIEW

#### Introduction

"The classic 'real' rape, for many people, is a rape by a stranger who uses a weapon--an assault done at night, outside (in a dark alley), with a lot of violence, resistance by the victim, and hence severe wounds and signs of struggle" (Burt, 1991, p.27). While most females are raped by males at night (Amir, 1971, 1975; Harlow, 1991; Katz and Mazur, 1979), rapes generally occur in the victim's or assailant's home (Ageton, 1983; Bart, 1975; Harlow, 1991; Kanin, 1984; Katz and Mazur, 1979), involve members of the same race (Amir, 1971, 1975; Curtis, 1976; Harlow, 1991), lack strong victim resistance (Amir, 1971, 1975), result from the use of psychological or verbal coercion rather than use of a weapon (Ageton, 1983; Amir, 1971, 1975; Harlow, 1991; Koss, 1988; Krasner, Meyer, and Carroll, 1976; Rapaport and Burkhart, 1984), and don't result in severe injuries (Bownes, O'Gorman, and Sayers, 1991; Harlow, 1991; National Victim Center, 1992; Yegidis, 1986). The majority of the rape victims were acquainted with the assailant (Aizenman and Kelley, 1988; Bailey, 1986; Bart, 1975; Copenhaver and Grauerholz, 1991; Koss, 1988; National Victim Center, 1992; Yegidis, 1986).

"The major threat that women face is not the unknown rapist lurking in the bushes at night, but men who women come into contact with on a day-to-day basis" (Bailey, 1986, p.35). Several studies (Aizenman and Kelley, 1988; Garrett-Gooding and Senter, 1987; Kanin, 1957, 1967, 1975, 1984; Kanin and Parcell, 1977; Koss, 1985, 1988; Koss and Dinero, 1988, 1989; Koss and Gidycz, 1985; Koss, Gidycz, and Wisniewski, 1987; Koss and Leonard, 1984; Koss, Leonard, Breezley, and Oros, 1985; Koss and Oros, 1982; Miller and Marshall, 1987; Rapaport and Burkhart, 1984), confirm the prevalence of acquaintance rape on college campuses.

According to Gibbs (1991), the classic rape "accounts for only 1 out of 5 of the attacks" (p. 48) on women.

·;		
J		

The remainder of this chapter reviews the literature pertaining to the topic of acquaintance or date rape among students who attend institutions of higher education in the United States The specific focus of this review is on the influence of gender-role stereotypes and rape attitudes on people's attribution of blame and responsibility in acquaintance rape situations. A special emphasis is on the attributional factors that influence the adjudication of acquaintance rape cases in both criminal justice and higher education judicial processes. Also reviewed are the general, legal, and social definitions of rape, emphasizing a feminist perspective. A comprehensive review of the prevalence of acquaintance rape in both society and higher education is provided to demonstrate the seriousness of the problem. A review of rape law is presented as a contrast to the social stereotypes that influence people's attribution of blame and responsibility in acquaintance rape situations.

# Definitions

#### General

"Rape definitions may be broad and inclusive or narrow and restrictive. At the inclusive extreme, radical feminists would maintain that all coerced sex is rape, whether the coercion used be physical, psychological, or economic. At the restrictive extreme, many people believe that there is no such thing as rape ..." (Burt and Albin, 1981, p. 213). Hueston and Burkhart (1986, p. 31) state that the definition of rape also varies according to "legal" and "social" definitions.

Whatever definition is used, rape in the United States is an offense overwhelmingly heterosexual, perpetrated by men against women (Amir, 1971, 1975; Harlow, 1991; Johnson, 1980; Russell and Howell, 1983)

There are references in the literature to three types of rape.

Koss and Harvey (1987) provide the following definitions:

(a) Stranger rape (classic rape) "involves a victim and offender who have no relationship to each other ..." (p. 11).

•

r;

; **;** 

- (b) Acquaintance rape "involves parties who knew each other prior to the assault and includes relatives, neighbors, or family friends" (p. 11).
- (c) Date rape happens "... when sexual intercourse occurs within a relationship superficially appropriate for sexual intimacy but where violence or threat of harm is used by the man, and the woman is not consenting" (p. 11).

Further defined, "acquaintance rape refers to assaults committed by anyone not a complete stranger to the victim which could include people she recognized even slightly or knew very well. Date rape is a specific type of acquaintance rape that involves a victim and perpetrator who have some level of romantic relationship between them" (Koss, 1992, p. 21).

According to Koss (1992, p. 22), "the essential meaning of the word rape is unaffected by the relationship of the parties. Legally, date and acquaintance rape are rape." Roark (1989) concurs, writing that "the relationship of the parties rarely enters into the legal definition ..." (p.41) of rape. Modern "rape laws are framed in such a way as to cover all rapes, whether perpetrated by strangers or by acquaintances" (Bohmer, 1991, p.317). Hughes and Sandler (1987) define rape as " ... an act of violence" (p. 1) that is "... often an attempt to assert power and anger" (p. 2). Rape, when "... stripped of its legal baggage ... refers to all forms of physical sexual intimacies in which significant physical force is used or threatened by one of the parties involved contrary to the will of the other" (Ellis, 1989, p. 2).

Burgess and Holmstrom (1980) further define stranger rape as a "blitz rape [that] occurs suddenly without warning and without prior interaction between assailant and victim ... Often he selects an anonymous victim and tries to remain anonymous himself." (p. 28). An acquaintance or "confidence rape is more subtle than the blitz rape [because] ... the assailant gains access to the victim under false pretenses, establishes [a] nonthreatening relationship, ... encourages her to trust him, and then he betrays this trust" (Burgess and Holmstrom, 1980, p. 29).

•.		
•		
') P•		
•		

Ľ

Rape may also be viewed as " ... a pseudosexual act, a pattern of sexual behavior that is concerned much more with status, hostility, control, and dominance than with sexual pleasure or sexual satisfaction" (Groth and Birnbaum, 1980, p. 21). Patterns of pseudosexual rape have been discerned by Groth (1979), and Groth and Birnbaum (1980), and Wilson (1978). The most commonly cited are those identified by Groth and Birnbaum (1980):

- (a) Anger rape involves sexuality that "becomes a means of expressing and discharging feeling of pent-up rage ... [and] is characterized by physical brutality" (p. 21).
- (b) Power rape is motivated by dominance. "... it is not the offender's intent to harm his victim but to possess her sexually. Sexuality becomes a means of compensating for underlying feelings of inadequacy and serves to express mastery, strength, control, authority and identity" (p. 22).
- (c) Sadistic rape involves a fusion of "... both sexuality and aggression ... into a single psychological experience .... Aggression itself is eroticized, and this offender finds the deliberate and intentional sexual abuse of his victim intensely exciting and gratifying" (p. 23).

Brownmiller (1975), offering a straightforward definition, stated that "if a woman chooses not to have intercourse with a specific man and the man chooses to proceed against her will" (p.18), then rape has occurred.

#### Legal

Courts often make decisions that interpret state and federal statutes and/or the rules and regulations of governmental agencies. These decisions, that provide further meaning, are referred to as the common law of the United States. "Common law, in short, is judge-made law rather than law that originates from constitutions or from legislatures or administrative agencies" (Kaplin, 1989, p. 13).

The typical common law definition, such as the one used in Alabama, defines rape "... as unlawful carnal knowledge of a woman not one's wife (intercourse), committed forcibly and without her consent" (Robin, 1977, p. 138). This traditional legal definition of rape encompasses four aspects (Bessmer, 1984; Estrich, 1987; Richmond, 1980; Robin, 1977;

Snelling, 1975; Tong, 1984; Virginia Law Review, 1975):

- (a) Penetration of the female vagina by the male penis.
- (b) The male uses force or threat of force against the female.
- (c) The female victim demonstrates nonconsent.
- (d) It is an extramarital occurrence.

According to Estrich (1987), this traditional or common law definition of rape "... remains the essence of even the most radical reform statutes" (p. 8). The Federal Bureau of Investigation (1989) continues to define rape as "... the carnal knowledge of a female forcibly and against her will" (p. 14). The most difficult issues involved in defining rape are determining the amount of actual or threatened force used, especially in acquaintance rape situations that are often ambiguous (Ellis, 1989).

Traditional rape laws in the United States have been heavily criticized as a visible symbol of the sexism that exists in society (Brownmiller, 1975; Clark and Lewis, 1977; Estrich, 1987; Griffin, 1979; MacKinnon, 1983; Medea and Thompson, 1974; Russell, 1975; Wood, 1973). As a result, rape law reforms have occurred in many of the states, most notably Michigan (Bessmer, 1984; Caringella-MacDonald, 1985; Chappell, 1984; Cobb and Schauer, 1974; Dean and deBruyn-Kops, 1982; Feild and Bienen, 1980; Galvin, 1985; Largen, 1988; Marsh, Geist, and Caplan, 1982; Polk, 1985; Praeger Publishers, 1977; Robin, 1977; Searles and Berger, 1987; Virginia Law Review, 1975). Further review of the legal system will occur later in this chapter.

# Social

Aside from the legal definitions, most people also define rape using social definitions (Hueston and Burkhart, 1986; Klemmack and Klemmack, 1976; Weis and Borges, 1973). According to Hueston and Burkhart (1986, p. 31), "the social definition of rape--that is, what people in general define as rape--is biased by a set of inaccurate assumptions or beliefs about the nature of rape." Commonly called rape

myths (Burt, 1980; Pineau, 1989; Schwendinger and Schwendinger, 1974), these assumptions are "... prejudicial, stereotyped, or false beliefs about rape, rape victims, and rapists [that] create a climate hostile to rape victims" (Burt, 1980, p. 217). These myths exist because they "... are simpler to understand and easier to accept and, therefore, more satisfying than the reality" (Groth, 1979, p. 11)

Several authors have written about rape myths prevelant in society (Barnett and Feild, 1977; Brownmiller, 1975; Burt, 1980; Griffin, 1975; Groth, 1979; Holmberg and MacDonald, 1990; Koss and Harvey, 1987; Krasner, Meyer, and Carroll, 1976; Malamuth, 1981; Medea and Thompson, 1974; Metzer, 1976; Schwendinger and Schwendinger, 1974; Warshaw, 1988). Examples of rape myths include:

- (a) All rapists are insane or sex-starved (Burt, 1980; Griffin, 1975).
- (b) Women secretly want to be raped (Brownmiller, 1975; Griffin, 1975).
- (c) Women like being raped (Griffin, 1975; Koss and Harvey, 1987).
- (d) Only "bad" girls are raped (Burt, 1980; Griffin, 1975; Koss and Harvey, 1987).
- (e) A woman means "yes" when she says "no" to sexual intercourse (Griffin, 1975; Koss and Harvey, 1987).
- (f) Rape victims provoke their own assault, e.g. type of clothes worn (Barnett and Feild, 1977; Brownmiller, 1975; Burt, 1980; Griffin, 1975; Schwendinger and Schwendinger, 1974).
- (g) Men can't control their animal instincts (Griffin, 1975; Groth and Birnbaum, 1980; Schwendinger and Schwendinger, 1975).
- (h) It's impossible to rape a woman who is unwilling (Brownmiller, 1975; Burt, 1980; Koss and Harvey, 1987; Schwendinger and Schwendinger, 1974).
- (i) A married woman can't be raped (Griffin, 1975).
- (j) Women falsely cry rape to be vindictive or to cover-up their actions (Burt, 1980).
- (k) The typical rape involves a stranger, occurs in a dark alley and results in injuries (Burt, 1991; Holmberg and MacDonald, 1990; Koss and Harvey, 1987).

The existence of these rape myths have been well established by researchers. Rape myths have been found prevelant in groups such as college students (Acock and Ireland, 1983; Barnett and Feild, 1977;

Bostwick, 1990; Bunting and Reeves, 1983; Check and Malamuth, 1985; Costin, 1985), criminal justice personnel (Chappell and Singer, 1977; Feild, 1978a; Feild and Bienen, 1980; Kalven and Zeisel, 1966; Kerstetter, 1990; LaFree, 1980; McCahill, Meyer, and Fischman, 1979), and in the general population (Amir, 1971, 1975; Burt, 1980; Burt and Albin, 1981; Costin, 1985; Feild, 1978a; Klemmack and Klemmack, 1976; Williams, 1979).

A court observer, after following 25 rape cases through the criminal justice system in Philadelphia, summarized the impact of rape myths as follows:

"The stereotypes which are seen in rape cases invariably portray a woman as degraded, oppressed, dependent, and victimized. They either desire assault and humiliation, are spiteful, or desperately in need of attention; they lose all control under threat and are thereby reduced to babbling idiots" (quoted in Krasner, Meyer, and Carroll, 1976, p. 27).

#### Theoretical Perspective

In the comprehensive book <u>Theories of Rape: Inquires into the Causes of Sexual Aggression</u>, Ellis (1989) identified three contemporary theories to help explain the causes of rape. The three theories, according to Ellis, "... have all been formulated in light of a growing body of research that has also examined the more prevalent, but far less violent forms of rape committed by acquaintances (date rape)" (p. 9). The theories identified by Ellis were evolutionary theory, social learning theory, and feminist theory.

This review will concentrate on feminist theory, offering only a very brief overview of evolutionary theory and social learning theory. This decision was made because evolutionary theory lacks relevance and because "social learning theory is similar to feminist theory in several respects; in fact, at times they have been virtually equated" (Ellis, 1989, p. 13).

·
.

·
.

·

•

# Evolutionary Theory

"The evolutionary theory of rape considers rape to be an act emanating from natural selection pressure for males to be more eager than females for copulatory experiences ..." (Ellis, 1989, p. 16). The emphasis of this theory is on biological reproduction, with the social roles of males and females determined by evolution. For example, "while natural selection may have favored males who were prone to use forceful tactics to inseminate females, natural selection would have also favored females who resisted such tactics" (Ellis, 1989, p. 15). According to evolutionary theory, rape is the result of the conflict that occurs between the male biological drive to reproduce, and the female desire to control the reproductive process.

# Social Learning Theory

"A major impetus for this theory came from the 1973 Presidential Commission on Pornography and Obscenity" (Ellis, 1989, p. 12). This theory postulates that males learn aggressive behavior toward women. The learning process is influenced by the mass media's portrayal of women, the association of sexuality and violence in our culture, the prevalence of rape myths, and the desensitization of society toward violence (Ellis, 1989). Social learning proponents focus more on cultural mores than political and socioeconomic exploitation (Ellis, 1989). The "... social learning theorists appear more receptive than feminist theorists to the view that rape may often reflect a genuine desire by rapists to have sex with their victims (especially regarding date rape)" (Ellis, 1989, p. 13; see also Kanin, 1984).

# Feminist Theory

Also referred to as "conflict theory" (Kerstetter, 1990; LaFree, 1989, p. 44) or "social control theory" (Koss and Leonard, 1984, p. 228), feminist theory "... considers rape to be the result of long and deep-rooted social traditions in which males have dominated nearly all

٠...

¥

•

• •

**(**:

• rf

•

•

**``.** 

·•

·

. *'* 

important political and economic activities" (Ellis, 1989, p. 10).

Because women are excluded from political decision-making, they are relegated to a status in society that is considered "inferior" to men.

As a result, "the assumption is that rape is ... expectable, even 'natural,' as a man's superior will or his 'having his way' with women is natural" (Morrison, 1980, p. 4). Thus, according to Metzger (1976), "rape is a ritual of power ... but it is also the symbolic enactment of social and cultural attitudes" (p. 405).

Feminist theory disagrees with the traditional psychiatric view that rapists are sexual deviants (Check and Malamuth, 1985). Instead, this theory "... regards rape as a male response to the social inequality between the sexes and the tendency of this inequality to affect the way men and women interact sexually" (Ellis, 1989, p. 10). According to Morrison (1980), "a woman, as man's inferior, finds her rightful place in society by fulfilling his destiny [because] she has no destiny of her own" (p. 4). Furthermore, Morrison (1980) stated that "... women, unrelated to men, have no intrinsic value and find their value ... only in relation to men" (p. 5) and that women are important to men "... as symbols of their power and as prizes of their prowess, thus bringing men status among their fellows ..." (p. 7).

Brownmiller (1975), in a comprehensive historical analysis of rape, documents that women were viewed as the property of their father under ancient Babylonia and Mosaic law. When a daughter was raped, her value was placed at fifty pieces of silver, payable to the father by the rapist. The silver was paid as compensation for "... the theft of virginity, an embezzlement of his daughter's fair price on the market" (Brownmiller, 1975, p. 18). When a married woman was raped, however, she was "... considered culpable, adulterous and irrevocably defiled" (Brownmiller, 1975, p. 19), no longer of value to her husband.

Lindemann (1984), after researching rape cases in eighteenthcentury Massachusetts, concluded that "the pattern is highly suggestive and supports Brownmiller's theory that rape was considered an offense

.. ı • • • 7 7 7

.

·••

8 L

pot

his

rap

(La

16)

not against a woman so much as against her father or husband" (p. 80; see also Dean and deBruyn-Kops, 1982). As a result, rape laws have historically "... developed not primarily to protect women who were raped, but to protect the property rights of their husbands or fathers" (LaFree, 1989, p. 46). "In this light, rape, as legally defined, appears more a crime against female monogamy than against female sexuality" (MacKinnon, 1983, p. 647).

Feminist writers (Brownmiller, 1975; Burt and Albin, 1981; Clark and Lewis, 1977: Griffin, 1975; MacKinnon, 1983; Metzger, 1976; Morrison, 1980; Russell, 1975; Sanday, 1981; Weis and Borges, 1973) stated that we live in a "rape culture" that supports rape myths and accepts violence against women. The belief that rape myths influence both individuals and the legal system is central to the feminist position. Koss and Leonard (1984) hypothesized that rape myths, by normalizing the use of force, allows men to avoid defining sexually aggressive actions as rape. Sanday (1981) in a cross-cultural study of rape, concluded that "rape is not an integral part of male nature, but the means by which men programmed [by the culture] for violence express their sexual selves" (pp. 25-26).

The following section provides a review of rape law in the United States.

#### Rape Law

Rape is a charge "... easily to be made and hard to be proved, and harder to be defended by the party accused, tho' never so innocent" (Hale, 1971, p. 635).

A noted British jurist from the late 1600s, Sir Matthew Hale's description of British common law symbolizes the scepticism Western society has toward rape victims. According to Harris (1976), this scepticism shaped the development of rape laws that were premised on the following basic assumptions:

(a) "... an unchaste woman is more likely than a chaste one to have consented to any particular sexual enounter" (p. 624).

- (b) "... the behavior of a rape complainant, whether she is sexually experienced or not, invited ... the sexual attack" (p. 624).
- (c) "... unchaste women are untruthful in general and often bring false charges of rape" (p. 626).
- (d) "distrust of the complainant's credibility has led to an exaggerated insistence on evidence of resistance" (p. 626).
- (e) "... a jury will sympathize with the alleged victim" (p. 627).

Tong (1984) stated that both contemporary and traditional legal thought on the subject of rape views women as "temptresses" and "liars" (p. 91). Reynolds (1974) writes that laws against rape act "... to punish men who do not rape appropriately" (p. 63). These beliefs were supported by the noted legal scholar Wigmore (1935) who wrote:

"Sexual offenses against women are justly looked upon by juries with particular abhorrence. On the other hand, there is always the risk of the jury's being misled by an attractive face and a plausible tale of woe. Modern psychiatry has further revealed that types of women are found whose unchaste temperament or diseased imagination leads them to the concoction of false charges of sexual offenses. The psychiatrists tell us that in all cases of such complaints the social history and emotional character of the complainant should be thoroughly investigated and reported on by experts before full credence can be given" (p. 303).

Since Wigmore wrote the aforementioned, the United States has experienced several social movements that have influenced the evolution of rape laws. Loh (1981) identified three social-political movements that "... pushed reform bills through state legislatures with uncommon swiftness and political acumen" (p. 34): Model Penal Code (see also Estrich, 1987; Robin, 1977; Virginia Law Review, 1975); Equal Rights Amendment (see also Praeger Publishers, 1977); and the crime control reform movement (see also Bessmer, 1984), from which the Michigan sexual conduct statutes evolved. The New York Penal Law was also viewed as a major model for rape law, serving as a contrast to the Model Penal Code (Virginia Law Review, 1975; see also Bessmer, 1984).

The Virginia Law Review (1975) identified four major models of rape law. The Common Law Model relied on the carnal knowledge statutes and

• r; \* \* **:**:

₹ • Ē · · • : . 4

was "... based on a concern with chastity and violation of a woman's virtue" (p. 1501). The Model Penal Code was published in 1962 and was built on the principles of the common law. The emphasis of the Model Penal Code was on "... specifying the minimum amount of coercion or deception necessary ... to classify all culpable conduct" (p. 1501). The New York Penal Law was adopted in 1961 and served as a contrast to the recommendations of the Model Penal Code, adopting very strict corroboration requirements (see also Yale Law Journal, 1972).

The fourth major model, the Michigan Criminal Sexual Conduct Statute, adopted in 1974, was viewed as the model rape law (Bessmer, 1984; Caringella-MacDonald, 1985; Chappell, 1984; Cobb and Schauer, 1974; Dean and deBruyn-Kops, 1982; Feild and Bienen, 1980; Galvin, 1985; Largen, 1988; Marsh, Geist, and Caplan, 1982; Polk, 1985; Praeger Publishers, 1977; Robin, 1977; Searles and Berger, 1987; Virginia Law Review, 1975). The Michigan Statute was considered a model because it codified definitions (e.g. sexual contact, sexual penetration, physically helpless) that may help determine a defendant's guilt or innocence (Cobb and Schauer, 1974) and used sex-neutral terminology (Caringella-MacDonald, 1985; Cobb and Schauer, 1974; Virginia Law Review, 1975). In addition, the Statute incorporated four degrees of criminal sexual contact that are distinguished by "... whether sexual penetration, as opposed to contact, occurred and whether certain forceful elements were present in the commission of the crime" (Cobb and Schauer, 1974, p. 221). Forcible elements are determined by amount of force or coercion used, the mental and physical capacity of the victim to give consent, the presence of a weapon, the amount of injury to the victim, the involvement of abetters, and the commission of other felonies (Cobb and Schauer, 1974).

Under common law, the female victim was expected to demonstrate clear nonconsent as a key element in determining if a rape occurred (Estrich, 1987; Harris, 1976; Loh, 1981). Part of nonconsent was demonstrating resistance "...high enough to assure that the resistance

VO.

and (Dw

als

:

·**.** 

: 💉

...

es

199

Mic

is th

fo

or

hi

i

:

[was] unfeigned and to indicate with some degree of certainty that the woman's attitude was not one of ambivalence or unconscious compliance and that her complaints [did] not result from moralistic afterthoughts" (Dworkin, 1966, p. 685). The amount of force used against a victim was also viewed as an indication of nonconsent (Harris, 1976; Virginia Law Review, 1975). Penetration of the vagina by the penis was an additional essential element in the common law view of rape (Robin, 1977; Tong, 1984).

According to Cobb and Schauer (1974), the model rape law in Michigan "... does not require the victim to resist the actor, nor does it require the victim's testimony to be corroborated. The prosecution is required to prove that force was used, but it does not have to prove the victim's nonconsent..., and [use of] the victim's past sexual conduct to prove consent is severely limited" (p. 223). The Statute focuses on the coercion used by the rapist, not on the victim's state of mind and presumes nonconsent when the victim is underage or physically or mentally helpless (Cobb and Schauer, 1974).

The Michigan Sexual Assault Law also limits admissible evidence to the specifics of the case and excludes evidence of the victim's sexual history or reputation, except when the evidence pertains to past sexual relations with the actor and to show the origin of pregnancy, disease, or semen (Cobb and Schauer, 1974). Sexual penetration is defined as insertion of any object, not limited to the penis, into any body cavity of the victim, including the mouth, anus, and vagina (Michigan Public Acts No. 266, 1974; see also BenDor, 1976 for a Common Law-Model Rape Law comparison).

While a landmark law, "Michigan's criminal sexual conduct law was clearly a necessary condition for the progress that has been made, but far from a sufficient one" (Marsh, Geist, and Caplan, 1982, p. 119; see also Caringella-MacDonald, 1988). For example, Estrich (1987) stated that Michigan law "...may obscure [rapes] unique indignity" (p. 81), focuses on gender-neutrality when in reality women do not rape men, and

opens to debate the question of "... whether rape should be considered sex or violence" (p. 82). The bill also does not limit the felony status of aggravated offenses to forcible felonies, nor does it define minimum sentences (Cobb and Schauer, 1974). Never-the-less, "the Michigan Sexual Assault Act reflects a major rethinking of the common assumptions about rape" (Cobb and Schauer, 1974, p. 235).

## Prevalence of Rape

## General

The Federal Bureau of Investigation (1980) reported that 71 out of every 100,000 females in the United States were victims of a forcible rape in 1980. This figure represented a 38% increase over the number of forcible rapes reported in 1976 (Federal Bureau of Investigation, 1976). In 1989, 94,504 females were victims of forcible rape, representing a 7 percent increase over reported forcible rapes in 1985 (Federal Bureau of Investigation, 1989). From 1973 to 1982, the Federal Bureau of Investigation (1982) showed an increase in reported forcible rapes of 51 percent. In 1989, one rape occurred every 6 minutes in the United States (Federal Bureau of Investigation, 1989).

The above figures represent only those offenses reported to the police and classified as rape or attempted rape by the legal authorities, thus distorting the true prevalence of the crime (Hindelang, Gottfredson, and Garofalo, 1978; Reiff, 1979). In fact, the Federal Bureau of Investigation (1984) recognizes forcible rape "... as one of the most underreported of all index crimes" (p. 14).

Furthermore, acquaintance rape victims are less likely than victims of a stranger rape to report the crime to police (Williams, 1984).

To gain more accurate incidence rates, a series of annual National Crime Surveys were instituted in the early 1970's. Under the direction of the Law Enforcement Assistance Administration, and later the Bureau of Justice Statistics, the surveys were carried out by the U.S. Bureau

7 ė v. . • **4**. 

res

pre

cri

of the Census (Russell, 1982). The National Crime Surveys asked residents of a standard sampling area to indicate those crimes that they or anyone else in their household had been victims of during the previous 6 months (Bureau of Justice Statistics, 1984). The survey results were then compared with the official crime statistics for the sampling area to get a better estimate of the prevalence of unreported crime (Koss, Gidycz, and Wisniewski, 1987).

In one of the early victimization surveys, the Law Enforcement Assistance Administration (1974) found that the level of criminal activity in 5 major cities was approximately double the number of offenses recorded by law enforcement agencies. One survey (Bureau of Justice Statistics, 1991b) found that female victimization rates changed very little between 1973 (23.9 per 1,000) and 1987 (24.0 per 1,000).

Harlow (1991) analyzed National Crime Survey data for the years

1973 through 1987 and found that "women ages 16 to 24 were 3 times more

likely to be raped than other women" (p. 8). Similar analysis confirmed

that females, ages 16 to 19 have the highest victimization rate per

1,000 people, with females ages 20 to 24 generally having the next

highest rate of victimization (Bureau of Justice Statistics, 1985b,

1986, 1987, 1991a; see also Amir 1971). The Federal Bureau of

Investigation (1989) reported that 44 percent of those arrested for

forcible rape in 1989 were under age 25, with the most frequent offender

being a 16 to 20 year old male (Federal Bureau of Investigation, 1976).

The National Crime Survey for 1991 yielded a statistically significant increase of 59 percent in the number of reported rapes and attempted rapes——from 130,260 in 1990 to 207,610 in 1991 (Squitieri, 1992). In contrast, a 1988 survey identified a total of 65,540 rape victims, of which 61 percent had been raped by nonstrangers (Bureau of Justice Statistics, 1990). The 1983 (Bureau of Justice Statistics, 1985a) and the 1986 (Bureau of Justice Statistics, 1988) National Crime Surveys discovered that nonstrangers were the rapists in 56 percent and 73 percent of the cases respectively. However, other National Crime

·

•

r

Surveys determined that only 50 percent of the reported rapes in 1987 (Bureau of Justice Statistics, 1989), 42 percent in 1985 (Bureau of Justice Statistics, 1987), and 12 percent in 1979 (McDermott, 1979) involved nonstranger rapists, suggesting methodological problems with the data collection methods.

The National Crime Surveys, while more accurate purveyors of victimization rates than the Federal Bureau of Investigation, were believed to still "... underestimate ... the prevalence of rape" (Russell, 1982, p. 81) among females in the United States. For example, the Law Enforcement Assistance Administration (1972), conducting a follow-up with 620 persons who reported crime victimization during 1970, discovered that only 84 percent of the rape attacks by strangers were reported to the National Crime Survey researchers. Furthermore, only 54 percent of the rape attacks by assailants known to the victim were disclosed in the original survey.

Russell (1982), in an attempt to provide a more accurate estimate of the prevalence of rape, conducted in-person interviews with 930 randomly selected females ages 18 and over in San Francisco. The survey, using the legal definition of rape in California, found that 41 percent of the women reported at least 1 completed or attempted extramarital rape. Only 9.5 percent of the extramarital rape or attempted rape victims reported the incident to the police. The results of this study represented a figure 13 times higher than the number of rapes reported to the Federal Bureau of Investigation.

Using a national probability sample of 1,725 adolescents between ages 11 and 17, Ageton (1983) surveyed the same sample each year for 5 years. The research discovered that between 5 and 11 percent (depending on the year) of the adolescent females in the sample experienced at least 1 sexual assault. Eighty percent of the offenders were known to the females, being primarily boyfriends or dates. Ageton (1983, p. 48) found that "most adolescent victims do not report their sexual assault experiences to police because they do not perceive them as legitimate

**→** 

.

·

•

•

· **,** 

•€-:` C.

. :•

(they do not involve strangers or substantial violence) ... "

Believing that both the Federal Bureau of Investigation and the National Crime Survey fail to detect a substantial proportion of rapes, a National Women's Study (National Victim Center, 1992) was designed, in part, to gather information on forcible rapes in the United States. The study was a longitudinal survey of a national probability sample of 4,008 American women ages 18 or older. Respondents were interviewed by phone, with 85 percent of the initial sample agreeing to participate in the research project.

The National Women's Study found that 13 percent of the women surveyed reported having been victims of at least 1 completed rape, an estimate 5.3 times larger than the 1990 National Crime Survey. Fifty-five percent of the rape victims were between the ages of 11 and 24 at the time of their rape, with 22.2 percent of the victims being between the ages of 18 and 24. Seventy-eight percent of reporting rape victims were assaulted by someone they knew or had seen before. Only 16 percent of the rapes were reported to the police. These findings, if generalized to the population, represent an estimated 683,000 female rape victims in 1990.

## Higher Education

Bogal-Allbritten and Allbritten (1985) documented the level of awareness campus officials have concerning the prevalence of courtship violence on college and university campuses. Eighteen percent of 345 Directors of Campus Housing and 30 percent of 228 Counseling Center Directors responding to a mail survey reported knowledge of more than 10 incidents of courtship violence on campus. In a second study, Bogal-Allbritten and Allbritten (1991) found that knowledge of rape incidents was reported by 20 percent of 294 Campus Security Directors, 27 percent of 495 Chief Student Affairs Officers, and 41 percent of 252 Health Center Directors.

Other sources of data on rape and sexual assault on campuses of

•

.

•

higher education include the Federal Bureau of Investigation and the Campus Violence Prevention Center. According to the Federal Bureau of Investigation (1980), approximately 268 rapes were reported by institutions of higher education in 1980. In 1985, approximately 228 rapes were reported (Federal Bureau of Investigation, 1985) and in 1989, approximately 360 institutions reported approximately 243 rapes (Federal Bureau of Investigation, 1989).

Smith (1988) stated that "... these totals are drastically understated ... [because] only about ... 15 percent of the total number of colleges and universities in the United States" (p. 17) report crime data to the Federal Bureau of Investigation. The schools reporting crime statistics to the Federal Bureau of Investigation also vary from year to year, making comparisons difficult (Cockey, Sherrill, and Cave, 1989).

In an effort to gather more accurate data on the prevalence of campus violence, including sexual assault and rape, the Campus Violence Prevention Center (1988) distributed a series of national yearly surveys to approximately 2,700 colleges and universities. The survey results showed that the average number of sexual assaults occurring on each reporting campus was 3.3 in 1986, 3.0 in 1987, and 1.7 in 1988. The 437 institutions responding to the 1990 survey (Campus Violence Prevention Center, 1990) reported occurrences of only 429 sexual assaults and 215 rapes. Sixty-two percent of these rapes and sexual assaults were perpetrated by dates or acquaintances.

Official sources do not appear to provide comprehensive data on the prevalence of rape and attempted rape occurring at institutions of higher education in the United States. Thus "in order to develop a more accurate estimate of the magnitude of this problem, researchers have developed measures based on a behaviorally anchored set of questions. This procedure allows a researcher to obtain reports of events via behavioral descriptions, which can then be matched to a criterion definition of rape and thus bypass the subject bias introduced by using

111

re;

:a;

. •

.

•

•

•

÷.

the word rape" (Burkhart and Stanton, 1988, p. 44). Using this procedure allows researchers: (a) to look at an individual's self-rated likelihood of raping under certain conditions; or (b) to determine self-reported incidents of sexual aggression using a continuum approach where rape represents the extreme point (Koss and Leonard, 1984).

Gilbert (1991) and Guttman (1990) were critical of the behavioral description approach to measuring sexual aggression. Gilbert (1991) stated that a "rape crisis movement agenda" (p. A14) was operating that distorted the definition of rape and reported false results (see also Hendrix, 1991; Irving, 1991). It was Gilbert's opinion that Bureau of Justice statistics were a more accurate reflection of the prevalence of rape in the United States. Gutmann (1990) cited official rape statistics from three universities as documentation that rape was occurring less than was being reported by researchers using survey methods.

Koss (1992), the primary object of Gilbert's criticism, persuasively refuted these critics by pointing out that the definition of rape used in her research was consistent with most legal statutes, that many other researchers have replicated her results, and that women knew they were victimized whether or not they realized they had been legally raped. Koss (1992) also noted that her national survey "... has not provoked any critiques in the professional literature..." (p. 24).

Ward, Chapman, Cohn, White, and Williams (1991) noted, however, that "a wide range of estimates of the incidence of sexual assault is found in the literature" (p. 65). They stated "this may be due in part to varying methodologies and in part to variations in the definition of the behavior covered by the estimates" (p. 65). Ellis (1989) agreed that the "... estimates are far too varied to be dismissed as resulting from sampling error...[thus] they appear to reflect major differences in criteria used to identify rape" (p. 4). Never-the-less, Ward et al. concluded that rape "... is an important social problem on college campuses, even if one uses the 'low'..." (p. 65) estimates of sexual

whe the Uni

,

1

ţ

,

**.•** .

.

.

...

Ş

•es,

su: wa on Ki

> of aç

> > pe ac in

t 8

i

victimization found by researchers. The problem is especially apparent when one considers that approximately 8.5 million students attend one of the 2,141 pulbic or private 4-year postsecondary institutions in the United States (The Nation, 1992).

Kirkpatrick and Kanin (1957) conducted a pioneering study using the survey techniques described by Burkhart and Stanton (1988). The study was designed to determine the patterns of female exploitation occurring on a University campus. Using a sample of 291 female students, Kirkpatrick and Kanin determined that 55.7 percent of the females reported being sexually offended at least once during the academic year at some level of erotic intimacy. More specifically, 20.9 percent were offended by a male's forceful attempts at intercourse and 6.2 percent by aggressively forceful attempts to have sexual intercourse. The offended women experienced an average of 6.3 aggressive episodes each.

Further investigation of male sex aggression by Kanin (1957), using a sample of 262 female university students, found that approximately 62 percent of the women reported experiencing offensive male sexual aggression during the prior year. Almost 9 percent of the sample indicated that they had been offended by a male who attempted sexual intercourse by using violence as the means. Fifty-eight percent of the total number of reported offensive episodes were not stimulated by prior sexual foreplay.

Kanin and Parcell (1977) replicated and extended the original study completed by Kirkpatrick and Kanin (1957), in part, to determine the impact of societal changes on sexual aggression. Results indicated that almost 51 percent of the 282 female respondents were offended at some level of erotic intimacy during the academic year. Approximately 32 percent of the females who experienced male sexual aggression, where intercourse was the goal, reported that rape had occurred. Fifty-four percent of the intercourse efforts were initiated by the male without previous sexual foreplay. Kanin and Parcell (1977) concluded that "the normative pattern of sexual aggression for males and its frequent

•

-

·

escalation into offending force, has apparently not been disturbed by the various social movements..." (p. 75). This supported Korman and Leslie (1982) who stated that women were victimized at approximately the same rate as in the 1950s, except that the aggressive episodes have focused more toward intercourse.

In the book I Never Called It Rape, Warshaw (1988) cited two other studies demonstrating the prevalence of acquaintance rape on college campuses. At St. Cloud State University research found that 29 percent of the women surveyed reported being physically or psychologically forced to have sexual intercourse. Sixteen percent of the women surveyed at Brown University reported having been raped by an acquaintance, while 11 percent of the men surveyed admitted to having forced a woman to have sexual intercourse (also cited in Sherman, 1985).

At Florida State University nearly 8 percent of the 362 females surveyed had been raped, and 17 percent were victims of attempted rape. Approximately 93 percent of these rapes and 81 percent of these attempted rapes were committed by acquaintances (Bailey, 1986). Wilson and Durrenberger (1982) sampled 447 college females and found that 15 percent had been raped and 18 percent had experienced an attempted rape while on a date. Check and Malamuth (1985) cited a study by Byers and Eastman at West Virginia University that found 38 percent of the survey respondents had been victims of forced sexual activity. At the University of Rhode Island, 8.7 percent of the undergraduate females surveyed reported having been sexually assaulted at least once at the University——27 percent had been sexually assaulted at least once in their lives somewhere other than the University (Lott, Reilly, and Howard, 1982).

At the University of South Dakota 21 percent of the 247 females sampled reported that they had been physically forced by a dating partner to have sexual intercourse (Sandberg, Jackson, and Petretic-Jackson, 1987). Researchers discovered from a sample of 362 female university students, that 12 percent had been raped and 19 percent had

• • **:** • : , • .. • ÷ . . • ÷ ŷ . 15 : .54 , •

experienced an attempted rape (Baier, Rosenzweig, and Whipple, 1991). Copenhaver and Grauerholz (1991) found that almost one-fourth of the 140 females sampled at a large Midwestern university were victims of an attempted rape, and 17 percent of the women had experienced an act of rape. Ninety-five percent of these women were acquainted with their attacker.

Additional research continues to show similar patterns of female victimization. For example, Miller and Marshall (1987) conducted a survey involving 472 men and 323 woman at two large universities. Using the Sexual Experiences Survey developed by Koss and Oros (1982), they concluded that "if date rape is defined as psychologically or physically forced sexual intercourse, then 27 percent of the women ... indicated that they had been victims of date rape while attending the university; 15 percent of the men indicated that they had victimized women through coercive sex while in dating situations at the university" (pp.44,46).

Muchlenhard and Linton (1987) surveyed 341 women and 294 male college students and found that 65.1 percent of the women and 50.9 percent of the men reported involvement in incidents of sexual aggression during college. Based on the results of a deviance survey administered to 836 male and 563 female students in classes at a midwestern university over a 4-year period of time, Eskridge (1989) discovered that 7 percent of the men had forced a female to have sexual intercourse against her will at least once. The Eskridge survey determined that 21 percent of the females had been raped; also discovering an escalating pattern of reported female rape that rose from 10.2 percent of the sample in 1985 to 25.2 percent in 1988.

At the University of Wisconsin-Whitewater, 8.2 percent from a sample of 147 females reported experiencing at least ! incident of forced unwanted vaginal, oral or anal penetration (Berger, Searles, Salem, and Pierce, 1986). Twenty-two percent of 348 female university students surveyed by Yegidis (1986) reported having been forced to engage in sexual activity by a male. Of the 22 percent, 52 percent had

been forced to have either intercourse or oral sex at some point in their lives. Five percent of the 348 person sample had experienced forced intercourse or oral sex during the previous year.

From a sample of 204 females at Rutgers University, Aizenman and Kelley (1988) discovered that 22 percent had been involved in a situation defined as acquaintance rape, and 51 percent reported that they had successfully avoided an acquaintance rape. Using Koss and Oros' (1982) Sexual Experiences Survey, Amick and Calhoun (1987) found that 75 percent of the 206 female students sampled reported having been victimized by a male. Ninety-four percent of the victimized females were acquainted with the person that victimized them. Fifty-eight percent of the sample had experienced sexual intercourse against their will.

Koss and Oros (1982) developed the Sexual Experiences Survey, a 12 question yes-no style research instrument to find hidden rape victims often missed by official sampling methods. The Sexual Experiences Survey was also designed to measure the level of sexual aggression associated with various degrees of coercion, threat, and force.

Additionally, the questions in the Sexual Experiences Survey reflected a continuum of sexual aggression, ranging from intercourse achieved through verbal coercion on one extreme to use of physical force (rape) at the opposite extreme (see also Koss and Gidycz, 1985 for data on reliability and validity).

The Sexual Experiences Survey was given by Koss and Oros (1982) to a representative sample of 3,862 university students (1,846 males, 2,016 females). The research determined that 6 percent of the females had been raped, 6.4 percent had been forced to participate in anal or oral intercourse because physical force or threats had been used, and 8.2 percent of the women had sexual intercourse because the man used some degree of physical force. Further analysis by Koss (1985) found that 38 percent of the females reported victimization that met the legal definition of rape or attempted rape, with only 4 percent reporting the

•

£

••

¥

2 h .

•.

assault to the police.

Koss, Gidycz, and Wisniewski (1987) administered the Sexual Experiences Survey to a national sample of 3,187 women and 2,972 men at 32 diverse institutions of higher education in the United States (see also the "Afterword" in Warshaw, 1988 for an explanation of survey methodology). The results of the survey indicated that 12.1 percent of the females had experienced attempted rape and 15.4 percent had been raped. The research also found a rape victimization rate 10-15 times greater than National Crime Survey rates. Koss, Gidycz, and Wisniewski concluded that "at least among students in higher education ... official surveys ... fail to describe the full extent of sexual victimization" (p. 168).

Researchers have demonstrated that men self-report involvement in sexually coercive behavior at levels below that reported by women. For example, Koss and Oros (1982) found that only 2.4 percent of the men in their sample reported obtaining sexual acts such as anal or oral intercourse by using threats or force and only 2.7 percent admitted using some physical force to have intercourse with a woman. In their national survey, Koss, Gidycz, and Wisniewski (1987) documented that 3.3 percent of the men reported involvement in attempted rape and 4.4 percent reported engaging in rape.

Other researchers have found higher reporting rates for males in terms of using sexual coercion. For example, Yegidis (1986) found that 5.7 percent of a male sample (n=300) reported forcing a female date to engage in sexual activity in the prior year. Kanin (1967) found that 26 percent of a sample of 341 male undergraduate college students described the use of aggressive tactics to obtain intercourse since entering college. In a study of 201 male university undergraduate students, Rapaport and Burkhart (1984) discovered that 28 percent of the men reported having used a directly coercive method at least once in the past to obtain sex from a woman. Furthermore, 15 percent admitted to having forced a woman to have intercourse at least once and 12 percent

acknowledged having physically restrained a woman to gain sexual advantage. Mosher and Anderson (1986) found in a sample of 175 men at the University of Connecticut that 13 percent had threatened to use force and 20 percent had used force to have sex with an unwilling woman. Fourteen percent of a sample of 99 men at Alfred University admitted using pressure to have intercourse with a woman (Peterson and Franzese, 1987).

## Influence of Alcohol and Other Drugs

According to Boyer (1987), citing a Carnegie Foundation survey, 42 percent of a 1984 national sample of college undergraduates believed alcohol abuse was a problem on their campus. Researchers at the U. S. Department of Education stated that binge drinking was a problem on college campuses, often being associated with the occurrence of sexual assaults (Cage, 1992). Polonko, Parcell, and Teachman, as cited by Bechhofer and Parrot (1991), concluded that the degree of intoxication of a man was the most important factor in determining whether an acquaintance rape occurred. Scully and Marolla (1984) found that around 70 percent of the incarcerated rapists (n=114) they interviewed told of their use of alcohol or drugs when giving an account of the rape.

Muehlenhard and Linton (1987) discovered that heavy drug or alcohol use was the most common on dates where sexual aggression occurred. Approximately 66 percent of 71 self-disclosed date rapists implicated excessive alcohol consumption as a causal factor in the assault (Kanin, 1984). Research involving 155 sexually victimized female students found that 65 percent of their offenders were moderately to very intoxicated, and 58 percent of the females reported they were not intoxicated or were only mildly intoxicated (Amick and Calhoun, 1987). At Bentley College, 373 men reported involvement in 241 acts of sexual assault. Of those men who reported involvement in a rape (n=27), 93 percent indicated that they had been drinking or taking drugs when they committed the assault

..

÷

•

.

`:

ī.

(4)∞

• :

3·

•

(Association of American Colleges, 1990). Twenty-one percent of 486 college women admitted engaging in unwanted sexual intercourse due to their state of intoxication (Muehlenhard and Cook, 1988).

Mosher and Anderson (1986) surveyed 175 college men and found that 19 percent had gotten drunk and forced a woman to have sexual intercourse, and 66 percent had used alcohol to get a woman drunk to have sex with her. Seventy percent of the male assailants involved in forced sexual interactions indicated that they were under the influence of alcohol or other drugs at the time (Miller and Marshall, 1987).

Makepeace (1981) found that alcohol was a factor in courtship violence when 46 percent of the men (n=99) and 35 percent of the women (n=103) reported being under the influence of alcohol at the time of the episode.

Koss (1988) in a national study of over 6,100 college students, found that 73 percent of the offenders and 55 percent of the victims admitted using drugs or alcohol at the time of the sexual victimization. Twenty-five percent of attempted rapes, where violence was used, involved the consumption of alcohol by the male (Kanin, 1957). Abbey (1987) found that misperceptions about sexual intent occurred most at parties where alcohol was freely available. Copenhaver and Grauerholz (1991) and Ward et al. (1991) found that males used drugs or alcohol in around 75 percent of the sexual victimizations that occurred.

Hammock and Richardson (cited by Richardson and Hammock, 1991) and Richardson and Campbell (1982) found that rapists were perceived to be less blameworthy and less responsible when drunk as compared to being sober. Both sets of authors also found that female victims were judged to be more responsible for the rape when consuming alcohol than when not. Additional research by George, Gournic, and McAfee (1988) supported the premise that males judge women who were drinking alcohol differently from those not drinking alcohol. For example, women who drank alcohol were seen as significantly less socially skilled, significantly more aggressive, and significantly more willing to have

sex than non-drinking women.

A review of the literature concerning alcohol and human sexuality by Crowe and George (1989) found, in part, that alcohol facilitates preexisting inclinations for sexual aggression and acts as a scapegoat to justify aggressive sexual behavior. An example included a study by Briddell, Rimm, Caddy, Krawitz, Sholis, and Wunderlin (1978) involving 48 undergraduate students that concluded "... alcohol's influence on sexually aggressive arousal patterns is mediated by psychological rather than physiological processes" (p. 428).

## Gender-Role Perception

#### Stereotypes

"The term stereotype is ... a cognitive structural concept, referring to a set of expectations held by the perceiver regarding members of a social group" (Hamilton, 1979, p. 65). McCauley, Sitt, and Segal (1980, p. 197) "suggest that stereotypes are those generalizations about a class of people that distinguish that class from others...[by using] differential trait attribution or differential prediction based on group membership." According to Hamilton (1979), "stereotyping is said to occur when a perceiver makes inferences about a person because of that person's membership in some group" (p. 54). Thus "stereotypic schemes conceivably could influence the processing of information about an individual member of a social group .... [by] focusing one's attention on a particular aspect of the person's behavior..., [by] leading the perceiver to interpret certain behaviors in a biased manner..., [and by] resulting in a selective retrieval of information from ... memory" (Hamilton, 1979, p. 68).

Researchers have documented that members of society have generally maintained consistent stereotypes about women and men since the 1940's.

Fernberger (1948) found that men and women held the same general opinions concerning gender stereotypes. Fernberger (1948) administered

\*\*

7 t - 1

:

• •

•

•

a sex difference test to 271 college students (143 men, 128 women) that found both genders believed men were more intelligent and possessed more all-around superior qualities. The same study determined that both genders perceived women to be less sexually passionate, more sensitive, and more talkative than men.

Sherriffs and Jarrett (1953) and Sherriffs and McKee (1957) found general agreement, using forced choice adjective questionnaires, between the mean scores of men and women with respect to their perceptions of the behaviors and characteristics attributed to each gender. A study of 100 (50 male, 50 female) college students by Sheffiffs and McKee (1957) found significant consistency between the genders in how males and females were described. For example, men were described using adjectives such as intelligent, rational, competent, bold, effective, and straightforward; while women were described by adjectives such as warm and concerned. Sherriffs and Jarrett (1953) concluded that "... virtually no behavior or quality escapes inclusion in either a male or female stereotype; and that these stereotypes are substantially the same whether held by men or women" (p. 167).

Rosenkrantz, Vogel, Bee, Broverman, and Broverman (1968) asked 74 males and 80 females to complete a 122-item stereotype questionnaire three times; the first as they believed a male would answer, secondly as they believed a female would respond, and finally as themselves. The results indicated a significant difference between the stereotypes associated with males and females, with stereotypical masculine characteristics perceived as significantly more socially desirable than stereotypical feminine characteristics. McKee and Sherriffs (1957) also found that both male (n=85) and female (n=93) college students ascribed significantly more favorable adjectives to males, with females ascribing significantly more unfavorable adjectives to themselves.

Administering the same questionnaire as Rosenkrantz et al. (1968) to almost 1,000 people, Broverman, Vogel, Broverman, Clarkson, and Rosenkrantz (1972, p. 61) concluded that:

- (a) "A strong consensus about the differing characteristics of men and women exists across groups..."
- (b) "Characteristics ascribed to men are positively valued more often than characteristics ascribed to women." Positive male traits formed clusters around things such as competence, rationality, and assertion, while positive feminine traits formed clusters reflecting warmth and expressiveness.
- (c) "The sex-role definitions are ... accepted to the extent that they are incorporated into the self-concepts of both men and women ... [and] are considered desirable by college students."

Ward and Balswick (1978) conducted a content analysis of the responses to several open ended questions by 1,234 high school students in Georgia. They also found that males and females demonstrated a high level of consensus on gender-role stereotypes. The most salient stereotypes for women were related to interpersonal virtues (weak, submissive, physically attractive, sexual, emotional). The stereotypes related to men involved strength, dominance, responsibility, and intelligence. Lunnenborg (1970) found similar results in a study of 532 college students that determined gender stereotypes were prevasive, with men and women having similar stereotypes regarding gender personalities (e.g. men are more intelligent). Deaux (1976) reviewed the attribution theory literature, also finding that men and women hold similiar gender-role stereotypes.

#### Dating Relationships

The gender-role stereotypes held by males and females in this culture influence the dating patterns of students. For example, LaPlante, McCormick, and Brannigan (1980) administered a Locus of Control Scale, a Sexual Script Questionnaire, and a Dating Attitudes Questionnaire to 71 male and 116 female unmarried college students to determine the extent to which traditional gender scripts influenced their lives. Using an analysis of variance, the researchers found it significant that students described gender differences consistent with traditional scripts. That is, men use strategies to have sexual intercourse whereas women use strategies to avoid intercourse (see also McCormick, 1979).

144 ٠, • ÷ **r** . ÷ • 4 •

Goodchilds, Zellman, Johnson, and Giarrusso (1988) read 432 adolescents a series of short vignettes, each involving a male-female encounter, and then asked a series of questions about who was responsible for what happened. The authors concluded that "the findings show significant and consistent sex differences in how adolescent respondents viewed the characteristics of dates and in the meaning they attached to behavioral and situational cues in dating situations.

Female respondents saw things in a less sexualized way——how people dress, act, or where they go were less likely to be taken as cues by female respondents that the actor desires sexual relations. In contrast, male adolescents tended to see the motivations of both male and female actors as sexual" (p. 264).

Muchlenhard, Friedman, and Thomas (1985) presented 268 male undergraduates with a series of written vignettes that manipulated who initiated the date, where the couple went, and who paid the expenses. The respondents then rated to what extent the female wanted to engage in sex and how justified the man was in proceeding against her wishes. Respondents also completed the Attitudes Toward Women Scale (Spence and Helmreich, 1972) that measures attitudes concerning the rights and roles of women in society.

Rape was rated significantly more justifiable if the man paid all the dating expenses, if the couple went to the man's apartment rather than to a religious function, and if the woman asked the man out on the date. Traditional men, as determined by the Attitudes toward Women Scale, had higher overall rape justifiability ratings than nontraditional men, although the differences did not reach significance. Never-the-less, Muehlenhard, Friedman, and Thomas (1985) "... argue that it is socially significant that 20 percent of the traditional men, as well as 12.9 percent of the nontraditional men, said rape was at least somewhat justifiable ..." (p. 307) when the woman initiated the date and then agreed to go to the man's apartment.

Approximately 48 percent of a sample of 408 college students stated

**,**· • • • • • ·. : 3 • . • 2 that sexual activity was expected when a dating partner paid for an expensive dinner and movie (Sandberg, Jackson, and Petretic-Jackson, 1987). Knox and Wilson (1983) found that unwanted pressure to engage in sexual activity was the most frequent dating problem defined by 227 university women. Both Goodchilds et al. (1988) and Kikuchi (as cited by Bechhofer and Parrot, 1991) discovered that both females and males found the use of force to have sex acceptable under a variety of dating circumstances, e.g. he spends alot of money on her, they have dated for a long time, she gets him sexually excited, she was wearing seductive clothing.

Muchlenhard and Scardino (1985) surveyed 309 male psychology undergraduate students and determined that perceived intelligence appeared to be more important than initiating a date in terms of influencing men's attributions about sexual activity. Women perceived as less intelligent were seen as less religious, as more casual daters, as less feminist, and as less truthful than perceived intelligent women (see also Kanin, 1975). Muchlenhard and Miller (1988) found no significance when comparing traditional men and nontraditional men's reactions to being asked out for a date by a woman.

Another study involving 272 female and 268 male undergraduates (Muehlenhard, 1988) found results very similar to the study conducted by Muehlenhard, Friedman, and Thomas (1985). Additionally, across all situations, men rated the woman in the vignette as more willing to have sexual intercourse than did women respondents. Men also generally believed that rape was more justifiable than did women, especially in circumstances involving who initiated the date and where the couple went on the date. Rape justifiability rating were consistently influenced by the level of traditionality, with 30 percent of the traditional respondents and 15 percent of the nontraditional respondents rating rape as at least somewhat justifiable under certain circumstances.

Bostwick (1990) replicated and extended Muehlenhard's (1988) study, finding that the men (n=201) in the sample significantly rated women as

.

•

.

having a greater willingness to have sexual intercourse than did the women (n=257) in the sample. This result was consistent with previous research done by Abbey (1982) and Abbey and Melby (1986) that determined college males view the world in a more sexualized manner than women, and thus were more likely to misinterpret ambiguous behavior on the part of females as willingness to have sexual intercourse.

Check and Malamuth (1983) found "... that sex role socialization processes are most likely to influence behavior within the date or acquaintance-rape situation, as contrasted with the stranger-rape situation" (p. 344). Weis and Borges (1973, p. 86) provided insightful commentary on the dating system and rape:

"The stereotypic notions of male and female roles and their relationship to conceptions of masculine and feminine sexuality, coupled with a situation that is fraught with ambiguous expectations, provide the ingredients for systematically socialized actors who can participate in the drama of rape. The dating system, by setting up such situations, accustoms the male to encounters in which he feels some justification and need to demand intimate sex and to approach this goal forcefully."

# Rape Myth Acceptance

Burt and Albin (1981), using data collected from 598 adults 18 and older, concluded "... that the breadth or narrowness of rape definitions depend in part on the rape-supportive attitudes held by the general public" (p. 226). Burt (1980) stated that these data also confirm "... the hypothesis that rape myth acceptance forms part of a larger and complexly related attitude structure that includes sex role stereotyping, feelings about sexuality, and acceptance of interpersonal violence" (p. 228). Research conducted by Wilson, Faison, and Britton (1983) involving 250 college students suggested that belief in rape myths influences the definitional boundaries between rape and normal sexual intercourse.

To test the thesis that acceptance of rape myths are related to people's views of women's social roles and rights, Costin (1985) modified two scales: (a) Attitudes Toward Rape Questionnaire (Feild,

• . • į.

1978a) that measures stereotypes and beliefs about rape; and the (b)
Attitudes Toward Women Scale (Spence and Helmreich, 1972). The
instruments were administered to a sample that included 432 female
college students, 140 male college students, and 114 women and 76 men
who were employed in the local community. All correlations for each
group were significant, indicating that belief in rape myths and related
views of rape were positively correlated with beliefs about women's
social roles. Further analysis indicated that men had higher mean
scores on both scales than did women.

Check and Malamuth (1985) reviewed studies concerning rape myths and concluded that "... evidence was found to support the hypothesis that rape and forced sexuality are widespread and, to a surprising extent, acceptable in North American society; that rape is not exclusively the product of a few bent and twisted minds; that beliefs in rape myths are linked to acts of aggression against women; and that sexual violence in the mass media plays a significant role in the fostering of rape myths ..." (p. 414).

Because many researchers investigated the influence of rape myths under artificial conditions, Check and Malamuth (1985) conducted a three phase study to determine whether scales measuring rape myth could predict reactions to fictional and real rape, as well as predict likelihood of raping. After giving the scales developed by Burt (1980) to 71 female and 57 male college students, the researchers concluded that the scales predicted responses to both fictional and real rape. Specifically, it was significant that students with a high level of belief in rape myths stated that raped women desired victimization, had a positive experience, were responsible for their rape and exhibited behavior that contributed to their rape.

Research has established that college students have a low level of awareness about rape and are likely to believe common rape myths. For example, Holmberg and MacDonald (1990) asked 96 students about their perceptions of circumstances where rape is most likely to occur. The

results indicated that 85 percent of the females stated that a dimly lit parking lot was either the most likely or second most likely place for rape to occur. Conversely, only 17 percent of the females stated that being alone with a date in either party's residence was the most likely place where rape could occur. Forty-two percent of the males, however, stated that a residence was the most likely or second most likely place for rape to occur. The authors of this study, using official university records, found that in reality 79 percent of all reported assaults took place in residence halls.

Using a sample of 323 female and 332 male university students, Jenkins and Dambrot (1987) had each student read 3 rape scenarios, plus complete the Sexual Experiences Survey (Koss and Oros, 1982) and Rape Myth Acceptance Scale (Burt, 1980). Results indicated that males had a significantly higher acceptance of rape myth than females. Both males and females with high rape myth acceptance were significantly more likely to blame the rape victim for the assault. There was also a significant main effect for males with high rape myth belief who stated that a rape involved little violence and that victims wanted sexual intercourse to happen.

Additional research also supports the link between rape myth and sexual assault. Koss, Leonard, Breezley, and Oros (1985) found that men with sexually aggressive experiences accepted gender-role stereotypes, believed in rape myths, saw rape prevention as a woman's responsibility, and viewed the intermingling of sexuality and aggression as normal. This research also determined that men who have threatened or used force for the purpose of having sexual intercourse with female acquaintances, differed from sexually nonaggressive men in their degree of adherence to rape-supportive attitudes.

Bunting and Reeves (1983) administered two self-developed scales designed to assess macho personality constructs and rape beliefs to 400 male college students, finding that gender-role ideology and general belief in rape myths were significantly correlated. Using the Bem Sex

•\_

.

• •

.

•

• •

w<sub>k</sub> \*

:

.

Role Inventory (Bem, 1974) and a rape questionnaire with 392 junior college students, Tieger (1981) discovered that gender-role stereotyping was causally related to rape attitudes. Mazelan (1980) used a sample of 350 adolescents to investigate the images people hold about rape victims, finding that rape victims were rated as significantly more neurotic and as not having boyfriends. Rape victims in this study were also viewed negatively and believed to have demonstrated behavior that provoked their rape.

Barnett and Feild (1977), using a sample of 200 male and 200 female college students, administered the Attitudes Toward Rape Questionnaire (Feild, 1978a) to measure their view of rape. Results indicated substantial gender differences in attitudes toward rape. A significant number of women viewed rape as an exercise in power over women, whereas a significant number of men viewed rape as being motivated by a desire for sex. Other finding reaching significance included:

- (a) Forty percent of the men, as compared to 18 percent of the women, stated that a woman's degree of resistance should be a major factor in determining if a rape occurred.
- (b) Thirty-two percent of the men and 8 percent of the women stated that it would do some women good to get raped.
- (c) Seventeen percent of the men and 7 percent of the women stated that if a woman is going to be raped, she might as well relax and enjoy it.
- (d) Fifty-nine percent of the men, contrasted with 38 percent of the women, stated that women provoke rape through their appearance or behavior.

College students believing in traditional gender-role stereotypes were found, when compared to students with liberal attitudes, to be more apt to excuse an acquaintance rapist's behavior, to be more likely to see the women as provoking her attack (e.g. wearing seductive clothes, going to the offenders home), and to attribute more blame to women in acquaintance rape situations (Seiverd, 1989). Harrison, Downes, and Williams (1991) and Quackenbush (1987) found that college males regarded acquaintance rape less seriously than stranger rape, expressing greater empathy for stranger rape victims and attributing greater responsibility for the rape to acquaintance rape victims. Quackenbush (1987) concluded

ŧ .;

:e

ţe

that "tantamount to curbing the problem of rape in our society is ... reducing ... generally the acceptance of rape myth" (p. 96).

### Gender-Role Attitudes

A survey of the sexual attitudes of 449 university students by Dull and Giacopassi (1987) found two dimensions of equal importance in understanding attitudes toward date rape. The first dimension measured individual attitudes toward dating, sex, and love. The second dimension measured attitudes concerning rape and male versus female responsibility for rape. This research also discovered strong evidence that gender was the most important variable in determining the formation of these attitudes, with males holding stronger attitudes than females that condoned sexual aggression.

Quackenbush (1987) conducted research with 90 male university students that supported the premise that gender-role socialization contributes to the development of rape-supportive attitudes. Briere and Malamuth (1983) used the scales developed by Burt (1980) with 352 male college students and determined "... that the antecedents to rape are cultural, socially transmitted attitudes about women, rape, and rapists which are stereotyped and prejudicial, and which serve as psychological releasers for sexual aggression" (p. 321).

From a study of 71 self-disclosed date rapists, Kanin (1984) concluded that these men were not sexually deprived, sexually inadequate, or socially and psychologically maladjusted. Most of these rapists were successfully involved in dating and sexual activity but were shown to be products of highly erotic peer group socialization (see also Malamuth, Haber, and Feshbach, 1980). Kanin (1967) also determined from a study of 341 male students that sexually aggressive behavior doesn't confer prestige, but it is supported by other males when viewed as a response to females who don't follow expected gender roles——thus resulting in a form of social control.

Both Garrett-Gooding and Senter (1987) and Horsley (1988) found

• **.**  that men have more traditional gender-role conceptions than females. However, traditional students of both genders were more likely to accept forced sex in certain situations as normative behavior (Fisher, 1986). Muchlenhard and Hollabaugh (1988) determined that women at intermediate levels of traditionality believed it more acceptable for men to use force and that women were expected to give token resistance, thus contributing to the dating problems addressed by Weis and Borges (1973). Additionally, Coller and Resick (1987) discovered that women who strongly believed in gender-role stereotypes were significantly more likely to blame a female for her victimization than were women who minimally believed in gender stereotypes.

McDonald (1988) cited a study of 300 males conducted by Koralewski and Conger at Purdue University that found sexually aggressive men were more likely than average men to accept violence, hold callous sexual attitudes, and believe that women enjoy being raped. These result were similar to those found by Koss and Dinero (1988) and Koss, Leonard, Breezley, and Oros (1985). For example, Koss and Dinero (1988) determined from their national sample of 2,972 college men that highly sexually aggressive males were typified by greater hostility toward women, a perception that force and coercion were legitimate ways to gain sexual compliance, and by involvement in peer groups that reinforced sexualized views of women.

Some people represent rape as a pseudosexual act, but Kanin (1984) stated that "... with respect to the college rapists ... it is probably more appropriate to stress the sexual dimension" (p. 104). Mahoney, Shively, and Traw (1986) studied 337 college students supporting the view "... that male sexual assault behavior is a function of traditional male socialization since the important predictors are opportunity and hypermasculinity" (p. 7). This also reflects, in part, the thinking of Mosher and Anderson (1986) who wrote that "common cultural notions of masculinity, emphasizing power, dominance, strength, toughness, competiveness, and aggression, may transform the sexual arena into one

ı

•

•

••

in which rape expresses the hypermasculine need for power, dominance, conquest and control\* (pp. 77-78).

### Likelihood of Raping

A review of the literature revealed that an average of 35 percent of the males studied indicated some likelihood of raping a female (Malamuth, 1981). Specifically, Koss (1988) found that 47 percent of self-identified male college student rapists stated that they expected to engage in similar behavior in the future. A study conducted by Check and Malamuth (1983) involving 289 university students found that 30 percent of the men indicated some likelihood of raping after reading a story about a rape. This result was consistent with additional research by Malamuth, Haber, and Feshback (1980) that found over one-half of 53 male college students would not rule out the possibility of engaging in rape if they could not be caught. Twenty-eight percent of 352 male students also indicated they might rape or use force if assured they wouldn't be caught (Briere and Malamuth, 1983).

Research involving 114 males at SUNY-Albany indicated that a male's likelihood of raping was associated with aggressive tendencies and with rape myth acceptance (Greendlinger and Byrne, 1987). Other research (Check and Malamuth, 1983; Malamuth, 1981; Malamuth and Check, 1980; Malamuth, Haber, and Feshback, 1980) found that males with strong gender-role stereotyping beliefs reported a greater likelihood of raping, believed in rape myths, and were affected by exposure to erotic stories.

### Attribution of Responsibility

## Theoretical Overview

A fundamental assumption of attribution theory is that people (perceivers) attempt to identify the causes of observed behaviors in order to find meaning in their social world. These behavioral causes

•

• ,

•

are linked to either the dispositional qualities (e.g. temperament, attitudes) of an actor that assumes the behavior was freely chosen or to environmental factors that assumes situational exigencies beyond the actors control influenced the behavior. Attributions may be derived from either first-hand observation or secondary reports, with both processes subject to error given the complexity of social stimuli that often results in the use of stereotypes as a way to categorize information (Shaver, 1975).

Heider (1958) theorized that ordinary or naive people make dispositional attributions based on the use of personal force that is defined by two principal factors: (a) power that is represented by constructs such as ability, temperament, knowledge, skill, and attitudes; and (b) motivation that represents constructs such as intention and exertion. The combination of ability, exertion, and task difficulty represents an action sequence that helps a perceiver determine if a behavior was based on dispositional or environmental considerations.

Jones and Davis (1965) speculated that a behavioral act reflects a personal choice on the part of an actor that has multiple effects on the environment or the actor. This emphasis states that an actor has choices between courses of action, allowing a perceiver to diagnose the intentions of the actor based on the perceiver's understanding of the desirability of the action. The assumptions made by the perceiver concerning desirability operates as a hypothesis that biases the inference process. "Thus, cultural assumptions or social stereotypes may obscure the true significance of an action" (p. 227) because of "the hedonic relevance of an effect ... for the perceiver ..." (p. 237).

Kelley (1973) stated that individuals often lack the time and motivation to make multiple observations of an event, relying instead on a single observation. The perceiver derives information from this event using a three-dimensional model that features an interaction between time/modality (context), persons (perceiver), and entities (stimulus).

• 1 2 £.

Ė

ref

a g

tha

a g

ra:

Also important to this theory is the role of a causal schema that "...
refers to the way a person thinks about plausible causes in relation to
a given effect" (p. 114). Specifically, a schema consists of existing
abstract ideas that provides a framework for new bits of information
that are analyzed within existing data patterns. Thus, "... the role of
a given cause in producing a given effect is discounted if other
plausible causes are also present" (p. 113).

Attribution theory is not concerned with the true cause of an event, rather it focuses on understanding how people attribute causation. For example, Walster (1966, 1967) stated that people need to live in an orderly world that is predictable and controllable. As such, they assign causal responsibility for an incident to something other than chance, believing they have the ability to avert the same type of disaster from happening to them (see also Shaw and Skolnick, 1971). Shaver (1970) suggested that a defensive attribution process operates, with less responsibility assigned to similar people and conversely more responsibility assigned to circumstances. Blame was more easily assigned when the stimulus person was different from the perceiver (see also Bradley, 1978). Lerner (1965, 1970, 1980) postulated that people believe in a just world where positive outcomes happen to good people and negative outcomes happen to bad people. Perceivers judge both a person's character and behavior to determine what levels of blame and responsibility to assign.

# Attributional Factors

Wong and Weiner (1981) conducted 5 self-probe experiments and concluded that "... we have strong evidence that [the] attributional search [process] primarily is focused on the locus and control dimensions of causality" (pp. 661-662). Kaplan and Miller (1987) and Phares and Wilson (1972) found that the greater the ambiguity surrounding the facts, the more difficult it was to attribute responsibility for an incident. Klemmack and Klemmack (1976)

. • *:*• • : **:** 

determined that people were willing to define a situation as rape when there were clear indications that the victim had no control over the event. Furthermore, when the situation was not congruent with rape stereotypes, the study's respondents looked for other interpretations that reflected the view that women are to blame for their own rape.

Force and Resistance: One interpretation that people used to judge the occurrence of rape was the amount of force used by the male and the level of resistance demonstrated by the female. A study by Krulewitz and Payne (1978) involving 118 female and 114 male college students determined that an assault was significantly more likely defined as rape when the level of force used by the male increased. Subjects generally blamed the male assailant more and attributed less responsibility and more respectability to the female victim when force was used. Shotland and Goodstein (1983), Skelton and Burkhart (1980), and Smith (1983) found very similar results concerning the use of force. A review of 124 rape cases also indicated that greater levels of violence increased a rape victim's credibility with criminal justice personnel (LaFree, 1980).

A study of 287 college students by Shotland and Goodstein (1983) discovered they were more likely to believe a rape occurred when the woman protested the male's sexual advances early in the encounter, using both verbal and physical means. Krulewitz and Nash (1979) found that college students (117 females, 112 males) believed a sexual assault was less likely to occur when a woman did not offer obvious physical resistance. An analysis of rape trials determined that alleged lack of resistance was commonly used as a defense strategy (Holmstrom and Burgess, 1975). Schultz and DeSavage (1975) found that "... after a certain stage in sexual intimacy a negotiation is reached and the female forfeits her right to change her mind or have second thoughts" (p. 81).

Howard (1984a, 1984b) conducted two studies involving 295 college students. The research discovered that greater attribution of blame for sexually assaulted females was made according to characteristics that

•. • • • • : • • • 1.3

**3**..
.

· ·

conform to the gender-role stereotypes for women. For example, women who demonstrated stereotypical behavior associated with submissiveness received more blame for their rape because of their perceived failure to resist. Conversely, increased attribution of blame for sexually assaulted men occurred when the behavior demonstrated during the assault contradicted male gender-role stereotypes, meaning they showed fear or failed to fight.

Attractiveness: Juries make decisions in criminal cases that are influenced by the attractiveness of the defendant (Sigall and Ostrove, 1975). Specifically, physical attractiveness influences criminal sentencing (Barnett and Feild, 1978; Reynolds and Sanders, 1975) and judgements concerning guilt (Efran, 1974). Physically attractive women were perceived as more sexually alluring (Dermer and Thiel, 1975), promiscuous (Thornton, 1977), and naive to the affect they have on men (Mazelan, 1980). Women perceived as unattractive were believed to provoke their victimization more than attractive women (Seligman, Brickman, and Koulack, 1977; Tieger, 1981).

Respectability: Women are accorded respect based on sexual characteristics, the most common being virginity (Barber, 1974; Kanekar and Vaz, 1983; Mazelan, 1980). Jones and Aronson (1973) conducted research that found female virgins were rated as more respectable than married or divorced women. This research further found that the more respected women were attributed the greatest level of blame for their rape. This finding was consistent with the Just World Theory (Lerner, 1965) that postulated when a negative event happened to a good person, the good person must have somehow contributed to the negative circumstances.

In contrast, Smith, Keating, Hester, and Mitchell (1976) found that the most respected person (a nun) was attributed significantly less responsibility for the rape. Cann, Calhoun, and Selby (1979), Feldman-Summers and Lindner (1976), Kanekar and Kolsawalla (1977), Kerr and Kurtz (1977), and McGraw (1987) found similar results, suggesting that

•

··

† ·

K.

women perceived as less respectable according to characteristics such as occupation, marital status, and/or sexual history received greater attribution of blame for their victimization.

Kahn, et al. (1977) also failed to replicate Jones and Aronson (1973), concluding that "... it appears likely that rape victims vary in a number of dimensions besides respectability, and these other dimensions may be more important in making attributions of fault. For example, subjects may believe the various victims differ in physical appearance, reason for walking alone at night, mode of resistance to the attacker, and so forth" (p. 106). Fulero and Delara (1976) also state that "... the process may be sex-dependent, with female perceivers affected by the victim's similarity to themselves, and male perceivers affected by the victim's respectability" (p. 557).

Acquaintanceship: McGraw (1987) found that people attributed significantly greater responsibility to the rapist when the assailant and victim were strangers. A review of criminal cases in Chicago determined when a rape victim and assailant were acquainted, that criminal justice personnel focused on the party's states of mind (Kerstetter, 1990). LaFree (1980) and Wilson (1978) found that acquaintance rapes were taken less seriously than stranger rapes by police and prosecutors. Chappell and Singer (1977) reviewed rape complaints in New York City and found that police believed acquainted couples only misinterpreted one another, thus believing that a real rape did not occur.

While "a prior relationship makes arrest easier ..., [it] ...
reduces the strength of the prosecutor's case by allowing a consent
defense" (LaFree, 1981, p. 589). Bohmer (1974) found judges less
sympathetic to acquaintance rape victims because they were perceived as
precipitating their own rape. Estrich (1987) stated that acquaintance
rape cases are downgraded by the legal system because they are viewed as
private disputes, may involve a claim of rights, may involve
contributory fault, and are viewed as less terrifying than stranger

• • • S -. ٠, .• • .• ? ÷

• •

rape

Ge:

rol

of

res

(19

ja

## Gender Differences

Thornton, Ryckman, and Robbins (1982) found that gender and genderrole attitudes were the two primary contributors concerning the amount
of perceived victim culpability. Gender differences were evident in
research conducted by Feldman-Summers and Lindner (1976), Scroggs
(1976), Stephan (1974), and Tolor (1978) that discovered the length of
jail sentences and the determination of guilt were different for males
and females judging criminal cases. Males viewed rape less seriously as
a crime than females, with females having a more sympathetic view of
rape victims than males (Tieger, 1981). Males also perceived rape as
more of a sexual act, thus attributing it to sexual causes (Bridges and
McGrail, 1989).

Males emphasize the personal characteristics of rape victims to a significantly greater degree than females (Calhoun, Selby, and Warring, 1976). Smith, Keating, Hester, and Mitchell (1976) found that females were more punitive toward the assailant than were males, while males judged rape victims as significantly more likely to precipitate their own rape. Additional research by Cann, Calhoun, and Selby (1979), Kanekar and Vaz (1983), and Selby, Calhoun, and Brock (1977), determined that women, as compared to men, perceived rape victims as having less responsibility for their own rape. Krulewitz and Nash (1979) found that women attributed greater responsibility to the victim of a stranger versus an acquaintance rape, while Tetreault and Barnett (1987) found just the opposite result.

Research has suggested that gender-role attitudes are the most important determinant concerning attribution of blame. For example, Williams (1979) found significant support for a link between gender-role attitudes and assessment of female blame for rape. This conclusion has been supported by the work of Acock and Ireland (1983), Feild (1978a, 1978b), Feild and Bienen (1980), Howard (1984a), Klemmack and Klemmack

(1976), and Shotland and Goodstein (1983). Miller, Smith, Ferree, and Taylor (1976) stated that "reactions to victims are not based on a psychological need to perceive the world as just, but are rather dependent upon the political and social context in which the perceiver, the victim, and the victimizer are embedded" (p. 358).

# Jury Decision-Making

"In the United States the criminal justice system perceives the crime of rape as an act against society and thus attempts to redress its social wrong " (Brodyaga, Gates, Singer, Tucker, and White, 1975, p. 113). When redress was delegated to a jury, however, it was common for the jury to ignore the strict application of the law and respond to the circumstances of each case according to the general stereotypes of society (Brooks and Doob, 1975; DiPerna, 1984; Frazier and Borgida, 1988; Guinther, 1988; L'Armand and Pepitone, 1982; Williams, 1976; see especially Feild and Bienen, 1980). The result may be an arbitrary hearing (Frank, 1949; Griffitt and Jackson, 1973; Simon, 1980; Stephan, 1975), where the rape victim and not the assailant was the person placed on trial (Borgida and White, 1978; Russell, 1975). For example, a study involving 360 rape trail jurors determined that the victim's gender-role behavior (e. g. alcohol/drug use, acquaintance with the assailant) influenced verdicts more than the evidence in cases that disputed the occurrence of a sexual assault (LaFree, Reskin, and Visher, 1985).

Langer, Blank, and Chanowitz (1978) demonstrated that people ignore relevant information when a situation was consistent with their prior expectations. Wrightsman, Kassin, and Willis (1987, p. 139) stated that "... jurors may very well enter the courtroom biased in their predisposition. If the evidence is mixed or ambiguous, jurors' predispositions could affect their decisions, "especially in rape cases (see also Gerbasi, Zuckerman, and Reis, 1977; Reskin and Visher, 1986). In fact, jurors often make their decisions before deliberations (Weld

in

.

•
.

and Danzig, 1968; Wishman, 1986). Vidmar (1972) also demonstrated that a jury weighs the consequences of a decision before dispensing individual justice.

In their landmark study of juries, Kalven and Zeisel (1966) determined that juries have a tendency "... to weigh the conduct of the victim in judging the guilt of the defendant" (p. 243). Juries often looked at an assumption of risk theme to determine the level of blame the woman shared for the alleged rape. In other words, the juries did not limit themselves to the central issue of consent. This study found that juries were four times as willing to convict in aggravated (stranger) rape than in simple (acquaintance) rape cases.

## Campus Disciplinary Procedures

Much has been written about the changing relationship between institutions of higher education and their students, particularly in terms of liability issues (Batson, 1986; Bhirdo, 1989; Bohmer and Parrot, 1993; Dodd, 1985; Fishbein, 1985; Helms, 1987; Kaplin, 1985; Redford, 1989; Reidhaar, 1985; Stamatakos, 1990; Szablewicz and Gibbs, 1987). The doctrine of "in loco parentis" was rendered inoperative in the 1960s and was replaced by a variety of other legal models, including constitutional, contractual, fiduciary, and unitary models (Stamatakos, 1990). Szablewicz and Gibbs (1987) argued that the doctrine of "in loco parentis" was experiencing a revitalization. However, other legal scholars (Batson, 1986; Bhirdo, 1989; Dodd, 1985; Stamatakos, 1990) argued that liability issues were more fully defined by the duties and responsibilities inherent in the special relationship between the student and college, which included the issue of "foreseeability."

Helms (1987) found that tort, statutory, property and procedural cases comprised the recent growth in higher education case law. Much of the case law cited by all the aforementioned authors largely dealt with situations of violence (e.g. stranger rape, murder, assault, kidnapping)

• • • • . •

that generally involved issues concerning a "breach of duty" and "foreseeability" on the part of the college. Courts have also looked at whether it was reasonable for a student to have expected protection from harm when judging liability issues (Steiner, 1989).

Law suits filed against colleges in acquaintance rape situations have been based partly on staff and procedural response issues (Bohmer and Parrot, 1993). For example, a former student at Central Michigan University sued the school claiming it failed to properly adjudicate her rape complaint against a student athlete (Associated Press, 1991). At Carleton College 4 women filed suit because the institution allegedly allowed men with known histories of sexual harassment to remain on campus (Gibbs, 1991). A former student at Colgate filed a negligence suit because University officials allegedly failed to investigate her rape case (Freeman, 1990). Similiar allegations have also been made by female students at George Mason, Arizona State, and Ohio State Universities (National Association, 1993). According to Wagner (1991, p. B2), "... if university authorities fail to act quickly enough to remove a student who is a known threat, the victim may suffer retaliation or other students may be injured in the meantime", thus indicating the importance of a thorough institutional response.

To properly protect female students and minimize liability, Batson (1986) stated that colleges must have "strong and well-publicized policies and judicial procedures to deal with gang and individual rapists" (p. 128; see also Bohmer and Parrot, 1993). Pavela (1992b) wants higher education to set clear standards for student behavior and acknowledge that ultimate authority for discipline resides with the campus administration. Stoner and Cerminara (1990) provided a comprehensive model for a student disciplinary code that followed due process requirements (Blaskey, 1988; Latourette and King, 1988; Picozzi, 1987; Swem, 1987) and represented generally the prevailing law. Their model stressed the importance of judicial advisors conducting investigations and the need for having hearing boards composed of

**.** 2

i

.

•

•

·

.

189

faculty, staff, and students impartially judging cases (Adams and Abarbanel, 1988 and Parrot, 1991a and 1991b also provided recommendations concerning judicial hearing procedures for sexual assault cases).

The standard of proof necessary in college discipline cases is less than the "beyond a reasonable doubt" standard used in criminal justice proceedings. Campus discipline hearings may use less stringent "by a preponderance" or "clear and convincing" standards of proof to judge student behavior that violates college rules (Bohmer and Parrot, 1993; Long, 1985). It is not considered "double jeopardy" to try a student in both criminal and university judicial processes and it is not unusual for the 2 processes to rule differently on the matter of guilt (Adams and Abarbanel, 1988; Parrot, 1991b).

Parrot (1991b) stated that judicial hearing boards should have a legal expert appointed as the chairperson and that board members should receive special training to deal with acquaintance rape cases. The training should address the laws and policies related to acquaintance rape, the psychological needs of victims and defendants, and the typical acquaintance rape patterns (Parrot 1991a). Few college hearing boards are well-prepared to hear acquaintance rape cases because most have only been trained to adjudicate vandalism and plagiarism incidents (Collison, 1991; Julian, 1992; Kalette, 1988). Although Lancaster, Cooper, and Harman (1993) found that student discipline is emerging as a profession, research conducted by Dannells (1991) concluded that judicial affairs professionals are handling the changing types of student misconduct cases with the same methods used 10 years ago.

In conclusion, Parrot (1991b, p. 379) stated that "acquaintance rape and sexual assault cases on campus will not stop until institutions of higher education take this problem seriously. The first step toward its elimination is the development of a campus policy on acquaintance rape and sexual assault, with related sanctions made clear to students.... When a case is reported, the victim must receive support

•:

.

..

. 4.

.

•.

, a

> . \* •

3 · ·

· £

and, if the allegations are proven, the assailant should be dealt with to the fullest extent permitted by the campus policy." In fact, the Higher Education Reauthorization Act of 1992 requires institutions to develop both prevention programs and procedural responses to deal with sexual assaults (National Association, 1992; Gehring, 1993).

### Summary

The aforementioned literature review demonstrated the prevalence of acquaintance rape in the United States. Acquaintance rape was especially apparent among college age individuals, with female samples reporting rape victimization rates averaging around 20 percent. The literature supported the feminist contention that acquaintance rape occurs because both genders are socialized to view sexual aggression by males as normal. This was especially true in dating relationships where people interacted according to gender-role stereotypes.

Research demonstrated that people's views of rape were influenced by their level of traditionality concerning the roles of males and females in society. Specifically, people who held traditional gender-role beliefs were more likely to believe in rape myths and thus viewed rape as justifiable under certain circumstances. Their level of traditionality also influenced people's perceptions of sexual violence and individual responsibility for acquaintance rape prevention.

The level of gender-role traditionality, furthermore, influenced the amount of blame attributed to the people involved in acquaintance rape situations. People with more traditional beliefs blamed females more for the rape, especially when they violated gender-role expectations related to, for example, their level of resistance, use of alcohol, previous sexual experiences, and consent. Research also demonstrated that people rely on rape stereotypes to judge ambiguous acquaintance rape situations, that are often not as clear as stereotypical stranger rape situations.

•		
<b>1</b>		
•		
•		
i		

Research with both real and mock juries demonstrated that verdicts in rape cases did not always reflect the strict application of the law. Individual jurors often arrived at decisions using rape myths as the basis, that in affect judged the victim and not the assailant. Furthermore, most of these studies used college students as subjects, showing that belief in rape myths and gender-role stereotypes influenced their judgement in acquaintance rape situations.

Studies demonstrated that acquaintance rape is a serious social problem on college campuses. However, many question the commitment of higher education to properly respond to the problem. This concern was reflected in the Higher Education Amendments of 1992 that required all higher education institutions receiving federal funds to develop policies and procedures aimed at both preventing and responding to sexual assault occurrences (NASPA Forum, 1992). Previously, two bills were introduced into the U. S. House of Representatives to ensure that colleges provided assistance to and protection for sexual assault victims (Freeman, 1991; NASPA Forum, 1991; see also Bohmer and Parrot, 1993 for an excellent legislative overview).

More recently, the Michigan House of Representatives passed a campus sexual assault victims' bill of rights that required institutions to develop comprehensive sexual assault policies (Michigan Campus Sexual Assault, 1992). The Michigan bill sought to guarantee the rights of sexual assault victims by requiring that victims be treated with dignity (not blamed for their assault), have access to counseling services, have an opportunity to report the assault to criminal justice authorities, and receive the full support and cooperation of institutional personnel. The bill further required that institutional disciplinary proceedings accord the victim the same rights as the accused, that included the right to council and the right to be notified of the outcomes.

The scope of the problem, the increased general awareness, the potential for increased reporting, and the possibility of legislative mandates highlight the importance of colleges developing comprehensive

imp adj

acq

ju sa

sp

at

pr

.

•

•

.

**2.**7

acquaintance rape policies and procedures. This may be especially important in the campus judicial affairs arena where effective adjudication requires special awareness of rape laws, understanding of acquaintance rape dynamics, and sensitivity to the involved parties.

This author was unable to find any published research related to how actual college judicial boards make decisions in acquaintance rape cases. Thus it appears that research into the criteria used by judicial boards to determine guilt or innocence and subsequent sanctioning is a valuable and missing part of the literature. Of special importance is the influence of an individual board member's attitudes toward gender roles and rape on his/her attributional processes. This type of research may assist colleges, especially judicial affairs functions, in effectively adjudicating acquaintance rape cases. This research has special implications related to the selection, training, and advising of judicial boards.

·

•

.

•

i

•

•

-

•

.

\*\*

a:

198

per

Fi th th

> do tı

> i b

1

### CHAPTER THREE

#### METHODOLOGY

### Introduction

Statement of the Problem: Heterosexual acquaintance rape is a social problem on college campuses, with approximately 27 percent of college women experiencing rape while on a date (Miller and Marshall, 1987). When reported, colleges often respond by adjudicating alleged perpetrators in acquaintance rape cases via institutional student judicial processes. These processes often include the option of having peer judicial boards hear the facts of the case, determine a verdict, and render discipline sanctions when applicable.

Critics charge that judicial boards are ill-prepared to adjudicate acquaintance rape cases because of inadequate training and lack of sensitivity concerning the dynamics of acquaintance rape situations.

Furthermore, research involving both actual and mock juries demonstrated that verdicts in acquaintance rape cases were often influenced more by the individual member's levels of belief in rape myths and gender-role stereotypes than by actual facts.

Purpose of the Study: The review of the literature clearly documented that college students, both male and female, believed in traditional gender-role stereotypes (Rosenkrantz, et al., 1968) that influenced dating relationships (Muehlenhard, Friedman, and Thomas, 1985) and perceptions concerning a willingness to engage in sexual intercourse (Bostwick, 1990). College students were also found to believe in rape myths (Check and Malamuth, 1985) that influenced the definitional boundaries between rape and normal sexual activity (Wilson, Faison, and Britton, 1983). Furthermore, rape myths were found to be related to views concerning the roles and rights of women in society (Costin, 1985), with students holding traditional views more likely to believe in rape myths and to accept sexual aggression (Greendlinger and Byrne, 1987).

A review of the literature on attribution demonstrated that people

•
.

•

1

defined a situation as rape according to their stereotypes of rape (Klemmack and Klemmack, 1976). Further research determined that gender and gender-role attitudes were the 2 primary contributors concerning the definition of rape (Thornton, Ryckman, and Robbins, 1982), with males viewing the rape less seriously than females (Calhoun, Selby, and Warring, 1976; Tieger, 1981).

Research involving both actual and mock juries demonstrated that verdicts in acquaintance rape cases were often influenced more by the individual member's level of belief in rape myths and gender-role attitudes than by actual facts (LaFree, Reskin, and Visher, 1985). In essence, jury members reflected the stereotypes of society. Thus, an important question was raised. Specifically, given the documented level of rape myth belief and traditional gender-role attitudes in the general college population, were college judicial board members more aware of the influence of rape myths and gender-role attitudes on their judgement of blame and responsibility in rape cases?

This was an important question given that judicial board members are trained to impartially judge all cases heard before them. If individual members subscribe to the same beliefs as their peers in terms of rape myth beliefs, Parrot (1991a, 1991b) and other critics may be correct in questioning the competence of judicial boards to adjudicate acquaintance rape cases on campus. Thus, this study was designed to answer the aforementioned question. Specifically, the purpose of this study was to determine what effect college judicial board members' levels of rape myth belief and gender-role attitudes had on the judgement of a male perpetrator's guilt or innocence for committing an act of rape. The rape was depicted by 2 written scenarios featuring an acquaintance assault and a stranger assault. The study specifically examined the differences between the acquaintance rape and stranger rape situations consistent with the purpose of this study.

Each of the two rape scenarios were judged by 50 percent of the subjects. Thus, 50 percent of the judicial board members judged the

acquaintance rape scenario. The remaining 50 percent from each sample judged the stranger rape scenario. All subjects completed the FEM Scale that measured attitudes toward feminism and sexism, the Attitudes Toward Rape Questionnaire, the Adjudication Questionnaire, and the Demographic Information Form.

Research Questions: This investigation was guided by the following questions:

- What relationship, if any, exists between the verdicts of guilt or innocence for the male perpetrator and rape myth beliefs across acquaintance rape and stranger rape modes?
- What relationship, if any, exists between the verdicts of guilt or innocence for the male perpetrator and gender-role attitudes across acquaintance rape and stranger rape modes?
- 3. What relationship, if any, exists between gender-role attitudes and rape myth belief on verdicts of guilt or innocence for the male perpetrator across acquaintance rape and stranger rape modes?

<u>Hypotheses</u>: The following directional hypotheses were tested in this study:

- As a group, subjects with high level belief in rape myths will most frequently select the not guilty verdict in the acquaintance rape mode.
- 2. As a group, subjects with low level belief in rape myths will most frequently select the guilty verdict in the acquaintance rape mode.
- 3. As a group, subjects with high level belief in rape myths will most frequently select the guilty verdict in the stranger rape mode.
- 4. As a group, subjects with low level belief in rape myths will most frequently select the guilty verdict in the stranger rape mode.
- 5. As a group, subjects with sexist beliefs will most frequently

••• . • : ; A. . ¥ i\* •

.

**T**;

select the not guilty verdict in the acquaintance rape mode.

- 6. As a group, subjects with feminist beliefs will most frequently select the guilty verdict in the acquaintance rape mode.
- 7. As a group, subjects with sexist beliefs will most frequently select the guilty verdict in the stranger rape mode.
- 8. As a group, subjects with feminist beliefs will most frequently select the guilty verdict in the stranger rape mode.
- 9. Overall, there will be a positive relationship between genderrole attitudes and rape myth belief.
- 10. Overall, gender-role attitudes will be more influential than rape myth belief in determining the verdict in the acquaintance rape mode.
- 11. Overall, gender-role attitudes will be more influential than rape myth belief in determining the verdict in the stranger rape mode.

The remainder of this chapter provides an overview of the methodology used in this investigation. Specifically addressed are the topics concerned with the subjects, instrumentation, data collection procedures, data scoring, and statistical analysis of the data.

## Subjects

Many colleges have a primary judicial board that adjudicates enrolled students who violate the more serious conduct rules (e.g. rape) of the institution (Lancaster, Cooper, and Harman, 1993). These boards are typically composed of enrolled students, faculty, and staff members. They are empowered by the institution to levy or recommend disciplinary sanctions, including suspension, when they find a student guilty, after a hearing, of violating the college's conduct rules. These boards are almost always advised by student affairs staff members who serve in some capacity as a judicial affairs officer for the college. Judicial board members are typically selected on an annual basis and receive training

. `e 1. . . **.** • ٠.

4:

19,2

•

to

in

of

tha

to assist in the performance of their duties.

The chief student affairs offices at 13 state-assisted institutions in Michigan were initially contacted via phone in November and December of 1992. The purpose was to determine if they had a judicial system that included a board empowered to hear cases that could result in suspension. Of the 13, only 10 of the institutions were deemed as having the requisites necessary for this study. The remaining 3 state-assisted institutions received no further consideration. The University of Michigan-Dearborn and The University of Michigan-Flint were never contacted during this investigation, as they were viewed as extensions of The University of Michigan-Ann Arbor that does not have a student code of conduct.

The chief student affairs offices at 7 independent institutions, all members of the Michigan Intercollegiate Athletic Association, were initially contacted via phone to assess the structure of their judicial systems. Only 6 of the colleges received further consideration for involvement in this study.

A second phone call was made in late February and early March of 1993 to the staff member contact at each of the colleges deemed as having the requisite judicial structure. The purpose was to discuss the specifics of this study and to seek their assistance and cooperation.

As a result, 6 independent and 9 state-assisted institutions agreed to cooperate with this research project.

Only enrolled students that served on judicial boards empowered to suspend others were selected for this study. In some cases, these boards only recommended suspension to the college's primary judicial affairs officer who made the final decision. The reasons for selecting this sample were: (1) Rape cases are typically viewed as serious and thus are sent to primary judicial boards because suspension is a possible outcome; (2) Primary judicial boards typically adjudicate more serious cases and are more experienced; (3) Primary judicial boards are generally advised by more experienced, higher level student affairs

· 1 • : . • • . • •

.\*

•

sta

are

rol

staff members; (4) Students often account for a majority on these boards or are at least represented equally with faculty and staff members; and (5) At institutions where the boards only recommend suspension, the recommendations are usually supported by the judicial affairs officer. Although colleges may have other judicial boards, they are generally not empowered to suspend students because they hear only routine violations of the rules (e.g. alcohol possession).

The judicial board members included in this study were identified by a student affairs staff member with judicial responsibilities at each of the respective institutions. Depending upon the institution, the number of enrolled students selected and trained to serve on primary judicial boards ranged from 3 to 20 people. Thus, the investigation was extended to both independent and state-assisted colleges to generate a sample size large enough to have each of the two written rape scenarios judged by at least 30 different judicial board members. According to Borg and Gall (1983) and Scheaffer, Mendenhall, and Ott (1990), a sample size of 30 closely approximates a normal distribution and is preferable.

originally, this investigation included 118 judicial board members enrolled as students for the spring semester of 1993 at 6 independent and 9 state-assisted colleges within the state of Michigan. Eighty-nine of the original judicial board members were selected from the state-assisted colleges, with the remaining 29 original board members representing the independent institutions. However, 5 of the survey packets were never distributed, resulting in a sample of 113 board members. This final sample was composed of 85 board members from state-assisted institutions and 28 board members from independent colleges.

Eighty-two judicial board members completed and returned the survey packets. This resulted in a 73 percent overall return rate. Fifty-four percent (n = 15) of the subjects at the independent institutions returned survey packets. Judicial board members at the state-assisted institutions returned 67 survey packets, for a 79 percent return rate.

The subjects in this study represented the following institutions:

### State-Assisted

Central Michigan University
Eastern Michigan University
Ferris State University
Grand Valley State University
Lake Superior State University
Michigan State University
Michigan Technological University
Oakland University
Saginaw Valley State University

### Independent

Adrian College Albion College Alma College Hope College Kalamazoo College Olivet College

## Instrumentation

FEM Scale: Originally developed by Smith, Ferree, and Miller (1975), this instrument was designed to measure attitudes toward feminism. A 57 item scale, later shortened to 48 items, was tested with 39 undergraduates (28 males, 11 females) at an Ivy League college. The researchers determined that the scale had a .92 reliability.

The authors shortened the scale to 27 items and administered it to 100 (52 females, 48 males) Harvard Summer School students. This version of the scale yielded a .90 reliability. The validity of the FEM Scale was also established in this study by finding a strong correlation with 2 scales developed by the authors measuring feminist identification (.629, p<.01) and activism (.392, p<.01). Furthermore, correlations between the FEM Scale and the Rubin-Peplau Just World Scale yielded a significant negative correlation of -.238 (p< .05), that was predicted because feminist beliefs are not compatible with belief in a just and fair society. Correlations with the Rotter Internality-Externality Scale that measured political efficacy resulted in a non-significant correlation of .048, that was predicted. The differences between the correlations for males and females on the 4 measures and the FEM Scale were not significant.

A factor analysis shortened the scale to 20 items. The final version of the FEM Scale demonstrated a .91 reliability.

Singleton and Christiansen (1977) conducted an independent validation of the FEM Scale by administering the instrument to 149

california. These researchers determined that the 20-item scale had a .91 reliability. Furthermore, they found the scale to have convergent validity, establishing a significant .638 (p<.001) correlation with a women's movement identification scale designed by the authors to measure support for feminist activities and involvement. A significant negative correlation with a dogmatism scale (-.506, p<.001) developed by the authors was predicted and verified. Finally, the authors found a negative correlation between the FEM Scale and an anti-black prejudice scale (-.462, p<.001) that they developed to measure a generalized tendency to stereotype. No significant differences between separate correlations for males and females and the 3 scales existed, but males were found to be more sexist and racist than females.

They also established the discriminant validity of the scale by giving the scale to 88 feminists (mean = 91.30 ) and 59 anti-feminists (mean = 51.03), finding a significant (p<.001) difference in the mean scores of the 2 groups. Female students (n = 149, mean = 77.24) and male students (n = 129, mean = 66.16) from the original sample scored in between the feminist and anti-feminist groups. In conclusion, Singleton and Christiansen (1977, p. 301) found the FEM Scale to be "... the shortest, most reliable measure of sexist or feminist attitudes yet available ..."

The FEM Scale was selected for this investigation because of the thorough validation that occurred, both by the initial developers as well as by independent investigators. The other primary instrument noted in the literature for measuring attitudes toward gender roles was the Attitudes Toward Woman Scale (Spence and Helmreich, 1972). The ATW Scale was not selected, in part, because a researcher at Kansas State University (personal communication, November, 1992) questioned the validity of the instrument as it related to her research on rape attitudes. This investigator also believed that the FEM Scale demonstrated better face validity than the ATW Scale (See Appendix A or

.

۲,

:

. ,

: .

+;

•

rev

cha

3,

att

raç of

th

cc

81

Q

,

1

B, Section D, for the complete FEM Scale).

Attitudes Toward Rape Questionnaire: Feild (1978a), believing that attitudes toward rape were multidimensional rather than unidimensional, reviewed both the scholarly and popular literature for statements characterizing people's attitudes toward: (a) the act of rape; (b) the rape victim; and (c) the rapist. The review resulted in an initial pool of 75 items representing affective, cognitive, and action components that were reduced to a 37-statement ATR Questionnaire.

The ATR Questionnaire was given by Barnett and Feild (1977) to 400 college student (200 men, 200 women). Barnett and Feild used t-tests to compare male and female responses and discovered that 72 percent of the items were significant at either the p<.01 or p<.001 levels, indicating substantial gender differences in attitudes toward rape. Following the administration of the questionnaire, the researchers interviewed selected respondents and modified the questionnaire to include 32 items.

Feild (1978a; see also Feild and Bienen, 1980) administered the ATR Questionnaire to 1,448 subjects comprised of 1056 (528 female, 528 male) adult citizens, 254 police officers, 20 committed rapists, and 118 female rape crisis counselors. A factor analysis was conducted with the citizen, police, and rape counselor samples that indicated the presence of the same 8 factors. The congruence coefficients for the factor structures among the 3 sub-groups ranged from .81 to .89 with a mean of .87. When all the subjects were combined and their responses analyzed using varimax rotation, the 8 factors accounted for 50 percent of the variance among the items. According to the researchers, all 8 factors possessed an adequate number of variables with loadings of sufficient magnitude (±.30) to warrant interpretation. The factors identified were:

- Factor 1: Woman's responsibility in rape prevention.
- Factor 2: Sex as motivation for rape.
- Factor 3: Severe punishment for rape.
- Factor 4: Victim precipitation of rape.

₹

•

; •

.

• 1

4

±.♥

.

9

i

- Factor 5: Normality of rapists.
- Factor 6: Power as motivation for rape.
- Factor 7: Favorable perception of a woman after rape.
- Factor 8: Resistance as woman's role during rape.

No data were collected at a second point to determine the reliability of the ATR factors. However, the researchers stated that internal consistency of the dimensions of rape were supported because separate factor analysis showed a high degree of similarity across subgroups and because the square root of the estimated commonalities of the factors (the theoretical lower bound of reliability) had a mean value of .62. The researchers acknowledged this low value but found it acceptable given the heterogeneity of the items, the number of factors found, and the items with loadings greater than .30.

The discriminant validity of the scale was established by using multivariate and univariate analysis of variance to differentiate the sub-groups within the sample. The MANOVA was significant (F = 44.30, df = 24,4159, p<.0001) indicating overall difference in the ways the 4 groups viewed all 8 dimensions of rape. Significant differences between groups were also evident in Factor 1, with an F-test score of 26.81 (p<.0001).

Factor 1 was defined by items emphasizing a woman's responsibility in preventing her rape. Individuals scoring high on this factor were viewed as having pro-rape attitudes. A review of the items comprising this factor revealed a listing of common rape myths generally defined in the professional literature (see Burt, 1980; Warshaw, 1988). Koss and Leonard (1984, p. 220) concur, writing that "... this factor could be interpreted as a general rape myth factor."

Within the citizen sample, men tended to believe to a greater extent than women in rape myths as defined by Factor 1 (r = .17, p<.001). A correlation of citizen responses on Factor 1 of the ATR Questionnaire and the Attitudes Toward Women's Scale (Spence and Helmreich, 1972) yielded a significant negative correlation of -.30 (p<

• *:•* : • **.**\*• .001

rape

for

res

var

use

.001), indicating that gender attitudes correlate with attitudes toward rape (See Appendix A or B, Section E, for the complete scale).

Originally, Factor 1 of the ATR Questionnaire was to be utilized for data analysis. However, subsequent investigation prompted this researcher to conduct a new principal components factor analysis with varimax rotations (Babbie, 1990) for the ATR Questionnaire. The results identified 6 latent factors as compared to the 8 differentiated by Feild (1978a). As a result, a new Unified Rape Myth Factor was created for use as the measure of rape myth belief specific to this investigation (See Appendix D for factor presentation; Chapter 4 also provides more detailed information on the creation of the Unified Factor).

A review of the literature determined that the ATR Questionnaire (Feild, 1978a) and the Rape Myth Acceptance Scale (Burt, 1980) were the primary instruments available to measure the level of rape myth belief in subjects. The ATR Questionnaire was selected because: (a) it demonstrated better face validity by incorporating a wider range of rape myths; (b) it used a more comprehensive sample to establish the instrument; and (c) Koss and Leonard (1984) promoted ATR Factor 1 as a measure of rape myth acceptance. Additionally, Burt (1980) incorporated questions into the Rape Myth Acceptance Scale that were differentiated by racial characteristic. This researcher believed the introduction of a racial element compromised the validity of the instrument.

Rape Scenarios: Two written rape scenarios representing acquaintance and stranger rape situations, originally developed by Quackenbush (1987), were modified for the purpose of this investigation. The acquaintance rape scenario was judged by this researcher as representative of acquaintance rape incidents responded to over a 14-year career working with students living in college residence halls. The acquaintance rape scenario was also shown to several full-time staff members in the Department of Residence Life at Michigan State University who viewed it as a typical representation.

Both scenarios were reviewed by the Sexual Assault Safety Program

 Coor

were advi

in 1

don

wor mir dia

ap

•

ा<u>प</u>

· •

Coordinator at Michigan State University. She agreed that the scenarios were typical of a college acquaintance rape situation. Based on her advise, an act of verbal resistance was added to line 2 of paragraph 7 in both scenarios, as she believed a woman would verbally respond (re: don't) earlier than was evident in the original versions.

This researcher also modified line 6 of paragraph 7, deleting the word "passionate" as an adjective describing a kiss. Other changes were minor; reflecting style of clothes, name of the movie, use of a compact disc rather than a record, and location (on-campus suite v. off-campus apartment).

The acquaintance rape and stranger rape scenarios provided identical demographic and situational information through the first 7 paragraphs. The remaining paragraphs of the scenarios were identical in terms of the female's response and the male's actions, the only difference being the stranger or acquainted condition of the alleged rapist. Both scenarios met the legal requirements within the State of Michigan concerning the commission of a rape by the male (See Appendix A, Section B, for the acquaintance rape scenario. See Appendix B, Section B, for the stranger rape scenario).

Adjudication Questionnaire: The adjudication questionnaire was developed by this researcher to measure the severity of sanction or punishment given to the rapist when viewed as guilty. The questionnaire was composed of 2 questions: (a) Question 1 asked the respondents to determine, after reading a scenario, whether in their opinion the man was guilty or not guilty of rape; (b) Question 2 asked the respondents to select the most appropriate, in their opinion, disciplinary sanction when they believed the man was guilty of rape.

The descriptions used to explain the meaning of the disciplinary sanctions listed in Question 2 were derived from written judicial materials obtained from 1 Michigan independent college and 4 state-assisted Michigan colleges. The researcher also relied on 14-years of experience working with judicial systems at 4 state-assisted colleges as

·

. .

•

a hearing officer and judicial board advisor. Furthermore, the descriptions were very consistent with those provided by Stoner and Cerminara's (1990) model disciplinary code (See Appendix A or B, Section C).

Demographic Information Form: The Demographic Information Form was developed by this investigator to gather personal data about each respondent. This instrument asked respondents to provide information concerning their gender, age, race, spiritual faith, type of college attended, academic class standing, major, and grade point average. This form also asked for information concerning previous experiences with a rape victim, both personally and within the context of hearing a case as part of judicial responsibilities.

Requested information concerning spiritual faith was included because several of the independent colleges in the Michigan Intercollegiate Athletic Association are affiliated with a religious organization. A Dean of Students at one of the independent institutions stated this would make a difference when comparing his institution with secular colleges (See Appendix A or B, Section F).

## Data Collection Procedures

This investigation utilized a descriptive survey research design (Babbie, 1990; Turney and Robb, 1971) that determines the distribution of certain characteristics, traits, attributes, and/or opinions within a sample at a given point in time. According to Babbie (1990, p. 52) ... the researcher is concerned not with why the observed distribution exists but merely with what that distribution is....although most surveys ... have the additional objective of making explanatory assertions about the population."

Subjects were all student members of judicial boards empowered by the institution to levy or recommend disciplinary sanctions, including suspension. No faculty or staff who served on judicial boards were

- -• ٠٠. ? • . • . 1. . ٠, 116

incl

this admi

puni

exp

C).

pro

included in this study. Subject's names and addresses were not known by this researcher. The institution's contact person was responsible for administering the survey to the appropriate judicial board members.

Each contact person was directly mailed in March 1993 the requisite number of survey booklets (See Appendix A and B), along with a letter explaining the instructions for administering the surveys (See Appendix C). The instructions emphasized the importance of getting a high return rate, encouraged the administration of the surveys within 2 weeks, and provided a system for distributing the survey booklets so that each version of the survey was given to 50 percent of the participants.

Regular follow-up phone calls were placed to each institutional contact person, that had not returned the survey booklets, to offer assistance and encouragement in completing the administration of the surveys.

Each survey booklet contained the same cover page explaining the purpose of the study, the average amount of time to complete the task, the confidential nature of their responses, and the voluntary nature of their participation. The cover letter also informed participants that returning the booklet constituted their informed and voluntary participation in this study. Additionally, participants were informed that they could contact the researcher at the listed address or phone number with any questions or comments.

Section A of the booklet contained the general instructions for completing the survey. This section explained to each participant that they would be completing 5 additional sections and urged them to answer all questions honestly. This section also reinforced the confidential and voluntary nature of their responses. Participants were informed that prior to each section, instructions for completing that portion of the booklet were provided.

Section B of the booklet featured one of the two rape scenarios, that is, one-half of the subjects received a survey booklet with the stranger scenario (See Appendix B) and one-half received the acquaintance scenario (See Appendix A). Section C, the Adjudication

Questionnaire, instructed the subjects to determine, according to only the information provided in the scenario, whether the male was guilty or not guilty of rape. Sections D and E featured the FEM Scale and the Attitudes Toward Rape Questionnaire respectively. Section F instructed people to complete the Demographic Information Form.

The sections in the survey booklet were ordered in an intentional manner. Specifically, the rape scenarios, followed by the adjudication questionnaire, were placed in front of the FEM Scale and Attitudes

Toward Rape Questionnaire in order to control for the influence the completion of the instruments might exert on the judgement of the male perpetrator. The FEM Scale was placed in front of the Attitudes Toward Rape Questionnaire to control for the emotional reactions people might have toward the word "rape" used in all of the ATR questions. Thus, this researcher wanted a measure of the participants gender-role attitudes free of any influence associated with the topic of rape.

To ensure that the instructions in the booklet were understandable and unambiguous, pilot studies were conducted on February 15 and 24, 1993 with 11 members of 2 residence hall judicial boards. The amount of time needed to complete the survey booklet was monitored. The researcher processed the instructions with each judicial board as a group to determine if people understood the intended meaning and made modifications accordingly.

# Scoring the Data

This study utilized 1 instrument comprised of 5 components. The Demographic Information Form and Adjudication Questionnaire provided nominal data that were used as independent variables, requiring only classification. The Rape Scenarios required no scoring.

The FEM Scale was a 20-item instrument designed to measure sexist or feminist attitudes. No sub-factors were identified in the scale. The instrument utilized a 5-point Likert Scale, scored according to a

ran

dis

SCC

ge:

·

.

range where: 1 = strongly agree; 2 = agree; 3 = no opinion; 4 = disagree; and 5 = strongly disagree. Questions 15, 16, 19, and 20 were scored in reverse order, where strongly agree = 5 and strongly disagree = 1. A high score was consistent with feminist or less traditional gender-role attitudes.

The Attitudes Toward Rape Questionnaire was composed of 8 subfactors. The instrument used a 6-point Likert Scale where: 1 = strongly
agree; 2 = agree; 3 = slightly agree; 4 = slightly disagree; 5 =
disagree; 6 = strongly disagree. As originally developed by Feild
(1978a), a high score was indicative of substantial rape myth belief, as
well as possessing anti-victim, pro-rape sentiments.

Subsequent modification of the ATR Questionnaire, because a new principal components factor analysis with varimax rotations (Babbie, 1990) was conducted, changed the scoring structure of the ATR Questionnaire. Specifically, items 2, 3, 20, and 22 were recoded so that strongly agree = 6 and strongly disagree = 1. These coding changes altered the direction of the scale. For the purpose of this investigation, a high score on the ATR Unified Rape Myth Factor was suggestive of low rape myth belief (See Chapter 4 for more specific information on the scoring changes associated with this instrument).

# Data Analysis

All respondents completing and returning the survey booklet were grouped according to whether their survey booklet included the stranger rape or acquaintance rape scenario. The groups were compared on the basis of gender, race or ethnic background, class standing, religious faith, academic major, judicial training, participation in a rape hearing, and personal knowledge of a rape victim. These comparisons were made using chi-square analysis, that "... is a convenient technique for determining the significance of the difference between the frequencies of occurrence in 2 or more categories ..." (Bartz, 1976, p.

297-298) when the data is "discrete" (Babbie, 1990, p. 252).

The groups were also compared according to age, academic grade point average, and months of judicial experience. A z-scale, "... the most widely used standard score scale in statistics ..." (Glass and Hopkins, 1984, p. 64) is often used to compare "continuous" data (Babbie, 1990, p. 252). However the t-test was chosen for this analysis because "the t-distribution ... looks very much like the normal curve..." (Bartz, 1976, p. 248). Additionally, "... it has become conventional always to use the t-distribution, regardless of sample size. This means that [one] can ignore the z-values ... in favor of a specific t-value which is determined by ... the degrees of freedom in [the] samples" (Bartz, 1976, p. 249).

Analysis of variance (ANOVA) was originally selected to determine what relationship, if any, existed between the judicial verdict, rape mode, and rape myth belief or gender-role attitudes. ANOVA was chosen because it "... is usually the appropriate method when the groups of observations are created by a categorical independent variable ....[and the desire is to] ... study ... the effect of a categorical variable on a metric (interval) variable (Iversen and Norpoth, 1987, p. 8). ANOVA also is the appropriate method for analyzing survey data, however it remains difficult to establish a causal effect (Iversen and Norpoth, 1987). For this study, the judicial verdict was considered the independent variable and the scores on the FEM Scale or Factor 1 of the Attitudes Toward Rape Scale were defined as the dependent variables.

Multivariate analysis of variance (MANOVA) was to be used to determine what relationship, if any, existed between gender-role attitudes and rape myth beliefs on the judicial verdict. MANOVA was originally selected because Bray and Maxwell (1985) state that "... it evaluates mean differences on two or more dependent variables simultaneously" (p. 8) and "... considers the correlations between the variables" (p. 9). MANOVA was also selected to determine what effect, if any, demographic variables (eg. race, age, gender) had on the

**\*.**\*\* ٠, ... : . . • relationships between judicial verdict, rape mode, rape myth belief and gender-role attitudes.

Regression was originally selected to determine the relationship, if any, between gender-role attitudes and rape myth belief. Regression was chosen because it is more effective than ANOVA when comparing one metric variable with another metric variable (Iversen and Norpoth, 1987). Additionally, multiple regression procedures were to be utilized to gain greater understanding concerning what variables had the greatest impact on the judicial verdicts. This decision to use multiple regression was made because Lewis-Beck (1990) stated it "... offers a fuller explanation of the dependent variable .... [and] second the effect of a particular independent variable is made more certain" (p. 47).

The aforementioned procedures were selected with the idea that the subjects participating in this investigation would choose both the guilty and not guilty verdicts. However, the unexpected finding that all 82 subjects judged the male to be guilty of rape limited the statistical analyses to t-tests, chi-squares, and correlation coefficients. No further statistical analyses were required.

According to Glass and Hopkins (1984), "a statistical hypothesis is presumed to be either true or false; by using inferential statistical methods the researcher makes a decision ... as to whether the statistical hypothesis ... is tenable or whether it must be rejected as false." (p. 204). A level of significance (p) of .05 was established for this study, which Borg and Gall (1983) determined to be the general standard used in educational research.

. • .

#### CHAPTER FOUR

#### PRESENTATION AND ANALYSIS OF THE DATA

## Introduction

The purpose of this study was to determine what effect college judicial board members' levels of rape myth belief and gender-role attitudes had on the judgement of a male perpetrator's guilt or innocence for committing an act of rape. In this study the differences between acquaintance rape and stranger rape situations were examined. The investigation was guided by the following questions:

- 1. What relationship, if any, exists between the verdicts of guilt or innocence for the male perpetrator and rape myth beliefs across acquaintance rape and stranger rape modes?
- 2. What relationship, if any, exists between the verdicts of guilt or innocence for the male perpetrator and gender-role attitudes across acquaintance rape and stranger rape modes?
- 3. What relationship, if any, exists between gender-role attitudes and rape myth belief on verdicts of guilt or innocence for the male perpetrator across acquaintance rape and stranger rape modes?

# Subjects

The subjects for this investigation were 113 student judicial board members at 6 independent and 9 state-assisted colleges in the state of Michigan. All the subjects were enrolled during the Spring Semester of 1993 and served on campus judicial boards empowered to suspend or recommend suspension for students found guilty of violating conduct rules. The final sample was composed of 85 board members from state-assisted institutions and 28 board members from independent colleges (See Chapter 3 for additional information).

Eighty-two judicial board members completed and returned the survey packets. This resulted in a 73 percent overall return rate. Fifty-four percent (n = 15) of the subjects at the independent institutions

• • . • • •

> • •

•

.

.

returned survey packets. Judicial board members at the state-assisted institutions returned 67 survey packets for a 79 percent return rate.

# Instrumentation

The instrument used in this study was comprised of 5 components. Two components, the FEM Scale (Smith, Perree, and Miller, 1975) and the Attitudes Toward Rape Questionnaire (Feild and Bienen, 1980) were administered in the same format as originally designed. The Acquaintance Rape and Stranger Rape Scenarios were modified, based on the expert advise of the Sexual Assault Safety Program Coordinator at Michigan State University, from the original versions developed by Quackenbush (1987). The Adjudication Questionnaire and the Demographic Information Form were developed by this investigator (See Chapter 3 for additional information on the instrumentation).

The Attitudes Toward Rape Questionnaire developed by Feild and Bienen (1980) was a 32-item instrument designed to measure people's attitudes toward rape. Based on the responses of 1,448 subjects, a principal components factor analysis with varimax rotations was used to derive 8 latent factors (eigenvalues > 1). A computed coefficient of congruence for each of the group pairings yielded a correlation between the factors that ranged from .81 to .89, with a mean of .87. The factors accounted for 50 percent of the variance among the items.

The researchers did not establish scale reliabilities, but determined that the square root of the estimated commonalities of the factors (the theoretical lower bound of reliability) had a mean value of .62. The researchers acknowledged this low value but found it acceptable given the heterogeneity of the items, the number of factors found, and the items with loadings greater than .30. Additionally, Feild (1978a) believed that face validity was implied because the 8 dimensions had been discussed in the literature. The 8 factors identified were:

Factor 1: Woman's responsibility in rape prevention.

- Factor 2: Sex as motivation for rape.
- Factor 3: Severe punishment for rape.
- Factor 4: Victim precipitation of rape.
- Factor 5: Normality of rapists.
- Factor 6: Power as motivation for rape.
- Factor 7: Favorable perception of a woman after rape.
- Factor 8: Resistance as woman's role during rape.

Feild and Bienen (1980. pp. 60-61) and Feild (1978a, pp. 162-164) presented their rotated factor structure in table form. However, some confusion surrounds the creation of the factors. Specifically, the researchers alluded to, but never clearly stated, that a loading of  $\pm$  .30 was used as the salient point for assigning ATR items to a particular factor (see Feild and Bienen, 1980, p. 58). Additionally, Feild and Bienen (1980) and Feild (1978a) did not clarify the ATR item assignments in the narrative sections of their writings.

Furthermore, the table used to present the rotated factor structure was not constructed consistent with the general recommendations for the presentation of such data (Child, 1990, p. 62; Harman, 1976, p. 277). The researchers only included those loadings greater than ± .09 in their table. As a result, the missing data created a false impression concerning the assignment of the ATR items to a particular factor. In fact, the confusion lead Koss and Leonard (1984) to conclude that Factor 1 of the ATR Questionnaire incorporated 26 of the 32 items, a conclusion not consistent with the principles of factor analysis (see Cattell, 1952).

Conflicting information was also presented in terms of interpreting Factor 3 (Severe Punishment for Rape) of the ATR Questionnaire.

Specifically, Feild (1978a, p. 161) wrote that "individuals who feel rapists should be severely punished score high on this factor (anti-rape attitude)." However, Feild and Bienen (1980, p. 58) contradicted this by writing "individuals who feel rapists should be severely punished would score low on this factor (antirape attitude)."

rev

· :

\*\*

•

· ·

•

**...** 

Finally, several of the loadings shown in the table of rotated factors were negative. Yet no mention was made by the researchers of reverse coding the scoring scale to overcome the bipolar nature (Child, 1990) of the factor loadings. This was important because some of the ATR questions were worded in a manner that was inconsistent with the direction of the scoring scale established by Feild and Bienen (1980). All in all, the reader was left to assume the particulars in the absence of specificity.

Thornton, Robbins, and Johnson (1981) conducted research involving 173 college undergraduates that utilized the Attitudes Toward Rape Questionnaire. The researchers conducted a principal components factor analysis with varimax rotations on the ATR Questionnaire for their sample. Using a salient criterion of .40 for factor inclusion, they discovered the emergence of only 4 factors:

- Factor 1: Victim precipitation and responsibility.
- Factor 2: Negative evaluation of rape.
- Factor 3: Sexual motivation as reason for rape.
- Factor 4: Power motivation as reason for rape.

Thornton et al. (1981) found that 4 of the ATR items failed to load appreciably on any of the 4 factors. Factor 1 accounted for 56 percent of the variance with an alpha of .89. The second factor accounted for 13 percent of the variance and demonstrated an alpha of .54. Factors 3 and 4 accounted for 12 percent (Alpha = .49) and 8 percent (Alpha = .60) of the variance respectively. The researchers concluded that their factor structure may have differed from Feild (1978a) because of sample differences. However, Thornton et al. acknowledged that comparable themes existed between the 2 investigations with regard to rape attitudes.

Given the aforementioned, a confirmatory principal components factor analysis with varimax rotations was conducted for the ATR Questionnaire specific to this investigation. The confirmatory analysis yielded a 6 factor rotation (See Appendix D for factor presentation).

• •

N 49.

.

· · · · · ·

•

um B The following factors emerged:

Factor 1: Rape precipitation and responsibility.

Factor 2: Rape motivated by sexual desire.

Factor 3: Rape as a crime.

Factor 4: Normality of rape.

Factor 5: Rape as a violent act.

Factor 6: Cultural influences on rape.

According to Child (1990, p. 39), an item may be included in a factor with loading values of  $\pm$  .30 or greater only when the sample size is larger than 100. Samples with fewer than 100 subjects should specify a level of significance similar to correlation coefficients when deciding the salience of the loadings. Thus a loading of .40, defined by Bartz (1976, p. 205) as a moderate correlation, was selected for this investigation as the salience level for inclusion of an item into a factor. As a result, 30 of the 32 items from the Attitudes Toward Rape Questionnaire were included in a factor (#1 and #17 failed to have appreciable loadings and were not included in any of the factors). The ATR items were categorized as follows:

#### Factor 1

- 6) A charge of rape two days after the act has occurred is probably not rape.
- 7) A woman should be responsible for preventing her own rape.
- 11) If a woman is going to be raped, she might as well relax and enjoy it.
- 13) Most women secretly desire to be raped.
- 14) It would do some women some good to get raped.
- 15) Women provoke rape by their appearance or behavior.
- 25) In most cases when a woman was raped, she was asking for it.
- 27) Rape of a woman by a man she knows can be defined as a "woman who changed her mind afterward."
- 30) The degree of a woman's resistance should be the major factor in determining if a rape has occurred.
- 31) A raped woman is a responsible victim, not an innocent one.

#### Factor 2

- 19) Rape is the expression of an uncontrollable desire for sex.
- 21) Rape is a sex crime.
- 23) During a rape, a woman should do everything she can do to resist.
- 24) Rapists are sexually frustrated individuals.
- 26) The reason most rapists commit rape is for sex.

#### Factor 3

- 18) In order to protect the male, it should be difficult to prove that a rape has occurred.
- 20) Rape is the worst crime that can be committed.
- 22) All rape is a male exercise in power over women.

#### Factor 4

- 3) Rapists are "normal" men.
- 5) All rapists are mentally sick.
- 10) A raped woman is a less desirable woman.
- 16) "Nice" women do not get raped.

#### Factor 5

- 2) The reason most rapists commit rape is for the thrill of physical violence.
- 8) A man who has committed rape should be given at least 30 years in prison.
- 28) A convicted rapist should be castrated.

## Factor 6

- 4) In forcible rape, the victim never causes the crime.
- 9) Women are trained by society to be rape victims.
- 12) Rape provides the opportunity for many rapists to show their manhood.
- 29) A woman should not feel guilty following a rape.
- 32) Rape serves as a way to put or keep women in their "place."

The factor analysis performed for this investigation yielded salient negative loadings on items 2 (~.570), 3 (~.573), and 18 (~.724). According to Guertin and Bailey (1970), this phenomena occurs in psychological tests and may be the result of the scoring mechanism. Logically, someone with a feminist orientation and enlightened attitudes

•

CC Wi

> a 1

.

:

3

(

toward rape would most likely select a response for items 2 and 3 from the lower end of the scale (SA = 1). Thus it was considered appropriate to reverse code the scoring scale for items 2 and 3 (SA = 6; SD = 1).

Reverse scoring item 18 (SA = 6; SD = 1) was not considered logical because a person with enlightened rape attitudes would most likely select a response from the "Strongly Disagree" end of the continuum. Thus item 18 was scored (SA = 1; SD = 6) as originally devised by Feild and Bienen (1980). Furthermore, items 20 and 22, the remaining items comprising Factor 3, were reviewed for their scoring consistency. It was determined that both items 20 and 22 should be reverse scored (SA = 6; SD = 1), as logic suggests that people with low rape myth beliefs would select responses from the lower end (SA) of the continuum.

Further review of the ATR items determined that items 4, 9, 12, 29, and 32 could logically be reverse coded (SA = 6; SD = 1) as well. This logic was also based on the belief that people with progressive views concerning rape would respond to each item at the lower end (SA) of the scale. Closer examination revealed that the aforementioned 5 items were all assigned to Factor 6 and were not disbursed among other factors. Thus any recoding only affected the negative/positive signs. The items assigned to Factor 6 were also not considered strong representations of common rape myths (Burt, 1980; Warshaw, 1988). In sum, the Factor 6 items were not recoded, as they had no direct utility for this investigation.

In review, only items 2, 3, 20, and 22 were reverse coded (SA = 6; SD = 1) for this analysis. These coding changes also altered the direction of the scale. In their study, Feild and Bienen (1980) interpreted a high score as indicative of substantial rape myth belief. For the purpose of this investigation, a high score on the ATR was suggestive of low rape myth belief.

A reliability check of the 6 latent factors demonstrated the coefficients shown in Table 1 below. Factors 1, 2, and 3 demonstrated the strongest alphas.

··

••

•

•

7

Table 1. Reliability Coefficients for the Revised ATR Factors

	<u>Mean</u>	SD	Alpha
Factors			
1	53.54	5.61	.807
2	18.59	4.57	.693
3	12.35	2.91	.529
4	17.99	3.13	.500
5	10.52	3.08	.413
6	19.39	3.91	.485

Correlations were obtained for the 6 latent factors. The results presented in Table 2 show that Factor 1 was significantly correlated with both Factors 2 and Factor 3.

Table 2. Correlation Between the Revised ATR Factors

		A	TR Factors			
	1	2	<u>3</u>	4	5	<u>6</u>
ATR Fact	ors					
1	1.0000	.2499*	.3215**	.0276	0622	0898
2	.2499*	1.0000	0027	.2540*	.0329	1949
3	.3215**	0027	1.0000	.0939	1535	0361
4	.0276	.2540*	.0939	1.0000	1506	1397
5	0622	.0329	1535	1506	1.0000	0589
5	0898	1949	0361	1397	0589	1.0000

\*\* Significant at .01 level

Based on the results of the reliability check and factor correlation, a Unified Factor was developed by combining Factors 1, 2, and 3. This was considered appropriate because of the significant correlations of Factor 1 with Factor 2 and Factor 3. The Unified Factor explained 84.1 percent of the variance, with a mean score of 84.48 (SD = 9.17). The alpha for the Unified Factor was .764 which demonstrated strong internal reliability. As a result, it was determined that the Unified Factor would be used as the measure of rape myth belief specific

to 1

sub the it

...

.

.

to this investigation.

Finally, 2 cross-tabulations were conducted. The first cross-tabulation was between the scores on the FEM Scale and the ATR Questionnaire prior to recoding any items. It was expected that subjects with more feminist viewpoints (symbolized by a higher score on the FEM Scale) would have more progressive attitudes toward rape and it's causes (symbolized by a lower score on the ATR Questionnaire).

Surprisingly, subjects with the highest FEM scores (> 91) tended to score higher on the ATR Questionnaire. Even more important was the fact that no subject within the highest FEM score grouping scored in the lowest quadrant of the ATR scores (< 112). This result was in contradiction to the literature (See Chapter 2) that documented the fact that people believe less in rape myth as their level of feminism increases.

A second cross-tabulation was conducted after recoding items 2, 3, 20, and 22 between the scores on the FEM Scale and the Unified Factor. The results were reversed from the first cross-tabulation. That is, subjects with high scores on the Unified Factor (symbolizing low rape myth belief) also scored higher on the FEM Scale (symbolizing feminist attitudes). Thus additional support for recoding ATR items 2, 3, 20, and 22 was apparent.

#### Summary

The identification of 6 factors rather than the 8 discovered by Feild and Bienen (1980) may be due to a much smaller sample used in this investigation. This sample was also much more specialized. However, a comparison of the factors identified by Thornton, Robbins, and Johnson (1981), Feild (1978a), and by this investigator suggest a basic underlying similarity.

The factor structures identified by the aforementioned researchers also demonstrated construct validity with consistent themes regarding rape attitudes. In particular, all 3 of the previously discussed factor

•• r . . ,<del>\*</del> • · · •• • • ÷ . • **†** 

.•

analyses contained a "victim responsibility" factor that was grounded in common rape myths.

The creation of a Unified Rape Myth Factor by combining Factors 1, 2, and 3 appears justified. The significant correlation of Factor 1 with Factor 2 and Factor 3 yielded a Unified Factor that explained 84.1 percent of the variance. The Unified Factor also demonstrated face validity as it contained a strong representation of rape myths identified in the literature (Burt, 1980; Warshaw, 1988).

The Unified Factor contained 10 of the 13 items Thornton et al. (1981) identified in their "victim precipitation and responsibility" factor. Assuming that Feild (1978a) used a salience level of  $\pm$  .30 and followed the standard principles of factor analysis, the Unified Factor also contained 8 of the 11 items used in the "women's responsibility in rape prevention" factor. Additionally, the Unified Factor contained 18 of the 32 items that comprised the Attitudes Toward Rape Questionnaire.

In summary, it appears that the Unified Factor is a comprehensive measure of rape myth belief. As such, it will be used as the basis for empirical analysis during this investigation.

#### General Profile of the Sample

The following section provides a descriptive overview of the general sample. Also provided are statistical analyses comparing the samples on the demographic variables associated with this study.

#### Gender

Research suggests that gender differences influence the attribution of blame and responsibility in rape situations (Barnett and Feild, 1977; Bridges and McGrail, 1989; Feldman-Summers and Lindner, 1976; Jenkins and Dambrot, 1987; Tieger, 1981; Tolor, 1978). These differences point to the importance of this variable.

Table 3 presents a comparison of the respondents' genders

·-

\* \*

.

•

differentiated by the rape scenarios. Women comprised 39 percent and men 61 percent of the acquaintance rape scenario. Conversely, the stranger rape scenario was represented by 59 percent females and 41 percent males. The gender of the respondents assigned to the acquaintance rape scenario was not significantly different from those in the stranger rape scenario (chi-square = 3.124, df = 1, p = .077).

Table 3. Comparison of Rape Scenarios By Gender

	<u>Female</u>	<u>Male</u>	<u>Total</u>
Scenario			
Acquaintance	16	25	41
Stranger	24	17	41
Total	40	42	82

#### Age

Age was viewed in relationship to the maturation process. As such, it was deemed a variable worthy of examination because of its influence on the development of emotional maturity and attitude/value formation (Astin, 1993; Pascarella and Terenzini, 1991).

Table 4 compares the respondents, by rape scenario, according to their mean age. No significant relationship existed between the two groups on the age variable (t = -.05, df = 80, p = .959). The ages for the overall sample ranged from 18 to 34.

Table 4. Mean Age of Respondents By Rape Scenarios

	Mean	<u>SD</u>	SE	
<u>Scenario</u>				
Acquaintance n = 41	21.34	2.56	.40	
Stranger n = 41	21.37	1.62	.25	

÷		
-		
•		
er		

# Ethnic Origin

Astin (1993) and Pascarella and Terenzini (1991) found that ethnicity was related to the development of a person's sense of identity, values, and interests. Thus, a person's ethnic origin was believed to be an important variable.

A comparison of ethnic origin within the scenario groupings is presented in Table 5. Minority respondents totaled 15 percent of the acquaintance scenario and 29 percent of the stranger scenario.

Ethnicity was not a significant factor in comparing the two groups for difference (chi-square = 3.588, df = 4, p = .465).

Table 5. Comparison of Rape Scenarios By Ethnicity

	<u>Acquaintance</u>	Stranger	Total
Ethnicity			
African-American	5	8	13
Hispanic/Chicano	1	2	3
Asian/Pacific Islander	-	1	1
White/Caucasian	35	29	64
Indian/Native People	-	1	1
Total	41	41	82

## Faith

Several of the independent institutions participating in this study were affiliated with religious organizations. Pascarella and Terenzini (1991, p. 303) reviewed the literature and concluded that "...institutional characteristics probably do play a role in the degree to which religious preferences, attitudes, values, and behaviors change during college."

Tables 6 and 7 show the number of respondents that practice a religious faith and their level of devoutness. Overall, 83 percent of the sample practiced a religious faith (signified by selecting "yes"). Thirty-five percent of the overall sample viewed themselves as extremely or strongly devout in their faith. A comparison of the two scenario

groups yielded a significant finding at the .01 level in terms of practicing a religious faith (chi-square = 8.613, df = 1, p = .003). A comparison of the groups on level of religious devotion (chi-square = 4.835, df = 4, p = .305) was not significant.

Catholicism (34 percent, n=28) and Protestantism (35 percent, n=29) were overall the most practiced religions. The Jewish, Hindu, and Muslim religions were each selected 1 percent (n=1 each) of the time, while the category of "Other" was selected by 11 percent (n=9) of the overall sample. Sixteen percent (n=13) of the overall sample did not consider themselves to be religious. Affiliation with a religious denomination (chi-square = 10.938, df = 6, p=.090) was not significant.

Table 6. Comparison of Rape Scenarios By Religious Practice\*

	<u>No</u> **	<u>Yes</u> **	<u>Total</u>	
Scenario				
Acquaintance	12	29	41	
Stranger	2	39	41	
Total	14	68	82	

<sup>\*</sup> Chi-square was significant at the .01 level

<sup>\*\*</sup> No = Did not practice a faith; Yes = Did practice a faith

!

.

•

Table 7. Comparison of Rape Scenarios By Religious Devoutness\*

	Acquaintance	Stranger	Total
<u>Devoutness</u>			
Extremely Devout	2	5	7
Strongly Devout	5	12	17
Somewhat Devout	11	13	24
Devout	7	8	15
Not Devout	4	1	5
Total	29	39	68

<sup>\*</sup> Only subjects who practiced a religious faith are reflected here

## Academic Class Standing

Academic class standing was viewed as both an indication of maturity (age) and collegiate experience. Juniors and Seniors were the most highly represented classes in the overall sample (70 percent), the acquaintance scenario (81 percent), and the stranger scenario (59 percent). No significant difference was found between the groups on this variable (chi-square = 8.839, df = 5, p = .116). Table 8 provides a complete overview.

Table 8. Comparison of Rape Scenarios By Academic Class

	Acquaintance	Stranger	Total
<u>Class</u>			
Freshman	4	3	7
Sophomore	4	9	13
Junior	15	9	24
Senior	18	15	33
Graduate	-	3	3
Other	-	2	2
Total	41	41	82

# Academic Major

Overall, 32 percent (n = 26) of the sample were majoring in the Social Sciences, 22 percent (n = 18) were majoring in Business, 10

: • • . . **:** 

percent (n = 8) in Arts and Letters, 9 percent (n = 7) in Engineering, 7 percent (n = 6) in Natural Sciences, and 6 percent (n = 5) each in Education and Health Science. Four percent (n = 3) of the sample were undeclared majors, with 3 percent (n = 2) and 1 percent (n = 1) majoring in Agriculture and Human Ecology respectively. One person failed to designate a major course of study.

#### Academic Grade Point Average

The cumulative grade point average was viewed as a symbol of academic involvement (Astin, 1993, p. 374) and achievement. Comparisons between the scenario groups did not yield a significant difference in this area (t = .72, df = 80, p = .475). The overall (n = 82) mean grade point average was 3.053 (SD = .458), with the acquaintance scenario group (n = 41) having a mean of 3.089 (SD = .480; SE = .075) and the stranger scenario group (n = 41) having a mean of 3.016 (SD = .438; SE = .068). The grade point averages ranged from 2.20 to a 4.00.

## Fraternity or Sorority Membership

Several studies have determined that membership in a fraternity or sorority negatively influences the development of cultural sophistication (Pascarella and Terenzini, 1991, p. 307). Thirty-one percent (n = 25) of the overall sample (n = 82), 34 percent (n = 14) of the acquaintance scenario sample (n = 41), and 27 percent (n = 11) of the stranger sample (n = 41) were members of fraternities or sororities. However, a comparison of the groups on this factor yielded no significance (chi-square = .518, df = 1, p = .472).

#### Judicial Board Experience

In order to gain insight into experience, the survey asked how many months the members had served on a judicial board and how many hearing they had participated in. Furthermore, the respondents were asked if they had ever served on a judicial board that adjudicated a campus rape

CASE.

Overall, the mean number of months served on a judicial board was 9.40 (SD = 8.305). When viewed from the perspective of each scenario, the acquaintance sample (n = 40) had served a mean of 8.63 (SD = 7.718; SE = 1.22) months and the stranger sample (n = 38) a mean of 10.21 (SD = 8.911; SE = 1.45) months. This relationship between the two samples was not significant (t = -.84, df = 76, p = .403). The range for this variable was 0 months (several new members were just taking positions) to 36 months.

Members had participated in an overall mean of 6.35 (SD = 9.243) hearings during their tenure as judicial board members. Table 9 provides an overview of the mean number of judicial hearings members had participated in. A comparison of the 2 samples did not result in any significance (t = -1.45, df = 76, p = .150). The range for this variable was 0 to 50 hearings.

Table 9. Number Hearings Participated In By Rape Scenarios

Mean	<u>ŞD</u>	<u>se</u>
4.88	6.43	1.02
7.90	11.37	1.85
	4.88	4.88 6.43

Twenty-six percent of the overall sample had participated in a hearing that involved a charge of rape (signified by choosing "yes"). Twenty-four percent of the stranger scenario sample and 28 percent of the acquaintance scenario sample had served as a hearing agent for a rape case. No significant relationship was present between the samples on this variable (chi-square = .205, df = 1, p = .651). Table 10 illustrates this variable.

Table 10. Participation in Rape Hearings By Rape Scenarios

	<u>No</u> *	Yes*	<u>Total</u>	
<u>Scenario</u>				
Acquaintance	28	11	39	
Stranger	29	9	38	
Total	57	20	77	

<sup>\*</sup> No = Have never participated in a rape hearing Yes = Have participated in a rape hearing

#### Training

Respondents were asked if they had ever received any formal training about the topic of rape and/or hearing rape cases. Bighty percent of the overall sample indicated that they had never received any training concerning the aforementioned. This pattern was consistent within the two samples, with 74 percent of the acquaintance scenario and 87 percent of the stranger scenario respondents indicating that they had received no training. A comparison of the groups yielded no significant differences on this variable (chi-square = 2.077, df = 1, p = .150).

See Table 11 for additional information.

Table 11. Received Training About Rape By Rape Scenario

	No*	Yes*	<u>Total</u>
<u>Scenario</u>			
Acquaintance	28	10	38
Stranger	33	5	38
Total	61	15	76

<sup>\*</sup> No = Have never received training; Yes = Have received training

## Knowledge Of Rape Victims

Sixty-eight percent of the total sample had knowledge of someone

2 <b>C</b>		
7		
•		
1		
• •		

who had been raped. This was further delineated by 71 percent of the acquaintance scenario subjects and 65 percent of the stranger scenario respondents having knowledge of a rape victim(s). No significance was present in the comparison between the samples (chi-square = .305, df = 1, p = .581). See Table 12 for details.

Table 12. Knowledge of Rape Victims By Rape Scenarios

	No*	Yes*	<u>Total</u>
<u>Scenario</u>			
Acquaintance	12	29	41
Stranger	14	26	40
Total	26	55	81

<sup>\*</sup> No = Don't know a rape victim; Yes = Do know a rape victim

Table 13 illustrates the relationship the judicial board members had with the rape victim(s) they were knowledgeable of. None of the relationships were significant in terms of a comparison across stranger rape and acquaintance rape samples. Of special note was the fact that 12 percent, all females, of the overall sample had experienced an act of rape against themselves.

Table 13. Relationship With Rape Victim By Rape Scenario

	Acqu	aintance	<u>Stra</u>	nger	Tota	<u>al</u>
	<u>No</u>	<u>Yes</u>	<u>No</u>	<u>Yes</u>	<u>No</u>	Yes
Victim Relationship						
Relative	37	4	35	5	72	9
Acquaintance	28	13	29	11	57	24
Spouse	41	-	40	-	81	-
Friend	19	22	21	19	40	41
Lover	39	2	38	2	77	4
I Have Been Raped	37	4	34	6	71	10

•

As stated above, none of the victim relationship comparisons were significant. The following is a statistical verification of that fact:

Relative: Chi-square = .154, df = 1, p = .694. Acquaintance: Chi-square = .172, df = 1, p = .678 Spouse: Unable to compute given the empty cells Friend: Chi-square = .307, df = 1, p = .579 Lover: Chi-square = .001, df = 1, p = .980 I have been raped: Chi-square = .514, df = 1, p = .473

#### Summary

The analyses of the demographic variables associated with this study revealed that subjects randomly assigned to the different rape scenarios were similar in terms of gender, age, ethnic origin, level of religious devotion, type of religion practiced, academic standing, academic grade point average, fraternity/sorority membership, judicial board experience, rape awareness training received, and level of personal knowledge about rape victims. The only significant difference between the 2 samples was in the area of practicing a religious faith.

The similarity of the two samples on the demographic variables demonstrates that confounding variables did not influence the outcomes associated with the completion of the FEM Scale, the Attitudes Toward Rape Questionnaire, and the Adjudication Questionnaire.

#### Empirical Analysis

The judicial verdicts of guilt or innocence selected by each subject after reading the rape scenarios were viewed as the independent variables. The scores from the FEM Scale and from the revised ATR rape myth factors were viewed as the dependent variables. Subjects were also asked to assign a judicial sanction to the male when they viewed him as guilty of committing an act of rape.

Table 14 presents a breakdown of the judicial verdicts by rape scenario. Interestingly, all 82 subjects judged the man as guilty of rape.

Table 14. Judicial Verdict By Rape Scenario

	Ver	dict	
	<u>Guilty</u>	Not Guilty	<u>Total</u>
<u>Scenario</u>			
Acquaintance	41	-	41
Stranger	41	-	41
Total	82	-	82

Judicial sanctions were viewed as a measure of how wrong a subject perceived the rape to be. The available sanctions ranged from a lenient "No Action" to the more severe "Expulsion/Dismissal". Ninety-three percent of the subjects either selected a sanction of "Suspension" or "Expulsion/Dismissal" from the college. Seven percent, all in the group that read the acquaintance rape scenario, selected the "Probation" sanction. No subject selected the sanctions of "No Action", "Warning", or "Censure". Overall, rape was viewed seriously and sanctions assigned accordingly (See Table 15).

Table 15. Judicial Sanctions By Rape Scenario

	Acquaintance	Stranger	Total
Sanctions			
No Action	_	_	-
Warning/Reprimand	_	_	-
Censure	-	_	-
Probation	6	_	6
Suspension	15	6	21
Expulsion/Dismissal	20	34	54
Total	41	40	81
	7.0		

# FEM Scale

Further analysis involving the FEM Scale was undertaken. Specifically, t-tests of significance were used to compare demographic

variables with their scores on the FEM Scale. A significant relationship was found for gender and previous knowledge of someone who has been raped. Significance was approached for the variables of race (African-American and Caucasian) and participation in a judicial hearing involving a rape case. The scores on this scale ranged from 60 to 100, indicating a full spectrum of attitudes toward gender roles.

Tables 16, 17, 18, and 19 present a statistical summary of the above.

## ATR Questionnaire (Unified Factor)

Demographic variables were compared according to their scores on the Unified Rape Myth Factor via the use of t-test procedures. A significant relationship was established for the gender variable. Subjects with previous knowledge of a rape victim approached significance in relationship to the Unified Rape Myth Factor. A full spectrum of rape myth belief was apparent, with scores ranging from 53 to 103.

Tables 20 and 21 provide a statistical verification of the relationships.

Table 16. Significance Test of FEM and Gender

	Mean	SD	SE	<u>t</u>	<u>df</u>	Prob
<u>Gender</u>						
Female n=40	85.55	10.47	1.66	3.43	80	.001*
Male	78.01	9.42	1.45			
n=42	*	Significa	nt at .001	level		

Table 17. Significance Test of FRM and Knowing a Rape Victim

	Mean	SD	SE	ţ	df	Prob
Know Victin	1					
No n=26	77.33	10.59	2.08			
n-20				-2.81	79	.006*
Yes n=55	84.05	9.80	1.32			
N-33	*	Significat	nt at the	.01 level		
		<del></del>		•		
	Table 1	8. Signifi	cance Test	t of FEM and	d Ethnici	ty
	Mean	<u>sp</u>	<u>SE</u>	<u>t</u>	<u>df</u>	Prob
Ethnicity						
Black/AA	75.92	9.31	2.58			

Table 19. Significance Test of FEM and Rape Hearing Participation

-1.95 75 .055

n=13

n=64

White 82.01 10.42 1.30

	Mean	<u>sd</u>	SE	<u>t</u>	<u>df</u>	Prob
Particip.						
No n=57	81.01	10.60	1.40	-1.78	75	.079
Yes n=20	85.80	9.58	2.14			

Table 20. Significance Test of ATR Unified Factor and Gender

	Mean	SD	SE	ţ	df	Prob
Gender						
Female n=40	87.74	7.20	1.14			
				3.33	80	.001*
Male n=42	81.37	9.83	1.52			
	*	Significa	nt at the	.001 level		

Table 21. Significance Test of ATR Unified Factor and Victim Knowledge

	Mean	<u>SD</u>	<u>se</u>	<u>t</u>	df	Prob
Know Victim	l					
No n=26	82.02	8.42	1.65	-1.73	79	.088
Yes n=55	85.75	9.38	1.27			

The hypotheses tested for this study are presented and analyzed in the following paragraphs. The analysis is limited by the fact that 100 percent of the subjects selected the guilty verdict in judgement of the male portrayed in the rape scenarios.

## Hypothesis 1

Hypothesis 1 states that as a group, subjects with high level belief in rape myths will most frequently select the not guilty verdict in the acquaintance rape mode. This hypothesis was not supported because no subjects selected the not guilty verdict. No further analysis was required.

#### Hypothesis 2

Hypothesis 2 states that as a group, subjects with low level belief in rape myths will most frequently select the guilty verdict in the acquaintance rape mode. This hypothesis was supported. However, caution is warranted because guilt was selected by all subjects regardless of their level of rape myth belief. No further analysis was required.

#### Hypothesis 3

Hypothesis 3 states that as a group, subjects with high level belief in rape myths will most frequently select the guilty verdict in the stranger rape mode. This hypothesis was supported. However, caution is appropriate because guilt was selected by all subjects irrespective of their level of rape myth belief. No further analysis was required.

## Hypothesis 4

Hypothesis 4 states that as a group, subjects with low level belief in rape myths will most frequently select the guilty verdict in the stranger rape mode. This hypothesis was supported. Caution is advised, however, because guilt was selected by all subjects regardless of their level of rape myth belief. No further analysis was required.

#### Hypothesis 5

Hypothesis 5 states that as a group, subjects with sexist beliefs will most frequently select the not guilty verdict in the acquaintance rape mode. This hypothesis was not supported because no subjects selected the not guilty verdict. No further analysis was required.

#### Hypothesis 6

Hypothesis 6 states that as a group, subjects with feminist beliefs will most frequently select the quilty verdict in the acquaintance rape

Ť 

mode. This hypothesis was supported. Caution is warranted because guilt was selected by all subjects regardless of their gender-role beliefs. No further analysis was required.

#### Hypothesis 7

Hypothesis 7 states that as a group, subjects with sexist beliefs will most frequently select the guilty verdict in the stranger rape mode. This hypothesis was supported. Caution is appropriate because guilt was selected by all subjects irrespective of their gender-role beliefs. No further analysis was required.

#### Hypothesis 8

Hypothesis 8 states that as a group, subjects with feminist beliefs will most frequently select the guilty verdict in the stranger rape mode. This hypothesis was supported. Caution is advised because guilt was selected by all subjects regardless of their gender-role beliefs. No further analysis was required.

## Hypothesis 9

Hypothesis 9 states that overall, there will be a positive relationship between gender-role attitudes and rape myth belief. A correlation of the scores on the FEM Scale with the re-configured ATR factor scores yielded significant correlations (See Table 22). This hypothesis was supported.

Table 22. Correlation Between FEM Scale and ATR Factors

	FEM Scale	
ATR Facto	rs	
Factor 1	.5677**	
Factor 2	.2471*	
Factor 3	.4679**	
Unified	.6190**	
	diministrant at OF lavel	
*	Significant at .05 level	
**	Significant at .01 level	

#### Hypothesis 10

Hypothesis 10 states that overall, gender-role attitudes will be more influential than rape myth belief in determining the verdict in the acquaintance rape mode. This hypothesis was not supported as all subjects selected the guilty verdict regardless of their level of rape myth belief or gender-role attitudes. No further analysis was required.

#### Hypothesis 11

Hypothesis 11 states that overall, gender-role attitudes will be more influential than rape myth belief in determining the verdict in the stranger rape mode. This hypothesis was not supported as all subjects selected the guilty verdict regardless of their level of rape myth belief or gender-role attitudes. No further analysis was required.

#### Descriptive Analysis

As presented in Tables 16 and 20, gender was found to be a significant factor in level of rape myth belief and attitudes toward women. Specifically, females held significantly more feminist attitudes and believed significantly less in rape myth than did males. Interestingly, 12 percent of the total sample (all females) had experienced an act of rape against themselves. Additionally, a correlation between gender and knowing a rape victim yielded a

coefficient of -.3450 that was significance at the .01 level, pointing to the fact that women know more rape victims than males.

The influence of gender on the FEM Scale and the Unified Rape Myth Factor is more clearly demonstrated in Tables 23 and 24 below. Males predominate the lower scores in both instances, whereas females clustered more at the upper end of the scales. As a reminder, the higher the FEM score, the more feminist the attitudes. A higher score on the Unified Factor symbolized lower belief in rape myth.

Table 23. Gender Influence On FEM Scale

	<u>Male</u> n=42	<u>Female</u> n=40
FEM Scores		
Range 60-73	14(33%)	7(18%)
Range 74-81	14(33%)	8(20%)
Range 82-92	11(26%)	10(25%)
Range 93-100	3 (07%)	15 (38%)

Table 24. Gender Influence On Unified Factor

	<u>Male</u> n=42	<u>Female</u> n=40
Unified Scores		
Range 53-79	17(40%)	5(13%)
Range 80-85 Range 86-92	11(26%) 10(24%)	11(28%) 10(25%)
Range 93-103	4(10%)	14(35%)

The variables of age, number of months on a judicial board, or the number of hearings participated in were not significant as they related to either the FEM Scale or the Unified Factor. Prior participation in a rape hearing was not significant either. Subject's practice of a religion or level of devoutness was not significant on either the PEM or Unified Factor. Being a member of a fraternity or sorority was not a significant variable on either of the scales. However, membership in a

Greek organization was significantly correlated with knowing a rape victim (p. = .05; .2304).

Knowledge of a rape victim was significant in relationship to the FEM Scale (See Table 17). Knowing a rape victim also approached significance in relationship with the Unified Factor (See Table 21). Overall, as presented in Table 12, 68 percent of the sample had personal knowledge of someone who had been raped. When asked to define their relationship to the victim (See Table 13), subjects indicated a significant correlation with the victim being a relative (p = .05; .2431), an acquaintance (p = .01; .4461), or a friend (p = .01; .6961). It was also significant that 10 women identified themselves as rape victims (p = .05; .2580).

Academic variables such as class standing, major, or grade point average were not significant in relationship to either the FEM or Unified Scale. Only 20 percent of the sample had ever received training concerning rape dynamics, thus this variable showed no significance in relationship to either of the attitudinal scales.

Table 18 outlined the near significant relationship that existed between ethnicity and the FEM Scale. Conversely, there was no significance between ethnicity (African-American/White) and the Unified Rape Myth Factor (t=-.88, df=75, p=.382). Table 25 below presents a breakdown of subjects by ethnic classification on the FEM Scale.

Table 25. Ethnic Influence on FEM Scale

					<del></del>	
FEM Scores						
	60-73	74-81	82-92	93-100	Total	
Ethnicity						
Black/AA Hispanic Asian White Indian	4 - 17	7 - - 15	1 2 1 17	1 1 - 15 1	13 3 1 64	
Total	21	22	21	18	82	

₹`` ₹`

.

## Chapter Summary

This chapter presented several analyses. First was an analysis of the Attitudes Toward Rape Questionnaire that determined a different factor structure than the one originally developed by Feild and Bienen (1980). As a result, a new rape myth factor called the Unified Factor was developed and utilized for this investigation. Additionally, the scoring of the ATR Questionnaire was modified so that a high score was suggestive of low rape myth belief.

The demographic characteristics associated with the 82 subjects were analyzed using t-test or chi-square procedures. The only significant difference between the acquaintance rape and stranger rape groups was in the area of practicing a religious faith. Overall, the samples were similar, with no confounding variables influencing the outcomes of this investigation.

All subjects in this investigation selected the guilty verdict when judging the male perpetrator, regardless of the rape scenario they read. Furthermore, subjects generally assigned the harsher judicial sanctions of suspension or expulsion from college. In fact, only 6 people assigned a sanction less than suspension and they all assigned a probationary term to the guilty male. The 100 percent guilt finding by the respondents limited the extent of statistical analyses that could be performed.

The FEM Scale was further analyzed using t-test procedures. The results indicated a significant relationship between the FEM Scale and the variables of gender and previous knowledge of a rape victim. Significance was approached between the FEM Scale and the variables of race (African-American/White) and participation in a judicial hearing involving a rape case.

A comparison of the demographic variables with the Unified Rape
Myth Factor yielded a significant relationship with gender. Subjects
with previous knowledge of a rape victim approached significance in

relationship to the Unified Factor.

Being a member of a fraternity or sorority was significantly correlated with knowing a rape victim. Overall, over 68 percent of the sample personally knew a rape victim. The relationship the subjects had with the victims yielded significance on the variables of relative, acquaintance, and friend. It was also significant that 12 percent of the overall sample, all women, had experienced an act of rape against themselves.

Hypotheses 2, 3, 4, 6, 7, 8, and 9 were supported by the findings of this investigation. However, special caution was warranted because a guilty verdict was selected by all the subjects regardless of the level of rape myth belief, gender-role attitudes, or rape scenario read. Hypothesis 9 did yield a positive, significant correlation between gender-role attitudes and rape myth belief.

Hypotheses 1, 5, 10, and 11 were not supported by the findings of this investigation. Once again, the fact that subjects selected guilt 100 percent of the time warrants caution in interpretation.

In summary, the subjects of this investigation demonstrated the expected range of attitudes. Specifically, a full range of rape myth belief was present. Attitudes toward the roles of women ran a full spectrum. Gender differences were found in terms of both rape myth belief and feminist attitudes. Thus it was expected that differences on the determination of the verdict would be found. This was not the case, as attitudinal variables did not play a role in the determination of the verdict.

#### CHAPTER FIVE

## SUMMARY, DISCUSSION, AND RECOMMENDATIONS

#### Introduction

This chapter contains a summary of the investigation, including the theoretical foundation, the statement of the problem, the purpose of the investigation, the need for the study, the guiding research questions, and the methodology employed. Also addressed are the major findings, the limitations of the study, and the generalizability of the results. Finally, this chapter presents the conclusions, implications, and recommendations for further research.

#### Summary

## Theoretical Foundation

Approximately 80 to 90 percent of female rape victims are assaulted by acquaintances (Ageton, 1983; Koss, 1988; National Victim Center, 1992). This phenomenon is especially apparent on college campuses where 12 percent of the women had experienced an attempted rape and 15 percent had been raped (Koss, Gidycz, and Wisniewski, 1987). Furthermore, Eskridge (1989) found that 7 percent of male college students had forced a female to have sexual intercourse against her will at least once.

Contemporary theories of rape take into account "... the more prevalent ... forms of rape committed by acquaintances (date rape)" (Ellis, 1989, p. 9). Feminist theory postulates that rapists are not sexual deviants (Check and Malamuth, 1985) and that rape is a result of male dominated social traditions that view women as inferior (Ellis, 1989; Morrison, 1980). This creates a "rape culture" (MacKinnon, 1983) where "... hostility, control, and dominance... " (Groth and Birnbaum, 1980, p. 21) are the goals.

According to feminist theory, a visible symbol of this "rape culture" are rape laws that maintain the status quo and represent the sexism present in society (Brownmiller, 1975; Estrich, 1987). Although

rape laws have been reformed, beginning with the Michigan Criminal Sexual Conduct Statute (Cobb and Schauer, 1974), they still maintain the vestiges of early common law (Estrich, 1987). This remains an important point given that rape laws attempt to objectively define the criteria used to judge guilt and innocence.

Attribution theory (Heider, 1958) postulates that objectivity is suspect regardless of the criteria used to judge a situation. The premise being that first-hand observations and secondary reports of an event are subject to error given the complexity of the social stimuli surrounding the event (Shaver, 1975). This is especially true in acquaintance rape situations where people rely on biased or inaccurate beliefs or assumptions about the nature of rape, commonly called rape myths (Burt, 1980).

Research conducted by Wilson, Faison, and Britton (1983) suggested that the level of belief in rape myths influences the definitional boundaries between rape and normal sexual intercourse. Thornton, Ryckman, and Robbins (1982) and Williams (1979) found that gender and gender-role attitudes were the 2 primary contributors concerning the amount of blame attributed to the victim. Additionally, gender-role ideology and general belief in rape myths have been found to be related (Briere and Malamuth, 1983; Check and Malamuth, 1983; Muehlenhard and Hollabaugh, 1988).

Langer, Blank, and Chanowitz (1978) demonstrated that people ignore relevant information when a situation is consistent with their prior expectations. As such, hearing bodies often ignore the strict application of the law and respond to the circumstances of each case according to the general stereotypes of society (Brooks and Doob, 1975; Feild and Bienen, 1980). The result may be an arbitrary hearing where the victim's gender-role behavior influenced the verdict more than the evidence of the rape case (LaFree, Reskin, and Visher, 1985).

## Statement of the Problem

Heterosexual acquaintance rape is a social problem on college campuses, with approximately 27 percent of college women experiencing rape while on a date (Miller and Marshall, 1987). When reported, colleges often respond by adjudicating alleged perpetrators in acquaintance rape cases via institutional student judicial processes. These processes often include the option of having peer judicial boards hear the facts of the case, determine a verdict, and render discipline sanctions when applicable.

Critics (Parrot, 1991b) charge that judicial boards are illprepared to adjudicate acquaintance rape cases because of inadequate
training and lack of sensitivity concerning the dynamics of acquaintance
rape situations. Furthermore, research involving both actual and mock
juries demonstrated that verdicts in acquaintance rape cases were often
influenced more by the individual member's levels of belief in rape
myths and gender-role stereotypes than by actual facts.

# Purpose of the Study

The review of the literature clearly documented that college students, both male and female, believed in traditional gender-role stereotypes (Rosenkrantz, et al., 1968) that influenced dating relationships (Muehlenhard, Friedman, and Thomas, 1985) and perceptions concerning a willingness to engage in sexual intercourse (Bostwick, 1990). College students were also found to believe in rape myths (Check and Malamuth, 1985) that influenced the definitional boundaries between rape and normal sexual activity (Wilson, Faison, and Britton, 1983). Furthermore, rape myths were found to be related to views concerning the roles and rights of women in society (Costin, 1985), with students holding traditional views more likely to believe in rape myths and to accept sexual aggression (Greendlinger and Byrne, 1987). Men with strong gender-role stereotyping beliefs reported a greater belief in rape myths and more likelihood of using aggression in sexual situations

(Malamuth, 1981).

A review of the literature on attribution demonstrated that people defined a situation as rape according to their stereotypes of rape (Klemmack and Klemmack, 1976). For example, rape was more often defined when the woman resisted (Krulewitz and Nash, 1979) and when she was not acquainted with the rapist (McGraw, 1987). Further research determined that gender and gender-role attitudes were the 2 primary contributors concerning the definition of rape (Thornton, Ryckman, and Robbins, 1982), with males viewing the rape less seriously than females (Calhoun, Selby, and Warring, 1976; Tieger, 1981).

The literature clearly and consistently identified the prevalence of rape myth beliefs and gender-role attitudes in the general college population. Also defined was the influence of these beliefs and attitudes on attributions of blame and responsibility in acquaintance rape situations. Of special note was the consistent results of the studies, over time, from a variety of colleges and universities.

Research involving both actual and mock juries demonstrated that verdicts in acquaintance rape cases were often influenced more by the individual member's level of belief in rape myths and gender-role attitudes than by actual facts (LaFree, Reskin, and Visher, 1985). In essence, jury members reflected the stereotypes of society. This was especially relevant when related to the effectiveness of peer judicial boards adjudicating acquaintance rape cases on college campuses. Specifically, given the documented level of rape myth belief and traditional gender-role attitudes in the general college population, were college judicial board members more aware of the influence of rape myths and gender-role attitudes on their judgement of blame and responsibility in rape cases?

This was an important question given that judicial board members are trained to impartially judge all cases heard before them. If individual members subscribe to the same beliefs as their peers in terms of rape myth beliefs, Parrot (1991a, 1991b) and other critics may be

.

*:* 

,

·

.

,

••

e.

٠,٠

.

.

••

•

correct in questioning the competence of judicial boards to adjudicate acquaintance rape cases on campus. Thus, this study was designed to answer the aforementioned question. Specifically, the purpose of this study was to determine what effect actual college judicial board members' levels of rape myth belief and gender-role attitudes had on the judgement of a male perpetrator's guilt or innocence for committing an act of rape. The rape was depicted by 2 written scenarios featuring an acquaintance assault and a stranger assault. The study specifically examined the differences between the acquaintance rape and stranger rape situations consistent with the purpose of this study.

#### Need for the Study

Pavela (1992a) stated that colleges have a responsibility to resolve acquaintance rape cases based on their obligation to: (a) promote the moral development of students; and (b) protect the safety of people using their premises. Furthermore, Pavela stated that college judicial boards play an important role in fulfilling the college's obligation to adjudicate acquaintance rape cases. Pavela also stated that college judicial boards are composed of qualified individuals, that when guided by a competent advisor are capable of rendering sound judgements in even the most complex cases.

Critics, such as Parrot (1991a, 1991b) however question the competence of judicial boards to adjudicate acquaintance rape cases. They question whether judicial boards are adequately trained and advised about acquaintance rape situations to make sound judgements. This question is especially important when one looks at attribution research (Howard, 1984a, 1984b; LaFree, 1980; Smith, et al., 1976; Wilson, et al., 1983) that indicated acquaintance rape cases are often decided on extra-legal factors based on rape myth belief and gender-role attitudes. Furthermore, much of the attribution research was conducted with college students, thus providing a baseline for inference about how college students attribute blame and responsibility in acquaintance rape

) É

•

*:* 

..

•.. •

•

: •

•

•

•.

situations.

Even given the current debate concerning the effectiveness of judicial boards, this author was unable to find published research that determined how actual college judicial boards adjudicate acquaintance rape cases. No research was found that measured the level of rape myth belief or the gender-role attitudes of actual judicial board members. Finally, no research was discovered regarding the impact of attitudes and beliefs on either the verdicts or discipline sanctions levied in rape cases.

The literature that has been published on this subject generally critiques college judicial programs and/or provided recommendations for policy implementation (Adams and Abarbanel, 1988; Parrot, 1991a, 1991b; Pavela, 1992a). Thus, this research appears to substantially contribute to the literature by providing empirical data concerning the effect of gender-role attitudes and rape myth beliefs on the adjudication of a hypothetical rape case. Depending on the outcome, this research could contribute to the effective advising, selection, and training of judicial board members.

# Research Questions

This investigation was guided by the following questions:

- What relationship, if any, exists between the verdicts of guilt or innocence for the male perpetrator and rape myth beliefs across acquaintance rape and stranger rape modes?
- 2. What relationship, if any, exists between the verdicts of guilt or innocence for the male perpetrator and gender-role attitudes across acquaintance rape and stranger rape modes?
- 3. What relationship, if any, exists between gender-role attitudes and rape myth belief on verdicts of guilt or innocence for the male perpetrator across acquaintance rape and stranger rape modes?

.

·

•.

.

.

r

.

i tea

:

## Research Methodology

<u>Subjects</u>: The subjects of this investigation included 113 judicial board members enrolled as undergraduates for the spring semester of 1993 at 6 independent and 9 state-assisted colleges within the state of Michigan. All subjects served on campus judicial boards empowered to suspend or recommend suspension for students found guilty of violating college conduct standards. Eighty-two judicial board members completed and returned the survey packets, resulting in a 73 percent overall return rate.

Instrumentation (See Appendix A or B): This investigation utilized 1 instrument comprised of 5 components. Two of the components, the FEM Scale (Smith, Ferree, and Miller, 1975) and the Attitudes Toward Rape Questionnaire (Feild and Bienen, 1980) were used as originally developed. The Acquaintance Rape and Stranger Rape Scenarios were modified, based on the expert advise of the Sexual Assault Safety Program Coordinator at Michigan State University, from the original versions developed by Quackenbush (1987). The Adjudication Questionnaire and the Demographic Information Form were developed by this investigator.

The FEM Scale is a 20-item scale measuring attitudes toward feminism and traditional gender-role norms. The instrument utilized a 5-point Likert Scale, scored according to a range from strongly agree (1) to strongly disagree (5). Questions 15, 16, 19, and 20 were scored in reverse order. A high score was consistent with feminist or less traditional gender-role attitudes.

The Attitudes Toward Rape Questionnaire is a 32-item questionnaire composed of 8 latent factors that measures people's multidimensional attitudes about rape. The questionnaire used both positively and negatively phrased questions that respondents answered by using a 6-point Likert Scale ranging from strongly agree (1) to strongly disagree (6). Originally, Factor 1 titled "Women's Responsibility In Rape Prevention", was to be used as a measure of rape myth belief. Koss and

V-•. • • • c •

Leonard (1984, p. 220) stated that "... this factor could be interpreted as a general rape myth factor."

As originally developed by Feild (1978a), a high score was indicative of substantial rape myth belief. Subsequent modification of the ATR, based on the identification of 6 latent factors via a confirmatory principal components factor analysis with varimax rotations (Babbie, 1990), changed the scoring structure. Specifically, items 2, 3, 20, and 22 were recoded so that strongly agree = 6 and strongly disagree = 1. Thus a high score on the newly created Unified Rape Myth Factor was suggestive of low rape myth belief (See Appendix D for factor presentation; Chapter 4 also provides more detailed information on the creation of the Unified Factor).

The Acquaintance Rape and Stranger Rape Scenarios provided identical demographic and situational information through the first 7 paragraphs. The remaining paragraphs of the scenarios were identical in terms of the female's response and the male's actions, the only difference being the stranger or acquainted condition of the alleged rapist.

The Adjudication Questionnaire was utilized to determine whether the respondents judged the alleged male rapist as guilty or innocent of rape. When respondents judged the alleged rapist as guilty, the questionnaire also instructed them to select a disciplinary sanction that most closely approximated their view of appropriate punishment for the offense.

The Demographic Information Form collected personal data (e.g. gender, race, type of college attended, etc.) from each subject.

Additional questions were asked that discerned each subject's level of exposure to the topic of rape.

Research Design: The chief student affairs offices at 13 stateassisted and 7 independent colleges in Michigan were initially contacted
via phone to determine if they had a judicial system that included a
board empowered to hear cases that could result in suspension. Ten

• : ٠, **†** · state-assisted institutions and 6 independent institutions were deemed as having the requisite judicial structure.

A second phone call was made approximately 2 months later to the student affairs contact person at the 16 institutions. The purpose was to enlist their support and cooperation in the implementation of this investigation. As a result, 6 independent and 9 state-assisted institutions agreed to cooperate with this research project.

Each student affairs contact person at the participating institutions was mailed in March of 1993 the survey booklets for the judicial board members. The identities of the judicial board members were unknown to this investigator. The contact people at each college were given instructions about administration and participants were asked to seal the completed survey packets in a pre-addressed, stamped envelope for return, thus insuring confidentiality.

The survey booklets were identical except for the rape scenario. Thus, 50 percent of the participants received a survey packet asking them to judge the guilt of the male within the context of an acquaintance scenario. The remaining 50 percent received a survey packet instructing them to judge the stranger scenario. All participants, regardless of the scenario, completed the FEM Scale, ATR Questionnaire, Demographic Information Form, and the Adjudication Questionnaire.

Data Analysis: All respondents completing and returning the survey booklets were grouped according to whether their survey booklet included the stranger rape or acquaintance rape scenario. The groups were initially compared on the basis of discrete demographic characteristics using chi-square analysis. The groups were also compared according to age, academic grade point average, and month of judicial experience by using t-test procedures. A level of significance, p. = .05, was established for this study.

The analysis of the demographic variables associated with this study revealed that subjects randomly assigned to the different rape

scenarios were similar in terms of gender, age, ethnic origin, level of religious devotion, type of religion practiced, academic standing, academic major, academic grade point average, fraternity/sorority membership, judicial board experience, rape awareness training received, and level of personal knowledge about rape victims. The only significant difference between the 2 samples was in the area of practicing a religious faith.

The similarity of the 2 samples on the demographic variables demonstrates that confounding variables did not influence the outcomes associated with the completion of the FEM Scale, the Attitudes Toward Rape Questionnaire, and the Adjudication Questionnaire.

#### Major Findings

## Rape Myth Belief

The scores on the Unified Rape Myth Factor ranged from a low of 53 to a high of 103, demonstrating a full spectrum of rape myth belief within the sample. A t-test yielded a significant difference between males and females (t = 3.33, df = 80, p = .001). Specifically, females (M = 87.74) in the sample believed significantly less in rape myths than did males (M = 81.37). This result was consistent with the literature (Barnett and Feild, 1977; Koss and Dinero, 1988; Koss, Leonard, Breezley, and Oros, 1985).

## Gender-Role Attitudes

The FEM scores ranged from a low of 60 to the highest score of 100, reflecting a variety of gender-role beliefs. A t-test determined that females (M = 85.55) in the sample were significantly (t = 3.43, df = 80, p = .001) more feminist in orientation than were males (M = 78.01). This finding was consistent with the literature (Garrett-Gooding and Senter, 1987; Horsley, 1988).

Previous knowledge of a rape victim (t = -2.81, df = 79, p = .006)

÷

was also significant. People who knew a rape victim had significantly higher mean scores on the FEM Scale than those who did not know a rape victim (84.05 v. 77.33).

## **Verdict**

All 82 subjects of this investigation determined that the male was guilty of rape. Of special note:

- a) Forty-one subjects judged the acquaintance rape scenario, with the remaining 41 subjects judging the stranger rape scenario.
- b) Male respondents (n = 25) were more represented in the sample that judged the acquaintance rape scenario than were female respondents (n = 16). The reverse was true for the respondents that judged the stranger rape scenario, with women (n = 24) more represented than men (n = 17).

The findings of this portion of the study are contrary to the vast majority of the literature (See Chapter 2). In general, the literature states that gender (Thornton, Ryckman, and Robbins, 1982; Tieger, 1981), level of belief in rape myths (Krulewitz and Payne, 1978; Mazelan, 1980; Shotland and Goodstein, 1983), type of rape judged (LaFree, 1980; McGraw, 1987), and traditionality of gender-role beliefs (Acock and Ireland, 1983; Williams, 1979) influences the definitional boundaries of rape. None of these variables, nor any of the other demographic variables associated with this investigation, proved to be significant in the selection of the verdict.

#### Sanctions

Six judicial sanctions were available for selection as punishment when the male was viewed as guilty of rape. The sanctions ranged from a minor "No Action" to the more severe "Expulsion/Dismissal" from the institution:

a) No Action: No formal sanction or action is undertaken.

- b) Warning or Reprimand: An official written statement that expresses disapproval of the conduct.
- c) Censure: A sanction that indicates that further violations during the specified time period for the censure will result in more severe disciplinary action, including probation, suspension, or expulsion.
- d) Probation: A sanction that indicates that further violations during the specified time period for the probation will result in more severe disciplinary action, including suspension or expulsion.
- e) Suspension: A sanction that ends a student's enrollment and removes him/her from campus for a specified period of time.
- f) Expulsion or Dismissal: A sanction that ends a student's enrollment and removes him/her from campus for an indefinite (often permanent) period of time.

The literature (LaFree, Reskin, and Visher, 1985) generally maintains that variables such as level of rape myth belief and gender-role attitudes influences the level of punishment granted. Another prevailing theme is represented by Bohmer and Parrot (1993, p. 191) who believe that "historically, when hearing boards have found assailants guilty of rape, the defendants have been given very light sentences, such as community service and counseling." Thus it was reasonable to expect a normal distribution of sanctions, with perhaps a lenient skew.

The results were contrary to the aforementioned. Of the 81 subjects that selected a sanction, 54 (67%) chose to expel/dismiss the male from the institution. Twenty-one (26%) of the subjects chose to suspend the male from the college. The remaining 6 (7%) subjects chose to place the male on probationary status. No sanctions less than probation were selected.

Overall, the punishments selected were clearly on the severe end of the spectrum. However, the sanctions assigned in the acquaintance rape situation were more lenient. For example, 34 (63%) of the expulsions represented judgements of the stranger rape condition, while 20 (37%) were assigned in the acquaintance rape situation. Of the suspensions, 6 (29%) were in response to the stranger rape and 15 (71%) were judgements in the acquainted condition. All 6 probations were for the assailant in the acquaintance rape situation.

## Hypotheses

Eleven hypotheses were originally developed for testing at the p = .05 level of significance. The hypotheses were directionally stated in accordance with the consistent outcomes found in the literature regarding the influence of gender-role attitudes and rape myth belief. Specifically, the literature stated that a person's gender-role attitudes were related to the level of belief in rape myths and to the amount of blame and responsibility assigned to female rape victims (Koss and Dinero, 1988; Koss, et al., 1985). The level of rape myth belief also influences the definitional boundaries of rape (Wilson, Faison, and Britton, 1983). Furthermore, acquaintance rape situations were often viewed less seriously than stranger rape cases (Bohmer, 1974).

Statistical analyses were chosen with the thought that the subjects participating in this investigation would choose both the guilty and not guilty verdicts. Subsequent findings limited the statistical analyses to t-tests, chi-squares, and correlation coefficients.

#### Hypothesis 1

Hypothesis 1 states that as a group, subjects with high level belief in rape myths will most frequently select the not guilty verdict in the acquaintance rape mode. This hypothesis was not supported because no subjects selected the not guilty verdict. No further analysis was required.

## Hypothesis 2

Hypothesis 2 states that as a group, subjects with low level belief in rape myths will most frequently select the guilty verdict in the acquaintance rape mode. This hypothesis was supported. However, caution is warranted because guilt was selected by all subjects regardless of their level of rape myth belief. No further analysis was required.

•

num Num

•

••

· · ·

#### Hypothesis 3

Hypothesis 3 states that as a group, subjects with high level belief in rape myths will most frequently select the guilty verdict in the stranger rape mode. This hypothesis was supported. However, caution is appropriate because guilt was selected by all subjects irrespective of their level of rape myth belief. No further analysis was required.

#### Hypothesis 4

Hypothesis 4 states that as a group, subjects with low level belief in rape myths will most frequently select the guilty verdict in the stranger rape mode. This hypothesis was supported. Caution is advised, however, because guilt was selected by all subjects regardless of their level of rape myth belief. No further analysis was required.

## Hypothesis 5

Hypothesis 5 states that as a group, subjects with sexist beliefs will most frequently select the not guilty verdict in the acquaintance rape mode. This hypothesis was not supported because no subjects selected the not guilty verdict. No further analysis was required.

## Hypothesis 6

Hypothesis 6 states that as a group, subjects with feminist beliefs will most frequently select the guilty verdict in the acquaintance rape mode. This hypothesis was supported. Caution is warranted because guilt was selected by all subjects regardless of their gender-role beliefs. No further analysis was required.

## Hypothesis 7

Hypothesis 7 states that as a group, subjects with sexist beliefs will most frequently select the guilty verdict in the stranger rape mode. This hypothesis was supported. Caution is appropriate because

guilt was selected by all subjects irrespective of their gender-role beliefs. No further analysis was required.

#### Hypothesis 8

Hypothesis 8 states that as a group, subjects with feminist beliefs will most frequently select the guilty verdict in the stranger rape mode. This hypothesis was supported. Caution is advised because guilt was selected by all subjects regardless of their gender-role beliefs. No further analysis was required.

## Hypothesis 9

Hypothesis 9 states that overall, there will be a positive relationship between gender-role attitudes and rape myth belief. A correlation of the scores on the FEM Scale with the Unified Rape Myth Factor yielded a significant correlation (r = .6190. p = .01). This hypothesis was supported.

## Hypothesis 10

Hypothesis 10 states that overall, gender-role attitudes will be more influential than rape myth belief in determining the verdict in the acquaintance rape mode. This hypothesis was not supported as all subjects selected the guilty verdict regardless of their level of rape myth belief or gender-role attitudes. No further analysis was required.

## Hypothesis 11

Hypothesis 11 states that overall, gender-role attitudes will be more influential than rape myth belief in determining the verdict in the stranger rape mode. This hypothesis was not supported as all subjects selected the guilty verdict regardless of their level of rape myth belief or gender-role attitudes. No further analysis was required.

#### Limitations of the Study

This section focuses on factors that might be considered inherent weaknesses in the design of the study. The limiting factors are identified below and should be accounted for when interpreting the results of this study.

## Limitations Associated With Adjudication Research

Diamond and Zeisel (1974) determined that mock juries were more likely to assume a different standard of reasonable doubt than actual juries, thus arriving at guilty verdicts more often. They also found that judges more often agreed with the decisions of mock juries than with real juries, in part because real juries were more cautious in determining a verdict.

Kerr, Nerenz, and Herrick (as cited in Gerbasi, Zuckerman, and Reis, 1977) found no significant difference between the predeliberation verdicts of mock and real college judicial board members. Kerr et al. also found that group verdicts between mock and real judicial boards did not differ, although real jurors were more lenient. This study also determined that females were generally more lenient than males.

The studies above demonstrate that real juries or judicial boards are less likely to find guilt and more lenient in assigning punishment than subjects in mock conditions. Since all the subjects in this investigation were actual judicial board members, one should be aware of the aforementioned tendencies when interpreting results.

## Limitations Associated With Social Phenomenon Research

According to Babbie (1990, p. 119), "... most concepts of interest to social researchers have no real meaning, no ultimate definitions."

That is because the concepts are often summaries of complex phenomenon that are only "... oversimplified, inevitably superficial, empirical indicators" (p. 119). Thus Babbie (1990) believes that researchers "...

never collect data, they create data" (p. 119) that provides artificial ranking and grouping of the respondents. As a result, one "... can never make accurate measurements [of complex social phenomenon], only useful ones" (Babbie, 1990, p. 120).

Further shortcomings are associated with the utilization of questionnaires to gather data. For example, researchers must be aware that "carelessness, faulty memory, faulty perception, and lack of interest [on the part of respondents] may adversely affect the quality of responses. Furthermore, there can be little assurance that all of the respondents will be truthful" (Turney and Robb, 1971, p. 130). Researchers must also consider that "... respondents might express attitudes even though they have never given any thought to the issue ..." (Babbie, 1990, p. 129).

The previously addressed limitations are important considerations for interpreting the findings of this investigation. Specifically, the utilization of the FEM Scale and the Attitudes Toward Rape Questionnaire to determine beliefs about gender roles and rape dynamics warrants caution given the complexity of measuring social phenomenon. However, it must be noted that Check and Malamuth (1985, p. 418) found "... that scales measuring rape myth acceptance and related attitudes do in fact predict responses to both fictional and real rape descriptions" Thus "... there is cross-situational generalizability with respect to the interrelationships of rape-related perceptions, behaviors, and attitudes."

## Limitations Associated With Judicial Descriptors

All institutions participating in this study used slightly different descriptors to explain their conduct rules and judicial sanctions. As such, the descriptors explaining the judicial sanctions in Section C (Adjudication Questionnaire) of the survey were a compilation. Interpretation of the results assumed that all respondents understood the descriptors before determining a punishment.

## Limitations Associated With Case Studies

The use of rape scenarios has limitations as well. Specifically, Babbie (1990, p. 33) believes that "whereas most research attempts to limit the number of variables considered, the case study seeks to maximize them." As a result, the case study method limits the generalizability of the results beyond the case study being analyzed.

LaFree (1989, pp. 153-154) believes "... that simulations, no matter how carefully designed, remain less compelling than real trials - a difference of particular importance in cases involving especially violent, personal crimes like rape." Case studies also limit subject exposure to extra-legal variables (defendant/victim characteristics) that influence outcomes of hearings (Gerbasi, Zuckerman, and Reis, 1977).

## Generalizability of the Results

According to Babbie (1990) the goal of social science research is "
... to expand the explanatory power of our findings to other forms of
behavior and other subsets of the population" (p. 25). Thus, "... the
sample must be 'typical' or representative of that population" (Turney
and Robb, 1971, p. 107).

This investigation did not utilize random sampling techniques (Scheaffer, Mendenhall, and Ott, 1990) to draw inferences about a larger population. Rather, the entire population of primary judicial board members enrolled at 15 selected higher education institutions in Michigan were surveyed. Primary boards were defined as those empowered to suspend or recommend suspension for students found guilty of violating conduct rules.

A 79 percent return rate (n = 67) from board members at stateassisted colleges makes it very likely that the final sample was representative of this population. Less representative was the 54 percent return rate (n = 15) demonstrated by board members enrolled at independent colleges. Overall, a 73 percent return rate (n = 82) makes it appropriate to generalize the results of this investigation as being representative of the primary judicial boards at the participating institutions.

The judicial systems used by the 15 participating institutions were fairly typical of those generally defined in the literature (Stoner and Cerminara, 1990). The experience of this investigator over 14 years of working with judicial boards at 4 colleges also suggests that these judicial systems were representative. Unfortunately, no published research exists (to the knowledge of this researcher) concerning the influence of rape myths on the adjudication of acquaintance rape cases by judicial boards. Thus it is difficult to determine, even given the general similarity of many judicial systems, how representative the results of this study are.

Although tempting given the aforementioned, the results of this research should not be generalized beyond the stated parameters. The reasons for this conclusion are highlighted as follows:

- a) All the respondents were drawn from institutions located in Michigan and thus may not be representative of judicial board members attending institutions in other states.
- b) While the judicial systems at the institutions represented in this study were relatively similar, they may not be representative of all college judicial systems in the United States.
- c) The criminal rape laws in states other than Michigan may be different, thus influencing the type of advising and/or training judicial board members receive.
- d) This investigation had fewer respondents representing independent institutions, making it more difficult to generalize to this type of college.

# Discussion of the Major Findings

The purpose of this section is to offer speculation concerning the outcomes of this research. The speculation should be viewed as circumstantial and is offered to assist the reader with interpreting results and developing potential conclusions.

·
·
·
·
·

%° € 3 The findings of this investigation were surprising in that all respondents chose the guilty verdict when judging the scenarios. The guilty verdict was selected regardless of: (a) the type of scenario judged; (b) the level of rape myth belief; (c) the attitudes toward gender roles; and (d) the gender of the respondent. No other demographic variables exerted significant influence on the determination of the verdict. Furthermore, the respondents clearly selected judicial sanctions from the severe end of the punishment continuum, indicating disapproval of the act.

The aforementioned results were contrary to the general findings of the literature. Thus it seemed clear that a mitigating factor(s) was influencing the outcomes of this research. Further review of the literature generally failed, with 1 notable exception, to offer substantial explanation for the unexpected results. The notable exception was the concept of pluralistic ignorance (Allport, 1924; Katz and Allport, 1931).

# Pluralistic Ignorance

Pluralistic ignorance explains situations where "... members of a group privately reject group norms yet believe that virtually all other group members accept them" (Miller and McFarland, 1987, p. 298).

Generally, pluralistic ignorance was postulated as a way to explain "... widespread conformity to social norms in the absence of widespread private support. [The premise is that] for a norm to be perpetuated, it is not necessary for the majority to support it, only for the majority to believe that the majority supports it" (Miller and McFarland, 1987, p. 298).

Breed and Ktsanes (1961, p. 383) believe that the concept of pluralistic ignorance encompasses the following elements:

- 1) "A distribution of opinions on an issue exists in any social system, and most members can give an estimate as to the direction or character of this distribution."
- 2) "These estimates may be inaccurate, and this we take to be pluralistic ignorance."

• • • . . 2 • .. . . •

- "The error will tend to favor the older existing beliefs in the system rather than the direction of change."
- "Pluralistic ignorance is correlated with attitudinal bias: the greater the 'inaccuracy', the greater the bias."
- 5) "Pluralistic ignorance will be more pronounced in large groups because there is less intercommunication, so that cognitive reorientation will occur more slowly there."

According to Miller and McFarland (1987, p. 304), pluralistic ignorance has 3 special consequences. "First, pluralistic ignorance can lead a group to act in a manner that is inconsistent with the inclinations of its members.... A second consequence ... is that its victims may actually adopt an erroneous perception not only of the group but of the situation in which the group finds itself.... A third consequence ... manifests itself in the psychological well-being of the individuals experiencing it."

One of the early investigations confirming the existence of pluralistic ignorance was conducted by Breed and Ktsanes (1961). They conducted 2 studies in the city of New Orleans. The first study involved 224 members of a large protestant church congregation, while the second study included 273 subjects drawn from a cross-section of New Orleans. The studies questioned subjects about the topic of segregation and determined that people generally ascribed more liberal attitudes to themselves while believing that others were more pro-segregation.

O'Gorman (1975) conducted a secondary examination of a 1968

National Election Study involving 1,408 white subjects. The original study asked respondents about their views on racial segregation.

Additionally, subjects were asked to estimate other's views related to segregation. The results clearly demonstrated that segregationists were repeatedly perceived to exist in much greater numbers than was true.

O'Gorman (1975) concluded that "... most white Americans grossly exaggerated the extent to which other whites supported racial segregation" (p. 322).

Prentice and Miller (1993) examined the relationship between college students' own attitudes toward alcohol use as compared to their

estimates of the attitudes of their peers. A series of 4 separate studies involving a total of 518 students concluded that:

- a) "... Students' comfort with alcohol use on campus ... manifest the classic characteristics of pluralistic ignorance.... [Generally] undergraduates believe that everybody is more comfortable with drinking than they are themselves" (p. 247).
- b) "... Men behaved in the way social influence theorists would expect: They changed their own attitudes toward drinking in the direction of the social norm. Women, on the other hand, showed no change in attitudes over time" (p. 248).
- c) "These results provide clear evidence that people who feel deviant from the norm of their social group are inhibited from acting [in a contrary manner]" (p. 251).
- d) "In short, students were victims of pluralistic ignorance:
  They believed that the private attitudes of other students
  were much more consistent with campus norms than were their
  own" (p. 252).

Prentice and Miller (1993) offered several explanations for the discrepancy between students' personal attitudes and those attributed to peers. The first possibility was that "... students might have hoped to convey the impression ... that they were more mature, more progressive, or more enlightened than their peers" (p. 252). A second possibility suggests that subjects are misrepresenting their own views and failing to recognize that others may not be truthful in communicating opinions. Third, "... students may fail to recognize how pronorm their public behavior actually is, mistakenly believing that their private discomfort with alcohol practices is clear from their words and deeds" (p. 253). Finally, "through an accurate reading of a biased distribution of publicly expressed opinions, students may have been led to erroneous perceptions of their peers' attitudes" (p. 253).

The aforementioned final point is especially relevant. For example, colleges are becoming more intentional about addressing social issues affecting students. A study conducted by the University of Puget Sound (as cited by Bash, 1993) determined that over 90 percent of liberal arts colleges will include seminars on social issues in fall orientation programs. The fastest growing component of orientation seminars are those dealing with sexual harassment and sexual assault.

In fact, a recent national survey (as cited by Bash, 1993) found that 48 percent of the mid-size institutions offered date rape seminars at orientation, a percentage almost double what it was 2 years ago.

Colleges are also presenting more on-going programs that address issues of sexuality, especially as they relate to acquaintance rape (Buhrke and Lustgraaf, 1988; Miller, 1988; Orzek, 1983; Pace and Zaugra, 1988; Sandberg, Jackson, and Petretic-Jackson, 1987). According to Miller (1988, p. 554), the goals of these programs are to:

- a) "To make students aware that the problem of date rape exists and that it can happen to them."
- b) "To educate them as to the role alcohol and drug use can play in creating an environment for undesirable sexual experiences."
- c) "To impress on students that their own health and sexual well-being are their responsibility and that they need to communicate assertively and clearly in the area of sexual relationships."
- d) "To let students know that if they do encounter a problem, there are places where they can go for help."

Colleges have also developed and publicized the existence of Rape Victim Support Services to respond to the needs of rape victims (Holmes and St. Lawrence, 1983; Koss and Harvey, 1987; Orzek, 1983). Mental health and medical personnel on many campus are also more actively involved in efforts to both prevent and respond to rape situations (Burkhart, 1991; Larrabee and McGeorge, 1989; Orzek, 1983; Parrot, 1991c). Furthermore, some college police forces like those at Michigan State University and The Ohio State University issue sexual assault response guarantees that encourage victim reporting.

In summary, college students receive constant messages from a variety of college sources about acquaintance rape. The messages generally increase awareness of the phenomena and communicate values that are more feminist or anti-rape. Students also receive messages concerning the topic of acquaintance rape from the media, including television and print sources (Freeman, 1991; Gibbs, 1991; Kalette, 1988; Sherman, 1985). The highly publicized acquaintance rape cases of boxer Mike Tyson and medical student William Smith communicated information to

## students as well.

Judicial board members, in addition to exposure to the aforementioned, may be subject to anti-rape messages from a variety of sources. For example, it is possible that judicial board members were influenced by a combination of the following: (a) Judicial board advisors may have communicated anti-rape and feminist values; (b) Exposure to training designed to increase awareness about sexual assault and rape issues; and/or (c) Actual experience hearing a rape case.

It is very probable that judicial board members were subjected to a biased information set about rape that influenced their perception of other member's opinions. Thus pluralistic ignorance is a real possibility when explaining why judicial board members selected guilty verdicts, even though a normal distribution of rape myth and gender-role attitudes existed within the sample.

## Standard of Evidence

Campus discipline hearings may use less stringent "by a preponderance" or "clear and convincing" standards of proof to judge student behavior that violates college rules (Bohmer and Parrot, 1993; Long, 1985). According to Bohmer and Parrot (1993, p. 188), "because the rules of evidence are much more flexible on a college campus than in the criminal courts, a simple sexual assault is more likely to result in a guilty verdict in the campus system. Therefore, more assailants may be punished if acquaintance rape cases are heard by the college judicial board or officer rather than (or in addition to) the criminal courts, provided that the college system is well designed and administered."

The aforementioned was confirmed by a judicial affairs officer at Michigan State University (personal communication, academic year 1992-93). It was this person's opinion that judicial boards almost always found the male guilty of acquaintance rape when a hearing was held. The premise for this opinion rested with the assertion that only cases with strong evidence were generally pursued by the victims. Also, the

•
.

,

. .

f :

•

.<del>3</del>

.

judicial officer believed the less stringent standards made it easier to find guilt.

In summary, both the concept of pluralistic ignorance and the standards of evidence used in judicial hearings may be explanatory factors for the results of this study.

### Conclusions

The following conclusions may be drawn from the results of this investigation. The conclusions are not irrefutable, having been derived within the context of the study's limitations. Also considered were the speculative factors previously addressed.

- a) Subjects' gender-role attitudes were not significantly influential in selecting a verdict in either the acquainted or stranger conditions.
- b) Subjects' level of rape myth belief was not significantly influential in selecting a verdict in either the acquainted or stranger conditions.
- c) In this investigation it was determined that a positive significant correlation existed between subjects' level of rape myth belief and gender role attitudes. This combination, however, did not significantly influence the selection of a verdict in either the acquainted or stranger condition.
- d) Males were significantly less feminist and believed significantly more in rape myth than did females. However, gender was not a significant factor in explaining the selection of the verdict in either the acquainted or stranger conditions.
- e) The judicial sanctions chosen as punishment were severe, indicating that rape, as portrayed in both conditions, was viewed seriously regardless of gender, gender-role

attitudes, or level of rape myth beliefs.

#### Theoretical Implications

The literature clearly stated that traditional gender-role attitudes correlated positively with belief in rape myths (Briere and Malamuth, 1983; Check and Malamuth, 1983; Costin, 1985; Muehlenhard and Hollabaugh, 1988). Traditional gender-role attitudes were related to increased assignment of female blame for the occurrence of rape (Acock and Ireland, 1983; Coller and Resick, 1987; Howard, 1984a; Klemmack and Klemmack, 1976; Muehlenhard and Hollabaugh, 1988; Shotland and Goodstein, 1983; Williams, 1979). People with traditional gender-role orientations also believed that forced sexual intercourse was normal (Check and Malamuth, 1985; Goodchilds, et al., 1988; Muehlenhard, 1988).

Research also suggests that the level of belief in rape myths influences the definitional boundaries between rape and normal sexual intercourse (Cann, Calhoun, and Selby, 1979; Dermer and Thiel, 1975; Howard, 1984a, 1984b; Klemmack and Klemmack, 1976; Krulewitz and Nash, 1979; Krulewitz and Payne, 1978; McGraw, 1987; Shotland and Goodstein, 1983; Skelton and Burkhart, 1980; Thornton, 1977; Tieger, 1981). People were also found to attribute significantly greater responsibility to the rapist when the assailant and victim were strangers (Bohmer, 1974; Chappell and Singer, 1977; McGraw, 1987; Chappell and Singer, 1977; Kerstetter, 1990; LaFree, 1980; Wilson, 1978).

Finally, research has determined that gender is a very important variable in determining the formation of gender-role attitudes and rape myth beliefs. Males generally believe in more traditional gender-roles and rape myths, thus making them more prone to using sexual aggression (Briere and Malamuth, 1983; Check and Malamuth, 1983; Dull and Giacopassi, 1987; Fisher, 1986; Garrett-Gooding and Senter, 1987; Greendlinger and Byrne, 1987; Horsley, 1988; Koss and Dinero, 1980; Mahoney, Shively, and Traw, 1986; Malamuth, 1981; Malamuth and Check,

1980; Malamuth, Haber, and Feshback, 1980; Quackenbush, 1987).

The general trend of the research, as briefly summarized above, is very consistent. Additionally, research has determined that jury members are influenced by their gender-role attitudes and belief in rape myths more than by the actual evidence of a case (Barber, 1974; Bohmer, 1974; Feild, 1978b; Feild and Bienen, 1980; Gerbasi, Zuckerman, and Reis, 1977; Kalven and Zeisel, 1966; LaFree, 1980; LaFree, 1989; LaFree, Reskin, and Visher, 1985; McCahill, Meyer, and Fischman, 1979; Reskin and Visher, 1986; Scroggs, 1976).

The results of this study were consistent with the literature. For example, the subjects demonstrated a full spectrum of rape myth belief and gender-role attitudes. Males were significantly different than females in terms of the aforementioned. However, the results also contradicted the literature in that all the subjects found the male assailant guilty (regardless of the type of rape judged) and assigned harsher sanctions as punishment.

The theoretical implication, then, is the premise that subjects' rape myth beliefs and gender-role attitudes in and of themselves did not exert significant influence over their judgement of rape cases. Thus, the judicial board members were influenced by other dynamics or factors that were strong enough to overcome the affects of their personal belief systems. The influences were consistent across gender, type of college attended, racial/ethnic status, religion, age, major, grade point average, fraternity/sorority status, judicial board experience, amount of rape awareness training received, and previous knowledge of a rape victim.

#### Practical Implications

On a practical level, this research questions the assumption that judicial boards are ill-prepared to adjudicate acquaintance rape cases (Collison, 1991; Julian, 1992; Kalette, 1988; Parrot, 1991a, 1991b). In

• . <del>.</del> . . • **)** .

fact, the results support Pavela (1992a) who believes that judicial boards are capable of rendering sound judgements when adjudicating acquaintance rape cases.

During a telephone conversation with a Dean of Students (personal communication, March/April 1993), it was explained to this investigator that acquaintance rape cases on his campus were never sent to judicial boards. Although judicial boards were a publicized option for students wishing to use the judicial system, the college denied this right when the case was an acquaintance rape. According to the Dean, this was done to protect the victim and because the college did not want to trust judicial board with this type of decision. The results of this investigation questions this practice as it denies consistent application of the procedures and underestimates the capability of the board members.

It appears, based on the results of this survey, that judicial boards are capable of rendering sound judgements in acquaintance rape cases. This may be especially true when institutions clearly communicate anti-rape messages to all students, properly train judicial board members, and provide competent judicial board advisors.

#### Recommendations for Further Research

The major findings, conclusions, limitations, discussions and implications associated with this investigation suggest that further research is needed. The following areas are suggested:

- a) This investigation should be duplicated using a broader and larger sample of primary judicial board members. The new sample should include judicial board members from a variety of institutions located in all the geographic regions of the United States.
- b) The investigation should include a comparison sample of enrolled students not serving on college judicial boards. The comparison sample should be drawn from the same institutions as the judicial

board members.

- c) The investigation should have judicial board advisors complete the survey instrument. Advisors should be also be surveyed for additional information about the college's: rape awareness programs, rape crisis services, counseling services, orientation programs, acquaintance rape occurrences, and judicial board training, etc.
- d) Faculty and staff members serving on primary judicial boards should be asked to complete the survey instruments.
- e) The investigation should consider utilizing a variety of acquaintance rape scenarios written in a manner allowing researchers to control for specific variables, e.g. alcohol consumption, resistance, force, etc.
- f) Rather than using written scenarios, the use of video tapes featuring competent actors portraying rape situations should be considered.
- qualitative research methods. In addition to completing a survey instrument, structured interviews with select judicial board members may be beneficial. If possible, interviews with victims and assailants who have appeared before judicial boards for adjudication of a rape case would be illuminating.

#### Concluding Commentary

The results of this study were surprising. The fact that all the judicial board members judged the male as guilty was unexpected. This was especially true given: (1) the general thrust of the literature; and (2) the results of a pilot study involving 11 residence hall judicial board members at Michigan State University. Specifically, the pilot study yielded a diversity of opinion regarding the guilt of the male in the acquaintance rape situation.

The findings of this investigation were encouraging, however, with regard to the competence of judicial boards effectively and fairly adjudicating campus rape cases. Never-the-less, the results remain preliminary and suggest the importance of additional research. Until such time, one cannot discount the possibility that these results are not a reflection of reality. That is, concerns about the influence of rape myths and gender-role attitudes on the adjudication of campus rape cases should remain alive and well.

### APPENDICES

### APPENDIX A

### Survey Instrument With Acquaintance Rape Scenario

(Format Not a True Facsmile)

#### Dear Judicial Board Member:

This survey is being given to judicial board members at many colleges and universities in Michigan. The purpose of this survey is to analyze the decision-making process of campus judicial boards. Specifically, this survey deals with an assault incident, an occurrence that is becoming more common on many campuses today. The insight gained from this survey will benefit judicial board members, like yourself, by contributing to the development of training programs and advising services.

Your participation in this study is important, yet voluntary. You may choose not to participate at all. You have the right to refuse to answer any questions. However, I would appreciate your answering all the questions in order to minimize the amount of missing information. Your responses will be kept strictly confidential and will not be analyzed individually.

This survey should take approximately 12-16 minutes to complete. Instructions for completing this survey are located in the next section.

If you have any questions about this project or you wish to receive a copy of the results, please contact me at the following address or phone number:

Department of Residence Life G-64 Wilson Hall Michigan State University E. Lansing, MI 48825-1208 (517) 336-2492

The return of this survey constitutes your informed and voluntary consent to participate in this research.

Thank you for your valuable participation in this project!

Sincerely,

Roger L. Serr

## **SECTION A**

#### **GENERAL INSTRUCTIONS**

The remainder of this survey contains Sections B through F. Directions for completing each section are provided in the introduction to that particular section. If the directions are not clear, please contact Roger Serr at the previously listed phone number or address.

Please answer all questions according to your personal beliefs. It is important to be candid and honest. As a reminder, your individual responses are strictly confidential and will be reported only as part of the larger group of responses.

Your participation in this survey is voluntary and you may choose not to answer some questions.

Please, however, consider answering all the questions to assist in the analysis of the survey's results.

After completing this survey, please seal it in the attached pre-addressed, stamped envelope and return it to the person from your institution that administered the survey to you. If you desire, you may return the survey directly to Roger Serr.

It is important that you return this completed survey by April 1, 1993.

### **SECTION B**

Directions: The following is a college dating scenario. Please read the scenario and then proceed to Section C.

Lee walked into the lobby of the residence hall and picked up the phone. He called Diane's room. When she answered, he told her he was waiting for her downstairs to take her to the movie. While waiting, he checked himself in the mirror. His 5-foot-10 inch, 160-pound frame fit well into the sport jacket and jeans he had donned for the occasion. Diane walked down the steps and over to Lee. Lee smiled and said that he really liked the green dress she had bought last week. She was 5-foot-2-inches and weighed 100 pounds.

As they walked to the car, Diane said she'd wanted to see this movie for a long time. As they drove to the theater they talked about their mutual friends and the party last weekend. Diane and Lee had met two months earlier and had seen each other a couple of times at first, and then every weekend for the past month. They each continue to date others on occasion.

After parking the car, the couple waited in line, making small talk until the ticket window opened. Lee bought the tickets and they went inside. They were spellbound by the movie; neither talked until the film was over. After the movie Lee suggested that they go back to his residence hall suite where they could listen to music, drink some wine, and talk. Diane said "okay."

Lee's suite opened onto a landscaped campus courtyard surrounded by many similar suites. Lee and Diane walked slowly through the courtyard, enjoying the night air, glancing toward each other from time to time.

When they got to his suite, Lee put on an a compact disc and poured some wine for both of them. They sat on the couch for a while, listening to music and talking. As they were talking their eyes would meet and then both would quickly look away. The fourth time their eyes made contact, Diane and Lee held their gaze and smiled. Lee moved closer to Diane, put his arm around her and gently stroked her shoulder. He kissed her softly.

Lee put both arms around Diane and held her close to him. He kissed her again, longer this time, and then opened his mouth slightly so that his tongue touched hers. He continued to kiss her like this for a while.

Lee slid his hand inside Diane's dress and began to fondle her breast; with the other hand he started unbuttoning the dress. Diane removed Lee's hand from her breast and said "Please don't." Lee continued and soon managed to finish unbuttoning Diane's dress and he slipped it off her shoulders. Kissing her so that their mouths were in continuous contact, he stroked her breasts rhythmically and then rubbed the inside of her thighs. Lee kissed Diane's breasts and stomach and touched her genital area. Then he slid her dress completely off and removed her underwear. They continued to kiss each other.

The phone rang, Lee answered. Just a wrong number; Lee hung up and returned, sitting next to Diane. With Diane totally naked, Lee leaned against her and pushed her back onto the couch until he was on top of her. Diane said, "No, Lee, don't." Ignoring this, Lee responded "It's okay," and quickly unzipped his pants and slid them down. Diane struggled and said, "I don't want to, let me go!" "Relax Diane, don't worry," Lee answered.

Diane protested once more, "Don't! Stop!" Lee held Diane and said, "Don't worry, I'll take care of everything." He stroked her breasts, "Relax, just take it easy," he said. Lee continued to kiss and fondle Diane. Soon, he penetrated her and intercourse occurred.

form a

## **SECTION C**

Directions: Please read the scenario in Section B prior to responding to the following statements. Based only on the information provided in the scenario, please answer Statement 1 as a judicial board member judging the man's guilt or innocence. Depending upon your response to Statement 1, you may be instructed to respond to Statement 2.

(1) I believe	the man was (Place a check mark by your one choice):
No	t Guilty of rape.
Gu	ilty of rape.
Please respond	to Statement 2 only if you indicated in Statement 1 that the man was guilty of rape.
•	believe the man was guilty of rape, please select one of the following disciplinary unishments that best fits the offense. Please read all the descriptions before deciding.
(2) I believe	the man should receive (Place a check mark by your one choice):
	No Action - No formal sanction or action is undertaken
	Warning or Reprimand - An official written statement that expresses disapproval of the conduct.
	Censure - A sanction that indicates that further violations during the specified time period for the censure will result in more severe disciplinary action, including probation, suspension, or expulsion.
	Probation - A sanction that indicates that further violations during the specified time period for the probation will result in more severe disciplinary action, including suspension or expulsion.
	Suspension - A sanction that ends a student's enrollment and removes him/her from campus for a specified period of time.
	Expulsion or Dismissal - A sanction that ends a student's enrollment and removes him/her from campus for an indefinite (often permanent) period of time.

## **SECTION D**

Directions: The following is a list of 20 statements. Please respond to each of the statements by circling the response that most identifies your feelings about the item. Respond to each statement by using the following scale:

- 1 = Strongly Agree (SA)
- 2 = Agree(A)
- 3 = No Opinion (N)
- 4 = Disagree (D)
- 5 = Strongly Disagree (SD)

1.

2.

3.

4.
5.
6.
7.
8.
9.
10.
11.
12.
13.
14.

		SA	A	N	D	SD
1. It is all rig be the basi	ht for women to work but men will always c breadwinners.	1	2	3	4	5
2. A woman or have the	should not expect to go to the same places same freedom of action as a man.	1	2	3	4	5
3. Realisticall made by n way.	y speaking, most progress so far has been nen and we can expect it to continue that	1	2	3	4	5
4. A woman when she r	should be expected to change her name narries.	1	2	3	4	5
typically fi	tho join the Women's Movement are ustrated and unattractive people who feel ut by the current rules of society.	1	2	3	4	5
	the household, the father should have final ver his children.	1	2	3	4	5
7. A woman with her hubroke up.	who refuses to give up her job to move usband would be to blame if the marriage	1	2	3	4	5
8. Profanity s woman.	ounds worse generally coming from a	1	2	3	4	5
	who refuses to bear children has failed in her husband.	1	2	3	4	5
10. Women are	basically more unpredictable than men.	1	2	3	4	5
_	ng vine" wife is justified provided she ly enough to please her husband.	1	2	3	4	5
12. One should woman.	never trust a woman's account of another	1	2	3	4	5
•	voman who sends her six month old baby center is a bad mother.	1	2	3	4	5
	uld not be permitted to hold political nvolve great responsibility.	1	2	3	4	5
	ele that women be appointed to police he same duties as men.	1	2	3	4	5
16. Women have sphere of act	the light to compose with men in con-	1	2	3	4	5

Section D, Continued	SA	Α	N	D	SD
17. A woman to be truly womanly should gracefully accept chivalrous attentions from men.	1	2	3	4	5
18. The unmarried mother is morally a greater failure than the unmarried father.	1	2	3	4	5
19. It is absurd to regard obedience as a wifely virtue.	1	2	3	4	5
20. Whether or not they realize it, most women are exploited by men.	1	2	3	4	5

•

..

# **SECTION E**

Directions: The following is a list of 32 statements. Please respond to each of the statements by circling the response that most identifies your feelings about the item. Respond to each statement by using the following scale (Please note that this scale differs from that used in Section D):

- 1 = Strongly Agree (SA) 2 = Agree (A)
- 3 = Slightly Agree (SLA)
- 4 = Slightly Disagree (SLD)
- 5 = Disagree (D)
- 6 = Strongly Disagree (SD)

	SA	Α	SLA	SLD	D	SD
1. A woman can be raped against her will.	1	2	3	4	5	6
The reason most rapists commit rape is for the thrill of physical violence.	1	2	3	4	5	6
3. Rapists are "normal" men.	1	2	3	4	5	6
4. In forcible rape, the victim never causes the crime.	1	2	3	4	5	6
5. All rapists are mentally sick.	1	2	3	4	5	6
<ol> <li>A charge of rape two days after the act has occurred is probably not rape.</li> </ol>	1	2	3	4	5	6
<ol> <li>A woman should be responsible for preventing her own rape.</li> </ol>	1	2	3	4	5	6
8. A man who has committed rape should be given at least 30 years in prison.	1	2	3	4	5	6
<ol><li>Women are trained by society to be rape victims.</li></ol>	1	2	3	4	5	6
10. A raped woman is a less desirable woman.	1	2	3	4	5	6
11. If a woman is going to be raped, she might as well relax and enjoy it.	1	2	3	4	5	6

Section E, Continued	SA		SLA	SLD		SD
12. Rape provides the opportunity for many rapists to show their manhood.	1	2	3	4	5	6
13. Most women secretly desire to be raped.	1	2	3	4	5	6
14. It would do some women some good to get raped.	1	2	3	4	5	6
<ol> <li>Women provoke rape by their appearance or behavior.</li> </ol>	1	2	3	4	5	6
16. "Nice" women do not get raped.	1	2	3	4	5	6
17. Most charges of rape are unfounded.	1	2	3	4	5	6
18. In order to protect the male, it should be difficult to prove that a rape has occurred.	1	2	3	4	5	6
<ol> <li>Rape is the expression of an uncontrollable desire for sex.</li> </ol>	1	2	3	4	5	6
20. Rape is the worst crime that can be committed.	1	2	3	4	5	6
21. Rape is a sex crime.	1	2	3	4	5	6
22. All rape is a male exercise in power over women.	1	2	3	4	5	6
23. During a rape, a woman should do everything she can do to resist.	1	2	3	4	5	6
24. Rapists are sexually frustrated individuals.	1	2	3	4	5	6
25. In most cases when a woman was raped, she was asking for it.	1	2	3	4	5	6
26. The reason most rapists commit rape is for sex.	1	2	3	4	5	6
27. Rape of a woman by a man she knows can be defined as a "woman who changed her mind afterward".	1	2	3	4	5	6
28. A convicted rapist should be castrated.	1	2	3	4	5	6
29. A woman should not feel guilty following a rape.	1	2	3	4	5	6
30. The degree of a woman's resistance should be the major factor in determining if a rape has occurred.	1	2	3	4	5	6
31. A raped woman is a responsible victim, not an innocent one.	1	2	3	4	5	6
32. Rape serves as a way to put or keep women in their "place".	1	2	3	4	5	6

## **SECTION F**

Directions: Please complete the following demographic questions by placing a checkmark by the most appropriate response. This information will be used for research purposes only and individual responses will not be reported. All responses are strictly confidential. 1. What is your gender? Female Male 2. What is your current age? 3. What is your racial or ethnic origin? ☐ Black/African American White/Caucasian Hispanic/Chicano/Latino Indian/Native People Asian/Pacific Islander Other (please specify) 4. Do you practice a religious faith? Yes No If yes: How religiously devout or devoted do you consider yourself? ☐ Extremely devout ☐ Strongly devout ☐ Somewhat devout ☐ Devout ☐ Not **Devout** What spiritual faith do you consider as your religion? Hindu Roman Catholic ☐ Muslim Protestant Other (please specify) Buddhist Jewish 5. What is your current academic class standing? Senior □ Freshman Graduate Student Sophomore Other (please specify) Junior 6. What is your current academic major? \_\_ 7. What is your current academic Grade Point Average? 8. Are you a member of a fraternity or sorority? ☐ Yes

9. How many months have you served as a member of a campus judicial (discipline) board?					
10. How many hearings have you participated in as a judical board member?					
11. Have you ever received any formal training as a judicial board member about rape and/or hearing rape cases?					
Yes No					
If yes: Please check from the following list	at any topics that were presented as part of the training:				
State Laws College Policies	Rape Evidence Requirements				
Judicial Hearing Procedures	☐ Victim's Sexual History/Reputation Standards ☐ Rape Stereotypes and Myths				
☐ Victim Resistance Requirements	Influence of Drugs/Alcohol				
Consent/Nonconsent Issues	Acquaintance and/or Stranger Rape Dynamics				
☐ Use of Force Standards	Testimony/Witnesses Corroboration				
Gender-Role Stereotypes	Other (please specify)				
12. Have you ever participated as a judical boar	rd member in a rape case hearing?				
13. Have you or anyone you know ever been ra	ped? Yes No				
If you answered yes in Question 13, pleas relationship with the victim.	e check from the following list any that explain your				
Relative	Lover				
Acquaintance	☐ I have been raped				
☐ Spouse	Other (please specify)				
☐ Friend					
Please place the completed survey in the attached pre-addressed, stamped envelope. The survey may be returned to the person that administered the survey to you or you may send it directly to:					
Roger Serr  Department of Residence Life  G-64 Wilson Hall  Michigan State University  East Lansing, MI 48825-1208					

The survey should be returned by April 1, 1993.

Thank you!

### APPENDIX B

## Survey Instrument With Stranger Rape Scenario

(Format Not a True Facsmile)

#### Dear Judicial Board Member:

This survey is being given to judicial board members at many colleges and universities in Michigan. The purpose of this survey is to analyze the decision-making process of campus judicial boards. Specifically, this survey deals with an assault incident, an occurrence that is becoming more common on many campuses today. The insight gained from this survey will benefit judicial board members, like yourself, by contributing to the development of training programs and advising services.

Your participation in this study is important, yet voluntary. You may choose not to participate at all. You have the right to refuse to answer any questions. However, I would appreciate your answering all the questions in order to minimize the amount of missing information. Your responses will be kept strictly confidential and will not be analyzed individually.

This survey should take approximately 12-16 minutes to complete. Instructions for completing this survey are located in the next section.

If you have any questions about this project or you wish to receive a copy of the results, please contact me at the following address or phone number:

Department of Residence Life G-64 Wilson Hall Michigan State University E. Lansing, MI 48825-1208 (517) 336-2492

The return of this survey constitutes your informed and voluntary consent to participate in this research.

Thank you for your valuable participation in this project!

Sincerely,

Roger L. Serr

ļ

### **SECTION A**

#### **GENERAL INSTRUCTIONS**

The remainder of this survey contains Sections B through F. Directions for completing each section are provided in the introduction to that particular section. If the directions are not clear, please contact Roger Serr at the previously listed phone number or address.

Please answer all questions according to your personal beliefs. It is important to be candid and honest. As a reminder, your individual responses are strictly confidential and will be reported only as part of the larger group of responses.

Your participation in this survey is voluntary and you may choose not to answer some questions. Please, however, consider answering all the questions to assist in the analysis of the survey's results.

After completing this survey, please seal it in the attached pre-addressed, stamped envelope and return it to the person from your institution that administered the survey to you. If you desire, you may return the survey directly to Roger Serr.

It is important that you return this completed survey by April 1, 1993.

### **SECTION B**

Directions: The following is a college dating scenario. Please read the scenario and then proceed to Section C.

Jim walked into the lobby of the residence hall and picked up the phone. He called Diane's room. When she answered, he told her he was waiting for her downstairs to take her to the movie. While waiting, he checked himself in the mirror. His 5-foot-10 inch, 160-pound frame fit well into the sport jacket and jeans he had donned for the occasion. Diane walked down the steps and over to Jim. Jim smiled and said that he really liked the green dress she had bought last week. She was 5-foot-2-inches and weighed 100 pounds.

As they walked to the car, Diane said she'd wanted to see this movie for a long time. As they drove to the theater they talked about their mutual friends and the party last weekend. Diane and Jim had met two months earlier and had seen each other a couple of times at first, and then every weekend for the past month. They each continue to date others on occasion.

After parking the car, the couple waited in line, making small talk until the ticket window opened. Jim bought the tickets and they went inside. They were spellbound by the movie; neither talked until the film was over. After the movie Jim suggested that they go back to his residence hall suite where they could listen to music, drink some wine, and talk. Diane said "okay."

Jim's suite opened onto a landscaped campus courtyard surrounded by many similar suites. Jim and Diane walked slowly through the courtyard, enjoying the night air, glancing toward each other from time to time.

When they got to his suite, Jim put on a compact disc and poured some wine for both of them. They sat on the couch for a while, listening to music and talking. As they were talking their eyes would meet and then both would quickly look away. The fourth time their eyes made contact, Diane and Jim held their gaze and smiled. Jim moved closer to Diane, put his arm around her and gently stroked her shoulder. He kissed her softly.

Jim put both arms around Diane and held her close to him. He kissed her again, longer this time, and then opened his mouth slightly so that his tongue touched hers. He continued to kiss her like this for a while.

Jim slid his hand inside Diane's dress and began to fondle her breast; with the other hand he started unbuttoning the dress. Diane removed Jim's hand from her breast and said "Please don't." Jim continued and soon managed to finish unbuttoning Diane's dress and he slipped it off her shoulders. Kissing her so that their mouths were in continuous contact, he stroked her breasts rhythmically and then rubbed the inside of her thighs. Jim kissed Diane's breasts and stomach and touched her genital area. Then he slid her dress completely off and removed her underwear. They continued to kiss each other.

The phone rang, Jim answered. He had to leave the suite, telling Diane, "Wait right here, I'll be back in ten minutes." Jim rushed out of the suite, leaving the door ajar as he left.

A moment later, before Diane had time to clothe herself, a man peered through the open door. He must have been walking in the courtyard. The man came toward Diane. With Diane totally naked, the man leaned against her and pushed her back onto the couch until he was on top of her. Diane said, "No, don't." Ignoring this, the man responded "It's okay," and quickly unzipped his pants and slid them down. Diane struggled and said, "I don't want to, let me go!" "Relax, don't worry," the man answered.

Diane protested once more, "Don't! Stop!" the man held Diane and said, "Don't worry, I'll take care of everything." He stroked her breasts, "Relax, just take it easy," he said. The man continued to kiss and fondle Diane. Soon, he penetrated her and intercourse occurred.

form s

## **SECTION C**

Directions: Please read the scenario in Section B prior to responding to the following statements. Based only on the information provided in the scenario, please answer Statement 1 as a judicial board member judging the man's guilt or innocence. Depending upon your response to Statement 1, you may be instructed to respond to Statement 2.

(1)	I believe ti	he man was (Place a check mark by your one choice):
	Not	Guilty of rape.
	Guil	ty of rape.
Pica	se respond	to Statement 2 only if you indicated in Statement 1 that the man was guilty of rape.
	•	lieve the man was guilty of rape, please select one of the following disciplinary sanctions that best fits the offense. Please read all the descriptions before deciding.
(2)	I believe t	he man should receive (Place a check mark by your one choice):
		No Action - No formal sanction or action is undertaken
		Warning or Reprimand - An official written statement that expresses disapproval of the conduct.
		Censure - A sanction that indicates that further violations during the specified time period for the censure will result in more severe disciplinary action, including probation, suspension, or expulsion.
		Probation - A sanction that indicates that further violations during the specified time period for the probation will result in more severe disciplinary action, including suspension or expulsion.
		Suspension - A sanction that ends a student's enrollment and removes him/her from campus for a specified period of time.
		Expulsion or Dismissal - A sanction that ends a student's enrollment and removes him/her from campus for an indefinite (often permanent) period of time.

## **SECTION D**

Directions: The following is a list of 20 statements. Please respond to each of the statements by circling the response that most identifies your feelings about the item. Respond to each statement by using the following scale:

- 1 = Strongly Agree (SA)
- 2 = Agree(A)
- 3 = No Opinion (N)
- 4 = Disagree (D)
- 5 = Strongly Disagree (SD)

	SA	A	N	D	SD
It is all right for women to work but men will always be the basic breadwinners.	1	2	3	4	5
2. A woman should not expect to go to the same places or have the same freedom of action as a man.	1	2	3	4	5
<ol> <li>Realistically speaking, most progress so far has been made by men and we can expect it to continue that way.</li> </ol>	1	2	3	4	5
<ol> <li>A woman should be expected to change her name when she marries.</li> </ol>	1	2	3	4	5
<ol><li>Women who join the Women's Movement are typically frustrated and unattractive people who feel they lose out by the current rules of society.</li></ol>	1	2	3	4	5
<ol> <li>As head of the household, the father should have final authority over his children.</li> </ol>	1	2	3	4	5
<ol> <li>A woman who refuses to give up her job to move with her husband would be to blame if the marriage broke up.</li> </ol>	1	2	3	4	5
8. Profanity sounds worse generally coming from a woman.	1	2	3	4	5
A woman who refuses to bear children has failed in her duty to her husband.	1	2	3	4	5
10. Women are basically more unpredictable than men.	1	2	3	4	5
11. The "clinging vine" wife is justified provided she clings sweetly enough to please her husband.	1	2	3	4	5
12. One should never trust a woman's account of another woman.	1	2	3	4	5
13. A working woman who sends her six month old baby to a daycare center is a bad mother.	1	2	3	4	5
14. Women should not be permitted to hold political offices that involve great responsibility.	1	2	3	4	5
15. It is desirable that women be appointed to police forces with the same duties as men.	1	2	3	4	5
16. Women have the right to compete with men in every sphere of activity.	1	2	3	4	5

Section D, Continued	SA	A	N	D	SD
<ol> <li>A woman to be truly womanly should gracefully accept chivalrous attentions from men.</li> </ol>	1	2	3	4	5
18. The unmarried mother is morally a greater failure than the unmarried father.	1	2	3	4	5
19. It is absurd to regard obedience as a wifely virtue.	1	2	3	4	5
<ol><li>Whether or not they realize it, most women are exploited by men.</li></ol>	1	2	3	4	5

Dire the follo

1.
2.
3.
4.
5.
6.
7.
8.

ŧ

1

## **SECTION E**

Directions: The following is a list of 32 statements. Please respond to each of the statements by circling the response that most identifies your feelings about the item. Respond to each statement by using the following scale (Please note that this scale differs from that used in Section D):

- 1 = Strongly Agree (SA)
- 2 = Agree(A)
- 3 = Slightly Agree (SLA)
- 4 = Slightly Disagree (SLD)
- 5 = Disagree (D)
- 6 = Strongly Disagree (SD)

		SA	Α	SLA	SLD	D	SD
1. A wor	nan can be raped against her will.	1	2	3	4	5	6
	eason most rapists commit rape is for the of physical violence.	1	2	3	4	5	6
3. Rapist	s are "normal" men.	1	2	3	4	5	6
4. In for crime.	cible rape, the victim never causes the	1	2	3	4	5	6
5. All ra	pists are mentally sick.	1	2	3	4	5	6
6. A cha	rge of rape two days after the act has red is probably not rape.	1	2	3	4	5	6
•	man should be responsible for preventing vn rape.	1	2	3	4	5	6
8. A mar given	n who has committed rape should be at least 30 years in prison.	1	2	3	4	5	6
9. Wome	on are trained by society to be rape	1	2	3	4	5	6
10. A rap	ed woman is a less desirable woman.	1	2	3	4	5	6
	roman is going to be raped, she might as relax and enjoy it.	1	2	3	4	5	6
1							

Section E, Continued	SA	Α	SLA	SLD	D	SD
<ol> <li>Rape provides the opportunity for many rapists to show their manhood.</li> </ol>	1	2	3	4	5	6
13. Most women secretly desire to be raped.	1	2	3	4	5	6
14. It would do some women some good to get raped.	1	2	3	4	5	6
15. Women provoke rape by their appearance or behavior.	1	2	3	4	5	6
16. "Nice" women do not get raped.	1	2	3	4	5	6
17. Most charges of rape are unfounded.	1	2	3	4	5	6
18. In order to protect the male, it should be difficult to prove that a rape has occurred.	1	2	3	4	5	6
19. Rape is the expression of an uncontrollable desire for sex.	1	2	3	4	5	6
20. Rape is the worst crime that can be committed.	1	2	3	4	5	6
21. Rape is a sex crime.	1	2	3	4	5	6
22. All rape is a male exercise in power over women.	1	2	3	4	5	6
23. During a rape, a woman should do everything she can do to resist.	1	2	3	4	5	6
24. Rapists are sexually frustrated individuals.	1	2	3	4	5	6
25. In most cases when a woman was raped, she was asking for it.	1	2	3	4	5	6
26. The reason most rapists commit rape is for sex.	1	2	3	4	5	6
27. Rape of a woman by a man she knows can be defined as a "woman who changed her mind afterward".	1	2	3	4	5	6
28. A convicted rapist should be castrated.	1	2	3	4	5	6
29. A woman should not feel guilty following a rape.	1	2	3	4	5	6
30. The degree of a woman's resistance should be the major factor in determining if a rape has occurred.	1	2	3	4	5	6
31. A raped woman is a responsible victim, not an innocent one.	1	2	3	4	5	6
32. Rape serves as a way to put or keep women in their "place".	1	2	3	4	5	6

## **SECTION F**

Directions: Please complete the following demographic questions by placing a checkmark by the most appropriate response. This information will be used for research purposes only and individual responses will not be reported. All responses are strictly confidential. 1. What is your gender? Female Male 2. What is your current age? \_\_\_\_ 3. What is your racial or ethnic origin? Black/African American White/Caucasian Hispanic/Chicano/Latino Indian/Native People Other (please specify)\_\_\_\_\_ 4. Do you practice a religious faith? Yes No. If yes: How religiously devout or devoted do you consider yourself? ☐ Extremely devout ☐ Strongly devout ☐ Somewhat devout ☐ Devout ☐ Not What spiritual faith do you consider as your religion? Roman Catholic ☐ Hindu **Protestant** Other (please specify) **Buddhist Jewish** 5. What is your current academic class standing? Freshman Sophomore ☐ Graduate Student Other (please specify) 6. What is your current academic major? 7. What is your current academic Grade Point Average? ☐ Yes 8. Are you a member of a fraternity or sorority?

9. How many months have you served as a member of a campus judicial (discipline) board?					
10. How many hearings have you participated in as a judical board member?					
11. Have you ever received any formal training as a judicial board member about rape and/or hearing rape cases?					
☐ Yes ☐ No					
If yes: Please check from the following list	st any topics that were presented as part of the training:				
State Laws	Rape Evidence Requirements				
College Policies	☐ Victim's Sexual History/Reputation Standards				
Judicial Hearing Procedures	Rape Stereotypes and Myths				
☐ Victim Resistance Requirements	Influence of Drugs/Alcohol				
Consent/Nonconsent Issues	Acquaintance and/or Stranger Rape Dynamics				
Use of Force Standards	Testimony/Witnesses Corroboration				
Gender-Role Stereotypes	Other (please specify)				
12. Have you ever participated as a judical boar	rd member in a rape case hearing?				
Yes No					
13. Have you or anyone you know ever been ra	uped? Yes No				
If you answered yes in Question 13, please relationship with the victim.	se check from the following list any that explain your				
Relative	Lover				
Acquaintance	☐ I have been raped				
Spouse	Other (please specify)				
Friend					
Please place the completed survey in the attached pre-addressed, stamped envelope. The survey may be returned to the person that administered the survey to you or you may send it directly to:					
Roger Serr					

Roger Serr
Department of Residence Life
G-64 Wilson Hall
Michigan State University
East Lansing, MI 48825-1208

The survey should be returned by April 1, 1993.

Thank you!

### APPENDIX C

Survey Instructions

March 16, 1993

To: Survey Administrators

From: Roger L. Serr

Subject: Survey Instructions

Thank you for your assistance collecting data concerning the influence of gender-role attitudes and rape stereotypes on the decision-making process of college and university judicial boards. Your willingness to help with this research project is greatly appreciated!

As you may recall, this research project involves administering surveys to the student members of the primary judicial boards at each participating institution. Primary judicial boards are defined as those empowered by an institution to levy or recommend disciplinary sanctions, including suspension, after finding a student guilty of violating the institution's conduct rules. It is very important that student members, and not faculty or staff, are the only people receiving the surveys.

The students have the right to decline participation in this study. However, it is very important for data analysis to receive a substantial return of completed surveys. For this reason, any endorsement on your part that encourages people to complete and return the surveys is of great value. In fact, I urge you to have the students either attend a meeting or come to your office to complete the surveys.

The following instructions should assist you in the administration of the surveys. It is important that you follow them exactly as outlined. If you have any questions, please call me at (517) 336-2492.

#### Materials:

You should have the following in your packet of materials:

- (a) One copy of a form called the Distribution Record.
- (b) One copy of a white survey booklet and 1 copy of a light gray survey booklet for your information only. Your copies should have "Administrator's Copy" written across the top. The survey booklets are exactly the same except for Section B, which features different dating scenarios.
- (c) Survey booklets in an amount equal to the number of student members serving on the judicial board. Approximately 1/2 of the booklets are white and approximately 1/2 are light gray. Each booklet has a distribution number written in the upper right corner of Page 1. Each booklet consists of 8 pages.
- (d) Pre-addressed and stamped return envelopes in an amount equal to the number of survey booklets you received. An envelope for returning the Distribution Record is also enclosed.

#### Precautionary Instructions:

It is very important to heed these instructions.

- (a) Please do not explain to subjects, prior to completing the surveys, the intent of the research beyond what is written in the cover letter on Page 1 of each survey booklet. It is especially important not to inform subjects in advance that the research deals with gender-role and rape issues.
- (b) Instruct subjects that after completing the survey, they should not discuss the survey with others scheduled to complete the survey at a later time.
- (c) If you schedule a group meeting to administer the surveys, please ensure that no conversation about the surveys takes place until all the surveys are completed and sealed in the return envelopes.

#### **Administrative Instructions:**

Please follow these instructions exactly as stated.

- (a) The number in the upper right corner of Page 1 represents the order that the surveys are to be distributed to the subjects. In other words, the first person to take the survey receives a booklet with the number 1 written in the upper right corner, the second person receives a booklet with the number 2, the third person receives booklet number 3, and so on.
  - Please explain to the subjects that the number is for distribution order only, and is not for identification purposes.
- (b) For confidentiality purposes, please have subjects seal their completed surveys inside the pre-addressed, stamped envelope before returning the surveys to you.
- (c) Please mail the envelopes with the surveys to me as you receive them. All completed surveys should be returned by April 1, 1993.
- (d) On Pages 2 and 8 of each of the survey booklets, the subjects are instructed to return completed surveys to you by April 1, 1993. If you wish, please feel free to change this date in order to establish an earlier deadline for return of the completed surveys to you.
- (e) Subjects are also informed on Pages 2 and 8 that they may send the completed surveys directly to me by April 1, 1993. This is provided as an option to help subjects feel comfortable about the confidentiality of their responses. Please don't discourage anyone from exercising this option, but any monitoring you can provide to encourage completion and return of the instruments is appreciated.
- (f) Please ensure that each of the appropriate student judicial board members at your institution are contacted and encouraged to participate in this study.
- (g) After administering all the surveys, please complete the DISTRIBUTION RECORD and return it to me in the pre-addressed, stamped envelope. IT IS VERY IMPORTANT THAT YOU RETURN THIS FORM.
- (h) If you need additional survey booklets, please do not hesitate to contact me at the aforementioned phone number.

Thanks!

# **DISTRIBUTION RECORD**

IMPORTANT: Please return this form to Roger L. Serr after administering all the surveys. A pre-addressed, stamped envelope is enclosed for your convenience.

Please	e answer the following questions:	
(a)	What is the name of your institution?	
<b>(b)</b>	How many survey booklets did you distribute to female judicial board members?	
(c)	How many survey booklets did you distribute to male judicial board members?	
(d)	What is the total number of females serving on the judicial board that you surveyed?	
(e)	What is the total number of males serving on the judicial board that you surveyed?	

Thanks!

## APPENDIX D

Varimax Rotated Factor Structure of ATR Questionnaire

Table 26. Varimax Rotated Factor Structure of ATR Questionnaire

		ATR Factor	ATR Factor Loadings*				
		* # 7	2**	# # #	***	ιυ  *	* * 9
ATR	ATR Items						
1	A woman can be raped against her will	.00961	06016	10132	30435	.17333	.32060
5	The reason most rapists commit rape is for the thrill of physical violence	04396	.08245	.24308	.14249	56954	.10553
æ.	Rapists are "normal" men	.21542	25251	.24043	57311	.07702	.08048
4.	In forcible rape, the victim never causes the crime	38281	00035	.18231	.02831	.08556	.51021
	All rapists are mentally sick	.03498	.20059	.32702	.58587	.00311	01328
•	A charge of rape two days after the act has occurred is probably not rape	.59417	.00468	.22737	.08496	00032	28773
7.	A woman should be responsible for preventing her own rape	.45814	06305	05328	23307	05262	03391
œ́	A man who has committed rape should be given at least 30 years in prison	16518	05607	.22301	12119	.61087	02393
6	Woman are trained by society to be rape victims	.05078	40300	.07839	.10651	.11769	.41525
10	10. A raped woman is a less desirable women	.10064	.04812	31946	.62667	11692	03689
11	<ol> <li>If a woman is going to be raped, she might as well relax and enjoy it</li> </ol>	. 66866	.03552	10354	.19233	.07754	.18040
12.	. Rape provides the opportunity for many rapists to show their manhood	.26594	.15794	19277	18017	03611	.62387
13.	. Most women secretly desire to be raped	.64319	.17310	03098	01622	.36118	.21042

•			
,			
•			
	•		
	:		
			•
	<i>y</i>		

٠.
(t)
Son
26 (
<b>616</b>
"Ta

14.	It would do some women some good to get raped	.69777	.16273	.03151	.11286	.07231	.26961
15.	Women provoke rape by their appearance or behavior	.67493	.17377	27781	.03529	14188	05486
16.	"Nice" women do not get raped	.43918	.06569	10562	.48504	34575	13156
17.	17. Most charges of rape are unfounded	.16332	.33353	32161	16219	.29421	02767
18.	In order to protect the male, it should be difficult to prove that a rape has occurred	.24702	01322	72413	03952	23089	12091
19.	Rape is the expression of an uncontrollable desire for sex	.04309	.75171	05507	.02149	00069	01539
20.	Rape is the worst crime that can be committed	22280	.13768	.68327	09902	02243	10412
21.	Rape is a sex crime	.01473	.63404	06749	.25049	.25775	11279
22.	22. All rape is a male exercise in power over women	.00637	11417	. 48001	12580	.02088	.04410
23.	. During a rape, a woman should do everything she can do to resist	.11318	.53400	.20285	49777	11771	.01941
24.	. Rapists are sexually frustrated individuals	.19719	.61361	.15657	.22494	17336	14608
25.	. In most cases when a woman was raped, she was asking for it	.66109	.04204	29371	.14569	12429	03194
26.	. The reason most rapists commit rape is for sex	.17369	.71040	03223	.16767	.00372	.01232
27	27. Rape of a woman by a man she knows can be defined as a "woman who changed her mind afterward"	.63235	.01380	33861	12368	29516	22296

		"Table 2	"Table 26 (Con't)."				
28	28. A convicted rapist should be castrated	.11207	.14355	.24830	.03416	.65601	13662
29	29. A woman should not feel guilty following a rape	12115	14894	.04665	03685	10355	. 45969
30	30. The degree of a woman's resistance should be the major factor in determining if a rape has occurred	. 69885	.11015	.00783	14774	.15534	.01371
31	31. A raped woman is a responsible victim, not an innocent one	.49072	.21712	12973	16418	34105	20879
32	32. Rape serves as a way to put or keep women in their "place"	.11652	07138	.07775	02745	23411	. 59996
	Alpha	.807	. 693	. 529	. 500	.413	.485
	Percent of Variance	31.5	20.9	8.5	9.8	9.5	15.3
	# Tages 1: 1. Colons						

Underlined coefficients signify item assignments to specific factors Rape precipitation and responsibility
Rape motivated by sexual desire
Rape as a crime
Normality of rape
Rape as a violent act
Cultural influences on rape Factor 1: Factor 2: Factor 4: Factor 5:

### **BIBLIOGRAPHY**

#### **BIBLIOGRAPHY**

- Abbey, A. (1982). Sex differences in attributions for friendly behavior: Do males misperceive females' friendliness? <u>Journal of Personality and Social Psychology</u>, 42, 830-838.
- Abbey, A. (1987). Misperceptions of friendly behavior as sexual interest: A survey of naturally occurring incidents. <u>Psychology of Women Quarterly</u>, <u>11</u>, 173-194.
- Abbey, A., and Melby, C. (1986). The effects of nonverbal cues on gender differences in perceptions of sexual intent. <u>Sex Roles</u>, <u>15</u>, 283-298.
- Acock, A. C., and Ireland, N. K. (1983). Attribution of blame in rape cases: The impact of norm violation, gender, and sex-role attitude.

  <u>Sex Roles: A Journal of Research</u>, 9, 179-193.
- Adams, A., and Abarbanel, G. (1988). <u>Sexual assault on campus: What colleges can do</u>. Santa Monica, CA: Santa Monica Hospital Medical Center.
- Ageton, S. S. (1983). <u>Sexual assault among adolescents</u>. Lexington, MA: Lexington Books.
- Aizenman, M., and Kelley, G. (1988). The incidence of violence and acquaintance rape in dating relationships among college men and women.

  <u>Journal of College Student Development</u>, 29, 305-311.
- Allport, F. H. (1924). Social psychology. Boston, MA: Houghton Mifflin.
- Amick, A. E., and Calhoun, K. S. (1987). Resistance to sexual aggression: Personality, attitudinal, and situational factors.

  <u>Archives of Sexual Behavior: An Interdisciplinary Research Journal</u>, 16, 153-163.
- Amir, M. (1971). <u>Pattern in forcible rape</u>. Chicago, IL: University of Chicago.
- Amir, M. (1975). Forcible rape. In L.G. Schultz (Ed.), Rape victimology (pp. 43-58). Springfield, IL: Charles C. Thomas.
- Associated Press. (1991, November 9). Ex-CMU student sues in rape case. Lansing State Journal, p. 3B.
- Association of American Colleges. (1990). Perceptions of rape. On Campus With Women, 20, 7.
- Astin, A. W. (1993). What matters in college. San Francisco, CA: Jossey-Bass.
- Babbie, E. (1990). Survey research methods. Belmont, CA: Wadsworth.
- Baier, J. L., Rosenzweig, M. G., and Whipple, E. G. (1991). Patterns of sexual behavior, coercion, and victimization of university students.

  <u>Journal of College Student Development</u>, 32, 310-322.

- Bailey, J. H. (1986). Every man is a potential rapist. <u>Campus Law Enforcement Journal</u>, <u>16</u>, 34-35.
- Barber, R. (1974). Judge and jury attitudes to rape. <u>Austrailia and New Zealand Journal of Criminology</u>, 7, 157-172.
- Barnett, N. J., and Feild, H.S. (1977). Sex differences in university students' attitudes toward rape. <u>Journal of College Student Personnel</u>, 18, 93-96.
- Barnett, N. J., and Feild, H. S. (1978). Character of the defendant and length of sentence in rape and burglary crimes. <u>The Journal of Social Psychology</u>, 104, 271-277.
- Bart, P. B. (1975). Rape doesn't end with a kiss. <u>Viva</u>, <u>11</u>, 39-42, 98-101.
- Bash, A. (1993, August 9). Orientations raise serious issues, from crime to drinking. <u>USA Today</u>, p. 2D.
- Batson, S. W. (1986). Minimizing liability for the college administrator: Female student protection. In T. N. Jones and D. P. Semler (Eds.), <u>School law update 1986</u> (pp. 120-128). Topeka, KS: National Organization On Legal Problems Of Education.
- Bartz, A. E. (1976). <u>Basic statistical concepts in education and the behavioral sciences</u>. Minneapolis, MN: Burgess.
- Bechhofer, L., and Parrot, A. (1991). What is acquaintance rape? In A. Parrot and L. Bechhofer (Eds.), <u>Acquaintance rape: The hidden crime</u> (pp. 9-25). New York, NY: John Wiley and Sons.
- Bem, S. L. (1974). The measurement of psychological androgyny. <u>Journal of Consulting and Clinical Psychology</u>, 42, 155-162.
- BenDor, J. (1976). Justice after rape: Legal reform in Michigan. In M. J. Walker and S. L. Brodsky (Eds.), <u>Sexual assault: The victim and the rapist</u> (pp. 149-160). Lexington, MA: Lexington Books.
- Berger, R. J., Searles, P., Salem, R. G., and Pierce, B.A. (1986). Sexual assault in a college community. <u>Sociological Focus</u>, <u>19</u>, 1-26.
- Bessmer, S. (1984). The laws of rape. New York, NY: Praeger.
- Bhirdo, K. W. (1989). The liability and responsibility of institutions of higher education for the on-campus victimization of students. The Journal of College and University Law, 16, 119-135.
- Blaskey, M. S. (1988). University students' right to retain counsel for disciplinary proceedings. <u>California Western Law Review</u>, <u>24</u>, 65-82.
- Bogal-Allbritten, R. B., and Allbritten, W. L. (1985). The hidden victims: Courtship violence among college students. <u>Journal of College Student Personnel</u>, <u>26</u>, 201-209
- Bogal-Allbritten, R., and Allbritten, W. (1991). Courtship violence on campus: A nationwide survey of student affairs professionals. <u>NASPA</u> <u>Journal</u>, 28, 312-318.
- Bohmer, C. (1974). Judicial attitudes toward rape victims. <u>Judicature</u>, <u>57</u>, 303-307.

- Bohmer, C. (1991). Acquaintance rape and the law. In A. Parrot and L. Bechhofer (Eds.), <u>Acquaintance rape: The hidden crime</u> (pp. 317-333). New York, NY: John Wiley and Sons.
- Bohmer, C., and Parrot, A. (1993). <u>Sexual assault on campus: The problem and the solution</u>. New York, NY: Lexington.
- Borg, W. R., and Gall, M. D. (1983). <u>Educational Research: An Introduction</u>. New York, NY: Longman.
- Borgida, E., and White, P. (1978). Social perception of rape victims: The impact of legal reform. <u>Law and Human Behavior</u>, 2, 339-351.
- Bostwick, T. D. (1990). <u>Effects of gender role identity and specific dating behaviors on perceptions of date rape</u>. Unpublished doctoral dissertation, Purdue University, W. Lafayette, IN.
- Bownes, I. T., O'Gorman, E. C., and Sayers, A. (1991). Rape A comparison of stranger and acquaintance assaults. <u>Medicine, Science, and the Law</u>, 31, 102-109.
- Boyer, E. L. (1987). <u>College: The undergraduate experience in America</u>. New York, NY: Harper and Row.
- Bradley, G. W. (1978). Self-serving biases in the attribution process: A reexamination of the fact or fiction question. <u>Journal of Personality and Social Psychology</u>, 36, 56-71.
- Bray, J. H., and Maxwell, S. E. (1985). <u>Multivariate analysis of variance</u>. Newbury Park, CA: Sage.
- Breed, W., and Ktsanes, T. (1961). Pluralistic ignorance in the process of opinion formation. <u>Public Opinion Quarterly</u>, <u>25</u>, 382-392.
- Bridges, J. S., and McGrail, C. A. (1989). Attributions of responsibility for date and stranger rape. <u>Sex Roles: A Journal of Research</u>, 21, 273-286.
- Brodyaga, L., Gates, M., Singer, S., Tucker, M., and White, R. (1975).

  Rape and its victims: A report for citizens, health facilities, and

  criminal justice agencies. Washington, DC: U. S. Government Printing

  Office.
- Briddell, D. W., Rimm, D. C., Caddy, G. R., Krawitz, G., Sholis, D., and Wunderlin, R. J. (1978). Effects of alcohol and cognitive set on sexual arousal to deviant stimuli. <u>Journal of Abnormal Psychology</u>, <u>87</u>, 418-430.
- Briere, J., and Malamuth, N. M. (1983). Self-reported likelihood of sexually aggressive behavior: Attitudinal versus sexual explanations. <u>Journal of Research in Personality</u>, <u>17</u>, 315-323.
- Brooks, W. N., and Doob, A. N. (1975). Justice and the jury. <u>Journal of Social Issues</u>, <u>31</u>, 171-182.
- Broverman, I. K., Vogel, S. R., Broverman, D. M., Clarkson, F. E., and Rosenkrantz, P. S. (1972). Sex-role stereotypes: A current appraisal. <u>Journal of Social Issues</u>, 28, 59-78.
- Brownmiller, S. (1975). Against our will. New York, NY: Simon and Schuster.

.

·
·
·
·
·
·
·

•

- Buhrke, R. A., and Lustgraaf, M. (1988). Date rape awareness program: A model for educating and consciousness raising. <u>Journal of College Student Development</u>, 29, 478-479.
- Bunting, A. B., and Reeves, J. B. (1983). Perceived male sex orientation and beliefs about rape. <u>Deviant Behavior</u>, 4, 281-295.
- Bureau of Justice Statistics. (1984). <u>Criminal victimization in the United States, 1982</u> (Publication No. NCJ-92820). Washington, DC: Bureau of Justice Statistics.
- Bureau of Justice Statistics. (1985a). Criminal victimization in the United States, 1983 (Publication No. NCJ-96459). Washington, DC: Bureau of Justice Statistics.
- Bureau of Justice Statistics. (1985b). The crime of rape (Publication No. NCJ-96777). Washington, DC: Bureau of Justice Statistics.
- Bureau of Justice Statistics. (1986). Teenage victims: A national crime survey report (Publication No. NCJ-103138). Washington, DC: Bureau of Justice Statistics.
- Bureau of Justice Statistics. (1987). <u>Criminal victimization in the United States</u>, 1985 (Publication No. NCJ-104273). Washington, DC: Bureau of Justice Statistics.
- Bureau of Justice Statistics. (1988). <u>Criminal victimization in the United States, 1986</u> (Publication No. NCJ-111456). Washington, DC: Bureau of Justice Statistics.
- Bureau of Justice Statistics. (1989). <u>Criminal victimization in the United States</u>, 1987 (Publication No. NCJ-115524). Washington, DC: Bureau of Justice Statistics.
- Bureau of Justice Statistics. (1990). <u>Criminal victimization in the United States</u>, 1988 (Publication No. NCJ-122024). Washington, DC: Bureau of Justice Statistics.
- Bureau of Justice Statistics. (1991a). <u>Criminal victimization in the United States: 1973-1988 trends</u> (Publication No. NCJ-129392). Washington, DC: Bureau of Justice Statistics.
- Bureau of Justice Statistics. (1991b). <u>Violent crime in the United States</u> (Publication No. NCJ-127855). Washington, DC: Bureau of Justice Statistics.
- Burgess, A.W., and Holmstrom, L.L. (1980). Rape typology and the coping behavior of rape victims. In S.L. McCombie (Ed.), <u>The rape crisis intervention handbook: A guide for victim care</u> (pp. 27-40). New York, NY: Plenum.
- Burkhart, B. R., and Stanton, A. L. (1988). Sexual aggression in acquaintance relationships. In G. W. Russell (Ed.), <u>Violence in intimate relationships</u> (pp. 43-65). New York, NY: PMA.
- Burt, M. R. (1980). Cultural myths and supports for rape. <u>Journal of Personality and Social Psychology</u>, 38, 217-230.

.

.

•

: .

- Burt, M. R. (1991). Rape myths and acquaintance rape. In A. Parrot and L. Bechhofer (Eds.), <u>Acquaintance rape: The hidden crime</u> (pp. 26-40). New York, NY: John Wiley and Sons.
- Burt, M. R., and Albin, R. S. (1981). Rape myths, rape definitions, and probability of conviction. <u>Journal of Applied Social Psychology</u>, <u>11</u>, 212-230.
- Cage, M. C. (1992, September 30). Forty-two percent of college students engage in binge drinking, survey shows. <u>The Chronicle of Higher</u> <u>Education</u>, p. A30.
- Calhoun, L. G., Selby, J. W., and Warring, L. J. (1976). Social perception of the victim's causal role in rape: An exploratory examination of four factors. <u>Human Relations</u>, 29, 517-526.
- Campus Violence Prevention Center. (1988). National campus violence survey 1986, 1987 and 1988 data tables. Towson, MD: Towson State University, Campus Violence Prevention Center.
- Campus Violence Prevention Center. (1990). 1990 national campus violence survey: General report. Towson, MD: Towson State University, Campus Violence Prevention Center.
- Cann, A., Calhoun, L. G., and Selby, J. W. (1979). Attributing responsibility to the victim of rape: Influence of information regarding past sexual experience. <u>Human Relations</u>, 32, 57-67.
- Caringella-MacDonald, S. (1985). Sexual assault prosecution: An examination of model rape legislation in Michigan. In C. SchWeber and C. Feinman (Eds.), <u>Criminal justice politics and women: The aftermath of legally mandated change</u> (pp. 65-82). New York, NY: Haworth.
- Caringella-MacDonald, S. (1988). Marxist and feminist interpretations on the aftermath of rape reforms. Contempory Crisis, 12, 125-144.
- Cattell, R. B. (1952). <u>Factor analysis: An introduction and manual for the psychologist and social scientist</u>. New York, NY: Harper & Brothers.
- Chappell, D. (1984). The impact of rape legislation reform: Some comparative trends. <u>International Journal of Women's Studies</u>, 7, 70-80.
- Chappell, D., and Singer, S. (1977). Rape in New York City: A study of materials in the police files and its meaning. In D. Chappell, R. Geis, and G. Geis (Eds.), <u>Forcible rape</u> (pp. 245-271). New York, NY: Columbia University.
- Check, J. V. P., and Malamuth, N. M. (1983). Sex-role stereotyping and reactions to depictions of stranger versus acquaintance rape. <u>Journal of Personality and Social Psychology</u>, <u>45</u>, 344-356.
- Check, J. V. P., and Malamuth, N. (1985). An empirical assessment of some feminist hypotheses about rape. <u>International Journal of Women's Studies</u>, 8, 414-423.
- Child, D. (1990). The essentials of factor analysis. London, England: Cassell.
- Clark, L., and Lewis, D. (1977). Rape: The price of coercive sexuality. Toronto, Ontario, Canada: Women's Press.

É

- Cobb, K. A., and Schauer, N. R. (1974). Michigan's criminal sexual assault law. <u>University of Michigan Journal of Law Reform</u>, 8, 217-236.
- Cockey, M., Sherrill, J. M., and Cave, R. B. (1989). Towson State University's research on campus violence. In J. M. Sherrill and D. G. Siegel (Eds.), Responding to violence on campus (pp. 17-27). San Francisco, CA: Jossey-Bass.
- Coller, S. A., and Resick, P. A. (1987). Women's attributions of responsibility for date rape: The influence of empathy and sex-role stereotyping. Violence and Victims, 2, 115-125.
- Collison, M. N. (1991, May 15). Increase in reports of sexual assaults strains campus disciplinary systems. The Chronicle of Higher Education, pp. A29-A30.
- Copenhaver, S., and Grauerholz, E. (1991). Sexual victimization among sorority women: Exploring the link between sexual violence and institutional practices. Sex Roles: A Journal of Research, 24, 31-41.
- Costin, F. (1985). Beliefs about rape and women's social roles. <u>Archives of Sexual Behavior</u>, <u>14</u>, 319-325.
- Crowe, L. C., and George, W. H. (1989). Alcohol and human sexuality: Review and integration. <u>Psychological Bulletin</u>, <u>105</u>, 374-386.
- Curtis, L. A. (1976). Present and future measures of victimization in forcible rape. In M. J. Walker and S. L. Brodsky (Eds.), <u>Sexual assault: The victim and the rapist</u> (pp. 61-68). Lexington, MA: Lexington Books.
- Dannells, M. (1991). Changes in student misconduct and institutional response over 10 years. <u>Journal of College Student Development</u>, <u>32</u>, 166-170.
- Dean, C. W., and deBruyn-Kops, M. (1982). The crime and the consequences of rape. Springfield, IL: Charles C. Thomas.
- Deaux, K. (1976). Sex: A perspective on the attribution process. In J. H. Harvey, W. J. Ickes, and R. F. Kidd (Eds.), New Directions In Attribution Research: Volume 1 (pp. 335-352). Hillsdale, NJ: Lawrence Erlbaum Associates.
- Dermer, M., and Thiel, D. L. (1975). When beauty may fail. <u>Journal of Personality and Social Psychology</u>, <u>31</u>, 1168-1176.
- Diamond, S., and Zeisel, H. A. (1974). A courtroom experiment on juror selection and decision-making. <u>Personality and Social Psychology</u> <u>Bulletin</u>, 1, 276-277.
- DiPerna, P. (1984). Juries on trail. New York, NY: Dembner Books.
- Dodd, V. J. (1985). The non-contractual nature of the student-university contractual relationship. <u>Kansas Law Review</u>, <u>33</u>, 701-731.
- Dull, R. T., and Giacopassi, D. J. (1987). Demographic correlates of sexual and dating attitudes: A study of date rape. <u>Criminal Justice and Behavior</u>, 14, 175-193.
- Dworkin, R. B. (1966). The resistance standard in rape legislation. Stanford Law Review, 18, 680-689.

•

.

; ;

r. •.

· .

•

- Bfran, M. G. (1974). The effect of physical appearance on the judgement of guilt, interpersonal attraction, and severity of recommended punishment in a simulated jury task. <u>Journal of Research in Personality</u>, 8, 45-54.
- Ellis, L. (1989). Theories of rape: Inquires into the causes of sexual aggression. New York, NY: Hemisphere.
- Eskridge, C. W. (1989). The extent of rape among college students: An exploratory analysis. Campus Law Enforcement Journal, 19, 32-37.
- Estrich, S. (1987). Real rape. Cambridge, MA: Harvard University.
- Federal Bureau of Investigation. (1976). Crime in the United States:

  <u>Uniform crime reports, 1976</u>. Washington, DC: U. S. Government Printing
  Office.
- Federal Bureau of Investigation. (1980). Crime in the United States:
  Uniform crime reports, 1980. Washington, DC: U. S. Government Printing
  Office.
- Federal Bureau of Investigation. (1982). <u>Crime in the United States:</u>
  <u>Uniform crime reports, 1982</u>. Washington, DC: U. S. Government Printing Office.
- Federal Bureau of Investigation. (1984). Crime in the United States:
  Uniform crime reports, 1984. Washington, DC: U. S. Government Printing
  Office.
- Federal Bureau of Investigation. (1985). Crime in the United States:
  Uniform crime reports, 1985. Washington, DC: U. S. Government Printing
  Office.
- Federal Bureau of Investigation. (1989). <u>Crime in the United States:</u>
  <u>Uniform crime reports, 1989</u>. Washington, DC: U. S. Government Printing Office.
- Feild, H. S. (1978a). Attitudes toward rape: A comparative analysis of police, rapists, crisis counselors, and citizens. <u>Journal of Personality and Social Psychology</u>, <u>36</u>, 156-179.
- Feild, H. S. (1978b). Juror background characteristics and attitudes toward rape. <u>Law and Human Behavior</u>, 2, 73-93.
- Feild, H. S., and Bienen, L. B. (1980). <u>Jurors and rape: A study in psychology and law</u>. Lexington, MA: Lexington Books.
- Feldman-Summers, S., and Lindner, K. (1976). Perceptions of victims and defendants in criminal assault cases. <u>Criminal Justice and Behavior</u>, 3, 135-149.
- Fernberger, S. W. (1948). Persistence of stereotypes concerning sex differences. The Journal of Abnormal and Social Psychology, 43, 97-101.
- Fishbein, E. A. (1985). New strings on the ivory tower: The growth of accountability in colleges and universities. <u>Journal of College and University Law</u>, 12, 381-398.
- Fisher, G. J. (1986). College student attitudes toward forcible date rape. <u>Archives of Sexual Behavior</u>, 15, 457-465.
- Frank, J. (1949). Courts on trail. Princeton, NJ: Princeton University Press.

- Frazier, P., and Borgida, E. (1988). Juror common understanding and the admissibility of rape trauma syndrome evidence in court. <u>Law and Human Behavior</u>, <u>12</u>, 101-122.
- Freeman, M. A. (1991). Lawmakers consider education bills. NASPA Forum, 12, 1, 4.
- Freeman, P. (1990, December 17). Silent no more. People, pp. 94-104.
- Fulero, S. M., and Delara, C. (1976). Rape victims and attributed responsibility: A defensive attribution approach. <u>Victimology: An International Journal</u>, 1, 551-563.
- Galvin, J. (1985). Rape: A decade of reform. Crime and Delinquency, 31, 163-168.
- Garrett-Gooding, J., and Senter, R. (1987). Attitudes and acts of sexual aggression on a university campus. <u>Sociological Inquiry</u>, <u>57</u>, 348-371.
- Gehring, D. D. (1993). Abreast of the law. NASPA Forum, 13, 5, 9-10.
- George, W. H., Gournic, S. J., and McAfee, M. P. (1988). Perceptions of postdrinking female sexuality: Effects of gender, beverage choice, and drink payment. <u>Journal of Applied Social Psychology</u>, <u>18</u>, 1295-1317.
- Gerbasi, K. C., Zuckerman, M., and Reis, H. T. (1977). Justice needs a new blindfold: A review of mock jury research. <u>Psychological Bulletin</u>, <u>84</u>, 323-345.
- Gibbs, N. (1991, June 3). When is it rape? Time, pp. 48-54.
- Gilbert, N. (1991, June 27). The campus rape scare. Wall Street Journal, p. A14.
- Glass, G. V., and Hopkins, K. D. (1984). <u>Statistical methods in education and psychology</u>. Englewood Cliffs, NJ: Prentice-Hall.
- Goodchilds, J. D., Zellman, G. L., Johnson, P. B., and Giarrusso, R. (1988). Adolescents and their perceptions of sexual interactions. In A. W. Burgess (Ed.), <u>Sexual assault: Vol. II</u> (pp. 245-270). New York, NY: Garland.
- Greendlinger, V., and Byrne, D. (1987). Coercive sexual fantasies of college men as predictors of self-reported likelihood to rape and overt sexual aggression. The Journal of Sex Research, 23, 1-11.
- Griffin, S. (1975). Rape: The all-american crime. In L. G. Schultz (Ed.), Rape Victimology (pp. 19-39). Springfield, IL: Charles C. Thomas.
- Griffin, S. (1979). Rape: The power of consciousness. San Francisco, CA: Harper and Row.
- Griffitt, W., and Jackson, T. (1973). Simulated jury decisions: The influence of jury-defendant attitude similarity-dissimilarity. <u>Social Behavior and Personality</u>, 1, 1-7.
- Groth, A. N. (1979). Men who rape: The psychology of the offender. New York, NY: Plenum.
- Groth, A, N., and Birnbaum, H. J. (1980). The rapist: Motivations for sexual violence. In S. L. McCombie (Ed.), <u>The rape crisis intervention handbook: A quide for victim care</u> (pp. 17-26). New York, NY: Plenum.

•

. 50

.

.:

• • •

. .

.--;

- Guinther, J. (1988). The jury in america. New York, NY: Facts On File.
- Guttman, S. (1990, October). Date rape: Does anyone really know what it is? Playboy, pp. 48-51.
- Hale, M. (1971). The history of the pleas of the crown, Vol. I. London, England: Professional Books.
- Hamilton, D. L. (1979). A cognitive-attributional analysis of stereotyping. In L. Berkowitz (Ed.), <u>Advances in experimental social psychology</u> (pp.53-84). New York, NY: Adademic Press.
- Harlow, C. W. (1991). <u>Female victims of violent crime</u> (Publication No. NCJ-126826). Washington, DC: Bureau of Justice Statistics.
- Harman, H. H. (1976). Modern factor analysis. Chicago, IL: The University of Chicago Press.
- Harris, L. R. (1976). Toward a consent standard in the law of rape. The University of Chicago Law Review, 43, 613-645.
- Harrison, P. J., Downes, J., and Williams, M. D. (1991). Date and acquaintance rape: Perceptions and attitude change strategies. <u>Journal of College Student Development</u>, 32, 131-139.
- Heider, F. (1958). The psychology of interpersonal relations. New York, NY: John Wiley and Sons.
- Helms, L. B. (1987). Patterns of litigation in postsecondary education: A case law study. <u>Journal of College and University Law</u>, <u>14</u>, 99-119.
- Hendrix, K. (1991, July 9). Defining controversy: Professor raises furor by claiming date rape statistics are inflated. <u>Los Angeles Times</u>, pp. E1-E2.
- Hindelang, M. J., Gottfredson, M. R., and Garofalo, J. (1978). <u>Victims of personal crime: An empirical foundation for a theory of personal victimization</u>. Cambridge, MA: Ballinger.
- Holmberg, C. B., and MacDonald, W. L. (1990). Student perceptions of assault and date rape. <u>Campus Law Enforcement Journal</u>, <u>20</u>, 16-19.
- Holmes, M. R., and St. Lawrence, J. S. (1983). Treatment of rape-induced trauma: Proposed behavioral conceptualization and review of the literature. Clinical Psychology Review, 3, 417-433.
- Holmstrom, L. L., and Burgess, A. W. (1975). Rape: The victim goes on trail. In I. Drapkin and E. Viano (Eds.), <u>Victimology: A new focus: Volume 3</u> (pp. 31-47). Lexington, MA: D. C. Heath.
- Horsley, K. R. (1988). <u>Demographic and religious factors and variation in sex role attitudes and behaviors</u>. Unpublished master's thesis, The University of Texas, Arlington.
- Howard, J. A. (1984a). Societal influences on attribution: Blaming some victims more than others. <u>Journal of Personality and Social</u>
  <u>Psychology</u>, <u>47</u>, 494-505.
- Howard, J. A. (1984b). The normal victim: The effects of gender stereotypes on reactions to victims. <u>Social Psychology Quarterly</u>, <u>47</u>, 270-281.

2		
£		
· <b>,</b>		
÷		
٠		

- Hueston, H. R., and Burkhart, B. (1986). Acquaintance rape: Developing a comprehensive prevention program. <u>Campus Law Enforcement Journal</u>, 16, 30-33.
- Hughes, J. O., and Sandler, B. R. (1987). <u>Friends raping friends: Could it happen to you?</u> Washington, DC: Association of American Colleges, Project on the Status and Education of Women.
- Irving, C. (1991, June 2). Date-rape statistics skewed, professor says.
   Chicago Tribune, p. 17.
- Iversen, G. R., and Norpoth, H. (1987). Analysis of variance. Newbury
  Park, CA: Sage.
- Jenkins, M. J., and Dambrot, F. H. (1987). The attribution of date rape: Observer's attitudes and sexual experiences and the dating system.

  <u>Journal of Applied Social Psychology</u>, 17, 875-895.
- Johnson, A. G. (1980). On the prevalence of rape in the United States. Signs: Journal of Women in Culture and Society, 6, 136-146.
- Jones, C., and Aronson, E. (1973). Attribution of fault to a rape victim as a function of respectability of the victim. <u>Journal of Personality and Social Psychology</u>, <u>26</u>, 415-419.
- Jones, E. E., and Davis, K. E. (1965). From acts to dispositions: The attribution process in person perception. In L. Berkowitz (Ed.), Advances in experimental social psychology: Volume 2 (pp. 219-266). New York, NY: Academic Press.
- Julian, F. H. (1992). Abreast of the law. NASPA Forum, 12, 7.
- Kahn, A., Gilbert, L. A., Latta, R. M., Deutsch, C., Hagen, R., Hill, M., McGaughey, T., Ryen, A. H., and Wilson, D. W. (1977). Attribution of fault to a rape victim as a function of respectability of the victim: A failure to replicate or extend. Representative Research in Social Psychology, 8, 98-107.
- Kalette, D. (1988, October 9). Schools wrestle with rape, the no. 1 crime. <u>Lansing State Journal</u>, p. 13A.
- Kalven, Jr., H., and Zeisel, H. (1966). The American jury. Boston, MA: Little, Brown.
- Kanekar, S., and Kolsawalla, M. B. (1977). Responsibility in relation to respectability. The Journal of Social Psychology, 102, 183-188.
- Kanekar, S., and Vaz, L. (1983). Determinants of perceived likelihood of rape and victim's fault. <u>The Journal of Social Psychology</u>, <u>120</u>, 147-148.
- Kanin, E. J. (1957). Male aggression in dating-courtship relations. The American Journal of Sociology, 63, 197-204.
- Kanin, E. J. (1967). Reference groups and sex conduct norm violations. The Sociological Quarterly, 8, 495-504.
- Kanin, E. J. (1975). Selected dyadic aspects of male sex aggression. In L. G. Schultz (Ed.), <u>Rape Victimology</u> (pp. 59-76). Springfield, IL: Charles C. Thomas.
- Kanin, E. J. (1984). Date rape: Unofficial criminals and victims. Victimology: An International Journal, 9, 95-108.

•.

• •

**3** .

4. 0

. -

•

. 2

, ;

- Kanin, E. J., and Parcell, S. R. (1977). Sexual aggression: A second look at the offended female. <u>Archives of Sexual Behavior</u>, <u>6</u>, 67-76.
- Kaplin, W. A. (1985). Law on the campus 1960-1985: Years of growth and challenge. The Journal of College and University Law, 12, 269-299.
- Kaplin, W. A. (1989). The law of higher education. San Francisco, CA: Jossey-Bass.
- Kaplan, M. F., and Miller, L. E. (1987). Reducing the effects of juror bias. In L. S. Wrightsman, S. M. Kassin, and C.E. Willis (Eds.), <u>In the jury box: Controversies in the courtroom</u> (pp. 114-137). Newbury Park, CA: Sage.
- Katz, D., and Allport, F. H. (1931). <u>Students' Attitudes</u>. Syracuse, NY: Craftsman Press.
- Katz, S., and Mazur, M. A. (1979). <u>Understanding the rape victim: A synthesis of research findings</u>. New York, NY: John Wiley and Sons.
- Kelley, H. H. (1973). The process of causal attribution. <u>American Psychologist</u>, 28, 107-128.
- Kerr, N. L., and Kurtz, S. T. (1977). Effects of a victim's suffering and respectability on mock juror judgements: Further evidence on the just world theory. <u>Representative Research in Social Psychology</u>, 8, 42-56.
- Kerstetter, W. A. (1990). Criminology: Gateway to justice: Police and prosecutorial response to sexual assaults against women. The Journal of Criminal Law and Criminology, 81, 267-313.
- Kirkpatrick, C., and Kanin, E. (1957). Male sex aggression on a university campus. <u>American Sociological Review</u>, <u>22</u>, 52-58.
- Klemmack, S. H., and Klemmack, D. L. (1976). The social definition of rape. In M. J. Walker and S. L. Brodsky (Eds.), <u>Sexual Assault: The</u> <u>Victim and the Rapist</u> (pp. 135-147). Lexington, MA: Lexington Books.
- Knox, D., and Wilson, K. (1983). Dating problems of university students.
  <u>College Student Journal</u>, 17, 225-228.
- Korman, S. K., and Leslie, G. R. (1982). The relationship of feminist ideology and date expense sharing to perceptions of sexual aggression in dating. <u>Journal of Sex Research</u>, <u>18</u>, 114-129.
- Koss, M. P. (1985). The hidden rape victim: Personality, attitudinal and situational characteristics. <u>Psychology of Women Quarterly</u>, 9, 193-212.
- Koss, M. P. (1988). Hidden rape: Sexual aggression and victimization in a national sample of students in higher education. In A. W. Burgess (Ed.), <u>Sexual Assault: Volume II</u> (pp. 1-25). New York, NY: Garland.
- Koss, M. P. (1992). Rape on campus: Facts and measures. <u>Planning For Higher Education</u>, 20, 21-28.
- Koss, M. P., and Dinero, T. E. (1988). Predictors of sexual aggression among a national sample of male college students. <u>Human Sexual Aggression</u>, 528, 133-147.
- Koss, M. P., and Dinero, T. E. (1989) Discriminant analysis of risk factors for sexual victimization among a national sample of college women. <u>Journal of Consulting and Clinical Psychology</u>, <u>57</u>, 242-250.

•		
•		
•		
•		
i		
•		
•		
-		
r		
,		
•.		
•		
· :		

- Koss, M. P., and Gidycz, C.A. (1985). Sexual experiences survey: Reliability and validity. <u>Journal of Consulting and Clinical Psychology</u>, <u>53</u>, 422-423.
- Koss, M. P., Gidycz, C. A., and Wisniewski, N. (1987). The scope of rape: Incidence and prevalence of sexual aggression and victimization in a national sample of higher education students. <u>Journal of Consulting</u> and <u>Clinical Psychology</u>, <u>55</u>, 162-170.
- Koss, M. P., and Harvey, M. R. (1987). The rape victim: Clinical and community approaches to treatment. Lexington, MA: Stephen Greene.
- Koss, M. P., and Leonard, K. E. (1984). Sexually aggressive men: Empirical findings and theoretical implications. In N. M. Malamuth and E. Donnerstein (Eds.), <u>Pornography and sexual aggression</u> (pp. 213-232). New York, NY: Academic.
- Koss, M. P., Leonard, K. E., Breezley, D. A., and Oros, C. J. (1985). Nonstranger sexual aggression: A discriminant analysis of the psychological characteristics of undetected offenders. <u>Sex Roles: A Journal of Research</u>, 12, 981-992.
- Koss, M. P., and Oros, C. J. (1982). Sexual experiences survey: A research instrument investigating sexual aggression and victimization. <u>Journal of Consulting and Clinical Psychology</u>, <u>50</u>, 455-457.
- Krasner, W., Meyer, L. C., and Carroll, N. E. (1976). Victims of Rape
  (Stock No. 017-024-00683-1). Washington, DC: U. S. Government Printing
  Office.
- Krulewitz, J. E., and Nash, J. E. (1979). Effects of rape victim resistance, assault outcome, and sex of observer on attributions about rape. <u>Journal of Personality</u>, <u>47</u>, 557-574.
- Krulewitz, J. E., and Payne, E. J. (1978). Attributions about rape: Effects of rapist force, observer sex and sex role attitudes. <u>Journal of Applied Social Psychology</u>, 8, 291-305.
- LaFree, G. D. (1980). Variables affecting guilty pleas and convictions in rape cases: Toward a social theory of rape processing. <u>Social Forces: An International Journal of Social Research</u>, <u>58</u>, 833-850.
- LaFree, G. D. (1981). Official reactions to social problems: Police decisions in sexual assault cases. <u>Social Problems</u>, <u>28</u>, 582-594.
- LaFree, G. D. (1989). <u>Rape and criminal justice: The social construction of sexual assault</u>. Belmont, CA: Wadsworth.
- LaFree, G. D., Reskin, B. F., and Visher, C. A. (1985). Jurors' responses to victims' behavior and legal issues in sexual assault trails. <u>Social Problems</u>, <u>32</u>, 389-407.
- Lancaster, J. M., Cooper, D. L., and Harman, A. E. (1993). Current practices in student disciplinary administration. <u>NASPA Journal</u>, <u>30</u>, 108-119.
- Langer, E., Blank, A., and Chanowitz, B. (1978). The mindlessness of ostensibly thoughtful action: The role of placebic information in interpersonal interaction. <u>Journal of Personality and Social Psychology</u>, 36, 635-642.

; ,

> \*\* . .

- LaPlante, M. N., McCormick, N., and Brannigan, G. G. (1980). Living the sexual script: College students' views of influence in sexual encounters. The Journal of Sex Research, 16, 338-355.
- Largen, M. A. (1988). Rape law reform. In A. W. Burgess (Ed.), Rape and sexual assault, Vol. II (pp. 271-292). New York, NY: Garland.
- L'Armand, K., and Pepitone, A. (1982). Judgments of rape: A study of victim-rapist relationship and victim sexual history. <u>Personality and Social Psychology Bulletin</u>, 8, 134-139.
- Larrabee, M. J., and McGeorge, S. A. (1989). Date rape: Understanding the acquaintance assault phenomenon. <u>Counseling and Human Development</u>. 22, 1-16.
- Latourette, A. W., and King, R. D. (1988). Judicial intervention in the student-university relationship: Due process and contract theories.

  <u>University of Detroit Law Review</u>, 65, 199-258.
- Law Enforcement Assistance Administration. (1972). San Jose methods test of known crime victims: Statistics technical report No. 1. Washington, DC: Law Enforcement Assistance Administration, National Institute of Law Enforcement and Criminal Justice Statistics.
- Law Enforcement Assistance Administration. (1974). Crime in the nation's five largest cities: Advance report. Washington, DC: Law Enforcement Assistance Administration, National Criminal Justice Information and Statistics Service.
- Lerner, M. J. (1965). Evaluation of performance as a function of performer's reward and attractiveness. <u>Journal of Personality and Social Psychology</u>, 1, 355-360.
- Lerner, M. J. (1970). The desire for justice and reactions to victims. In J. Macaulay and L. Berkowitz (Eds.), <u>Altruism and helping behavior</u> (pp. 205-229). New York, NY: Academic Press.
- Lerner, M. J. (1980). The belief in a just world: A fundamental delusion. New York, NY: Plenum.
- Lewis-Beck, M. S. (1990). <u>Applied regression: An introduction</u>. Newbury Park, CA: Sage.
- Lindemann, B. S. (1984). To ravish and carnally know: Rape in eighteenth-century Massachusetts. <u>Signs: Journal of Women In Culture and Society</u>, <u>10</u>, 63-82.
- Loh, W. D. (1981). Q: What has reform of rape legislation wrought? A: Truth in criminal labelling. <u>Journal of Social Issues</u>, <u>37</u>, 28-52.
- Long, N. T. (1985). The standard of proof in student disciplinary cases. The Journal of College and University Law, 12, 71-81.
- Lott, B., Reilly, M. E., and Howard, D. R. (1982). Sexual assault and harassment: A campus community case study. <u>Signs: Journal of Women in Culture and Society</u>, <u>8</u>, 296-319.
- Lunnenborg, P. W. (1970). Stereotypic aspect in masculinity-femininity measurement. <u>Journal of Consulting and Clinical Psychology</u>, <u>34</u>, 113-118.
- MacKinnon, C. A. (1983). Feminism, marxism, method, and the state:
  Toward feminist jurisprudence. Signs: Journal of Women in Culture and
  Society, 8, 635-658.

<u> 4.</u> 

- Makepeace, J. M. (1981). Courtship violence among college students.

  Family Relations: Journal of Applied Family and Child Studies, 30, 97102.
- Malamuth, N. M. (1981). Rape proclivity among males. <u>Journal of Social</u> <u>Issues</u>, <u>37</u>, 138-157.
- Malamuth, N. M., and Check, J. V. P. (1980). Penile tumescence and perceptual responses to rape as a function of victim's perceived reactions. <u>Journal of Applied Social Psychology</u>, <u>10</u>, 528-547.
- Malamuth, N. M., Haber, S., and Feshbach, S. (1980). Testing hypotheses regarding rape: Exposure to sexual violence, sex differences, and the normality of rapists. <u>Journal of Research in Personality</u>, <u>14</u>, 121-137.
- Marsh, J. C., Geist, A., and Caplan, N. (1982). Rape and the limits of law reform. Boston, MA: Auburn House.
- Mazelan, P. M. (1980). Stereotypes and perceptions of the victims of rape. <u>Victimology: An International Journal</u>, <u>5</u>, 121-132.
- McCahill, T. W., Meyer, L. C., and Fischman, A. M. (1979). The aftermath of rape. Lexington, MA: Lexington Books.
- McCauley, C., Stitt, C. L., and Segal, M. (1980). Stereotyping: From prejudice to prediction. <u>Psychological Bulletin</u>, <u>87</u>, 195-208.
- McCormick, N. (1979). Come-ons and put-offs: Unmarried students' strategies for having and avoiding sexual intercourse. <u>Psychology of Women Quarterly</u>, 4, 194-211.
- McDermott, M. T. (1979). Rape victimization in twenty-six American cities (Publication No. SD-VAD-6). Washington, DC: Law Enforcement Assistance Administration, National Criminal Justice Information and Statistics Service.
- McDonald, K. (1988, January 13). Male students who commit date rape found to have callous sexual attitudes. The Chronicle of Higher Education, p. A35.
- McGraw, T. M. (1987). <u>Attribution of responsibility for rape as a function of victims' and observers' characteristics</u>. Unpublished master's thesis, Angelo State University, San Angelo, Texas.
- McKee, J. P., and Sherriffs, A. C. (1957). The differential evaluation of males and females. <u>Journal of Personality</u>, <u>25</u>, 356-371.
- Medea, A., and Thompson, K. (1974). Against rape. New York, NY: Farrar, Straus, and Giroux.
- Metzger, D. (1976). It is always the woman who is raped. The American Journal of Psychiatry, 133, 405-408.
- Michigan Campus Sexual Assault Victims' Bill of Rights Act. (1992). House Bill No. 5624.
- Michigan Public Acts No. 266. (1974). Michigan Compiled Laws Annotated, 750.316-759.End, (1991).

·

;

.

.

:

**C**.

å

- Miller, B. (1988). Date rape: Time for a new look at prevention. <u>Journal of College Student Development</u>, 29, 553-555.
- Miller, B., and Marshall, J. C. (1987). Coercive sex on the university campus. <u>Journal of College Student Personnel</u>, <u>28</u>, 38-47.
- Miller, D. T., and McFarland, C. (1987). Pluralistic ignorance: When similarity is interpreted as dissimilarity. <u>Journal of Personality and Social Psychology</u>, <u>53</u>, 298-305.
- Miller, F. D., Smith, B. R., Ferree, M. M., and Taylor, S. B. (1976). Predicting perceptions of victimization. <u>Journal of Applied Social Psychology</u>, 6, 352-359.
- Morrison, C. H. (1980). A cultural perspective on rape. In S. L. McCombie (Ed.), The rape crisis intervention handbook: A guide for victim care (pp. 3-16). New York, NY: Plenum.
- Mosher, D. L., and Anderson, R. D. (1986). Macho personality, sexual aggression, and reactions to guided imagery of realistic rape.

  <u>Journal of Research in Personality</u>, 20, 77-94.
- Muchlenhard, C. L. (1988). Misinterpreted dating behavior and the risk of date rape. <u>Journal of Social and Clinical Psychology</u>, 6, 20-37.
- Muehlenhard, C. L., and Cook, S. W. (1988). Men's self-reports of unwanted sexual activity. The Journal of Sex Research, 24, 58-72.
- Muchlenhard, C. L., Friedman, D. E., and Thomas, C. M. (1985). Is date rape justifiable? <u>Psychology of Women Quarterly</u>, 9, 297-310.
- Muehlenhard, C. L., and Hollabaugh, L. C. (1988). Do women sometimes say no when they mean yes? The prevalence and correlates of women's token resistance to sex. <u>Journal of Personality and Social Psychology</u>, <u>54</u>, 872-879.
- Muehlenhard, C. L., and Linton, M. A. (1987). Date rape and sexual aggression in dating situations: Incidence and risk factors. <u>Journal of Counseling Psychology</u>, <u>34</u>, 186-196.
- Muehlenhard, C. L., and Miller, E. N. (1988). Traditional and nontraditional men's responses to women's dating initiation. Behavioral Modification, 12, 385-403.
- Muehlenhard, C. L., and Scardino, T. J. (1985). What will he think? Men's impressions of women who initiate dates and achieve academically. <u>Journal of Counseling Psychology</u>, <u>32</u>, 560-569.
- NASPA Forum. (1991). Congressman introduces sexual assault bill. 11, 1, 3.
- NASPA Forum. (1992). Campus safety and security update. 13, 8.
- National Association for Women in Education. (1992). Congress requires campus sexual assault policies. About Women On Campus, 2, 1.
- National Association for Women in Education. (1993). Sexual assault. About Women On Campus, 2, 1-2.
- National Victim Center. (1992). Rape in America: A report to the nation. Arlington, VA: National Victim Center.

á

<u>.</u> -• . 3

- O'Gorman, H. J. (1975). Pluralistic ignorance and white estimates of white support for racial segregation. <u>Public Opinion Quarterly</u>, 39, 313-330.
- Orzek, A. M. (1983). Use of rape crisis center services by a university community. <u>Journal of College Student Personnel</u>, 24, 465-466.
- Pace, D., and Zaugra, J. (1988). Model of a date rape workshop for college campuses. <u>Journal of College Student Development</u>, 29, 371-372.
- Parrot, A. (1991a). Institutional response: How can acquaintance rape be prevented? In A. Parrot and L. Bechhofer (Eds.), <u>Acquaintance rape:</u>
  <u>The hidden crime</u> (pp. 355-367). New York, NY: John Wiley and Sons.
- Parrot, A. (1991c). Medical community response to acquaintance raperecommendations. In A. Parrot and L. Bechhofer (Eds.), <u>Acquaintance rape</u>: The hidden crime (pp. 304-316). New York, NY: John Wiley and Sons.
- Parrot, A. (1991b). Recommendations for college policies and procedures to deal with acquaintance rape. In A. Parrot and L. Bechhofer (Eds.), Acquaintance rape: The hidden crime (pp. 368-380). New York, NY: John Wiley and Sons.
- Pascarella, E. T., and Terenzini, P. T. (1991). How college affects students. San Francisco, CA: Jossey-Bass.
- Pavela, G. (1992a). Should colleges resolve sexual assault cases? Synthesis: Law and Policy in Higher Education, 4, 276-278.
- Pavela, G. (1992b, July 29). Today's college students need both freedom and structure. The Chronicle of Higher Education, pp. B1-B2.
- Peterson, S. A., and Franzese, B. (1987). Correlates of college men's sexual abuse of women. <u>Journal of College Student Personnel</u>, <u>28</u>, 223-228.
- Phares, E. J., and Wilson, K. G. (1972). Responsibility attribution: Role of outcome severity, situational ambiguity, and internal-external control. <u>Journal of Personality</u>, <u>40</u>, 392-406.
- Picozzi, J. M. (1987). University disciplinary process: What's fair, what's due, and what you don't get. The Yale Law Journal, 96, 2132-2161
- Pineau, L. (1989). Date rape: A feminist analysis. Law and Philosophy, 8, 217-243.
- Polk, K. (1985). Rape reform and criminal justice processing. <u>Crime and Delinquency</u>, <u>31</u>, 191-205.
- Praeger Publishers. (1977). Women's rights and the law: The impact of the ERA on state laws. New York, NY: Praeger.
- Prentice, D. A., and Miller, D. T. (1993). Pluralistic ignorance and alcohol use on campus: Some consequences of misperceiving the social norm. <u>Journal of Personality and Social Psychology</u>, <u>54</u>, 243-256.
- Quackenbush, R. L. (1987). <u>Differences among androgynous, masculine sextyped</u>, and undifferentiated males on dimensions of attitudes toward rape. Unpublished doctoral dissertation, Washington State University, Pullman.

\*

-		
•		
٤		
÷		
-		
•		
•		
•		
÷		

- Rapaport, K., and Burkhart, B. R. (1984). Personality and attitudinal characteristics of sexually coercive college men. <u>Journal of Abnormal Psychology</u>, 93, 216-221.
- Redford, M. E. (1989). Pennsylvania's college and university security information act: The effect of campus security legislation on university liability for campus crime. <u>Dickinson Law Review</u>, <u>94</u>, 179-197.
- Reidhaar, D. L. (1985). The assault on the citadel: Reflections on a quarter century of change in the relationships between the student and the university. The Journal of College and University Law, 12, 343-361.
- Reiff, R. (1979). The invisible victim: The criminal justice system's forgotten responsibility. New York, NY: Basic Books.
- Reskin, B. F., and Visher, C. A. (1986). The impacts of evidence and extralegal factors in jurors' decisions. <u>Law and Society Review</u>, <u>20</u>, 423-438.
- Reynolds, D. E., and Sanders, M. S. (1975). Effect of defendant attractiveness, age, and injury on severity of sentence given by simulated jurors. The Journal of Social Psychology, 96, 149-150.
- Reynolds, M. (1974). Rape as social control. Catalyst, 8, 62-67.
- Richardson, D., and Campbell, J. (1982). The effect of alcohol on the attributions of blame for rape. Personality and Social Psychology Bulletin, 8, 468-476.
- Richardson, D. R., and Hammock, G. S. (1991). Alcohol and acquaintance rape. In A. Parrot and L. Bechhofer (Eds.), <u>Acquaintance rape: The hidden crime</u> (pp. 83-95). New York, NY: John Wiley and Sons.
- Richmond, A. E. (1980). Rape law and the judicial process. In S. L. McCombie (Ed.), The rape crisis intervention handbook: A guide for victim care (pp. 79-95). New York, NY: Plenum.
- Roark, M. L. (1989). Sexual violence. In J. M. Sherrill and D. G. Siegel (Eds.), <u>Responding to violence on campus</u> (pp. 41-52). San Francisco, CA: Jossey-Bass.
- Robin, G. D. (1977). Forcible rape: Institutional sexism in the criminal justice system. <u>Crime and Delinquency</u>, 23, 136-153.
- Rosenkrantz, P., Vogel. S., Bee, H., Broverman, I., and Broverman, D. M. (1968). Sex-role stereotypes and self-concepts in college students.

  <u>Journal of Consulting and Clinical Psychology</u>, 32, 287-295.
- Russell, D. B. H. (1975). The politics of rape: The victim's perspective. New York, NY: Stein and Day.
- Russell, D. E. H. (1982). The prevalence and incidence of forcible rape and attempted rape of females. <u>Victimology: An International Journal</u>, 7, 81-93
- Russell, D. E. H., and Howell, N. (1983). The prevalence of rape in the United States revisted. Signs: Journal of Women in Culture and Society, 8, 688-695.
- VSanday, P. R. (1981). The socio-cultural context of rape: A cross-cultural study. <u>Journal of Social Issues</u>, <u>37</u>, 5-27.

- Sandberg, G., Jackson, T. L., and Petretic-Jackson, P. (1987). College students' attitudes regarding sexual coercion and aggression:

  Developing educational and preventive strategies. <u>Journal of College Student Personnel</u>, 28, 302-311.
- Scheaffer, R. L., Mendenhall, W., and Ott, L. (1990). <u>Elementary survey</u> sampling. Boston, MA: PWS-Kent.
- Schultz, L. G., and DeSavage, J. (1975). Rape and rape attitudes on a college campus. In L. G. Schultz (Ed.), Rape victimology (pp. 77-90). Springfield, IL: Charles C. Thomas.
- Schwendinger, J. R., and Schwendinger, H. (1974). Rape myths: In legal, theoretical, and everyday practice. <u>Crime and Social Justice</u>, <u>1</u>, 18-26.
- Scroggs, J. R. (1976). Penalties for rape as a function of victim provocativeness, damage, and resistance. <u>Journal of Applied Social Psychology</u>, 6, 360-368.
- Scully, D., and Marolla, J. (1984). Convicted rapists' vocabulary of motive: Excuses and justifications. <u>Social Problems</u>, <u>31</u>, 530-544.
- Searles, P., and Berger, R. J. (1987). The current status of rape reform legislation: An examination of state statutes. <u>Women's Rights Law Reporter</u>, 10, 25-43.
- Seiverd, K. D. (1989). <u>Acquaintance rape: Attribution of blame as a function of respondent's sex, attitudes toward woman and heterosexual relationships, and acceptance of interpersonal violence.</u> Unpublished master's thesis, The University of Arizona, Tucson.
- Selby, J. W., Calhoun, L. G., and Brock, T. A. (1977). Sex differences in the social perception of rape victims. <u>Personality and Social Psychology Bulletin</u>, 3, 412-415.
- Seligman, C., Brickman, J., and Koulack, D. (1977). Rape and physical attractiveness: Assigning responsibility to victims. <u>Journal of Personality</u>, <u>45</u>, 554-563.
- Shaver, K. G. (1970). Defensive attribution: Effects of severity and relevance on the responsibility assigned for an accident. <u>Journal of Personality and Social Psychology</u>, <u>14</u>, 101-113.
- Shaver, K. G. (1975). An introduction to attribution processes. Cambridge, MA: Winthrop.
- Shaw, J. I., and Skolnick, P. (1971). Attribution of responsibility for a happy accident. <u>Journal of Personality and Social Psychology</u>, <u>18</u>, 380-383.
- Sherman, B. (1985, October 23). A new recognition of the realities of date rape. The New York Times, pp. C1, C14.
- Sherriffs, A. C., and Jarrett, R. F. (1953). Sex differences in attitudes about sex differences. <u>The Journal of Psychology</u>, <u>35</u>, 161-168.
- Sherriffs, A. C., and McKee, J. P. (1957). Qualitative aspects of beliefs about men and women. <u>Journal of Personality</u>, <u>25</u>, 451-464.

- Shotland, R. L., and Goodstein, L. (1983). Just because she doesn't want to doesn't mean it's rape: An experimentally based causal model of the perception of rape in a dating situation. <u>Social Psychology Quarterly</u>, 46, 220-232.
- Sigall, H., and Ostrove, N. (1975). Beautiful but dangerous: Effects of offender attractiveness and nature of the crime on juridic judgement. <u>Journal of Personality and Social Psychology</u>, 31, 410-414.
- Simon, R. J. (1980). The jury: Its role in American society. Lexington, MA: Lexington Books.
- Singleton, R., and Christiansen, J. B. (1977). The construct validation of a shortform attitudes toward feminism scale. Sociology and Social Research: An International Journal, 61, 294-303.
- Skelton, C. A., and Burkhart, B. R. (1980). Sexual assault: Determinants of victim disclosure. <u>Criminal Justice and Behavior</u>, 7, 229-236.
- Smith, E. R., Ferree, M. M., and Miller, F. D. (1975). A short scale of attitudes toward feminism. Representative Research in Social Psychology, 6, 51-56.
- Smith, M. C. (1988). Coping with crime on campus. New York, NY: Macmillan.
- Smith, M. J. (1983). <u>Sex role stereotypes and acquaintance rape: A study of rape attitudes</u>. Unpublished doctoral dissertation, University of Arkansas, Fayetteville.
- Smith, R. E., Keating, J. P., Hester, R. K., and Mitchell, H. E. (1976). Role and justice considerations in the attribution of responsibility to a rape victim. <u>Journal of Research in Personality</u>, <u>10</u>, 346-357.
- Snelling, H. A. (1975). What is non-consent (in rape)? In L. G. Schultz
  (Ed.), Rape victimology (pp. 157-163). Springfield, IL: Charles C.
  Thomas.
- Spence, J., and Helmreich, R. (1972). The attitudes toward women scale: An objective instrument to measure attitudes toward the rights and roles of women in contemporary society. <u>JSAS Catalog of Selected Documents in Psychology</u>, 2, 66.
- Squitieri, T. (1992, April 20). Plague of crime more violent. <u>USA Today</u>, p. 1A.
- Stamatakos, T. C. (1990). The doctrine of in loco parentis, tort liability and the student-college relationship. <u>Indiana Law Journal</u>, 65, 471-490.
- Steiner, C. (1989). Pursuit of knowledge or pursuit of suspects: Rights, legalities, liabilities. In J. M. Sherrill and D. G. Siegel (Eds.).

  Responding to violence on campus (pp. 65-75). San Francisco, CA:

  Jossey-Bass.
- Stephan, C. (1974). Sex prejudice in jury simulation. The Journal of Psychology, 88, 305-312.
- Stephan, C. (1975). Selective characteristics of jurors and litigants: Their influences on juries' verdicts. In R. J. Simon (Ed.), <u>The jury system in america: A critical overview</u> (pp. 97-121). Beverly Hills, CA: Sage.

÷

14

•

•

- Stoner, B. N., and Cerminara, K. L. (1990). Harnessing the spirit of insubordination: A model student disciplinary code. The Journal of College and University Law, 17, 89-121.
- Swem, L. L. (1987). Due process rights in student disciplinary matters. The Journal of College and University Law, 14, 359-382.
- Szablewicz, J. J., and Gibbs, A. (1987). Colleges' increasing exposure to liability: The new in loco parentis. <u>Journal of Law and Education</u>, <u>16</u>, 453-465.
- Tetreault, P. A., and Barnett, M. A. (1987). Reactions to stranger and acquaintance rape. <u>Psychology of Women Quarterly</u>, <u>1</u>, 353-358.
- The Nation. (1992, August 26). The Chronicle of Higher Education Almanac, p. 3.
- Thornton, B. (1977). Effect of rape victim's attractiveness in a jury simulation. Personality and Social Psychology Bulletin, 3, 666-669.
- Thornton, B., Robbins, M. A., and Johnson, J. A. (1981). Social perception of the rape victim's culpability: The influence of respondents' personal-environmental causal attribution tendencies. Human Relations, 34, 225-237.
- Thornton, B., Ryckman, R. M., and Robbins, M. A. (1982). The relationships of observer characteristics to beliefs in the causal responsibility of victims of sexual assault. <u>Human Relations</u>, <u>35</u>, 321-330.
- Tieger, T. (1981). Self-reported likelihood of raping and the social perception of rape. <u>Journal of Research in Personality</u>, <u>15</u>, 147-158.
- Tolor, A. (1978). Women's attitudes toward forcible rape. Community Mental Health Journal, 14, 116-122.
- Tong, R. (1984). Women, sex, and the law. Totowa, NJ: Rowman and Allanheld.
- Turney, B., and Robb, G. (1971). Research in education: An introduction. Hinsdale, IL: Dryden.
- Vidmar, N. (1972). Effects of decision alternatives on the verdicts and social perceptions of simulated jurors. <u>Journal of Personality and Social Psychology</u>, <u>22</u>, 211-218.
- <u>Virginia Law Review</u>. (1975). Recent statutory developments in the definition of forcible rape. <u>61</u>, 1500-1543.
- Wagner, E. N. (1991, August 7). Campus victims of date rape should consider civil lawsuits as alternatives to criminal charges or colleges' procedures. The Chronicle of Higher Education, p. B2.
- Walster, E. (1966). Assignment of responsibility for an accident. Journal of Personality and Social Psychology, 3, 73-79.
- Walster, E. (1967). Second-guessing important events. <u>Human Relations</u>, <u>20</u>, 239-250.
- Ward, D., and Balswick, J. (1978). Strong men and virtuous women: A content analysis of sex roles stereotypes. <u>Pacific Sociological Review</u>, <u>21</u>, 45-65.

- Ward, S. K., Chapman, K., Cohn, E., White, S., and Williams, K. (1991). Acquaintance rape and the college social scene. <u>Family Relations</u>: <u>Journal of Applied Family and Child Studies</u>, <u>40</u>, 65-71.
- Warshaw, R. (1988). <u>I never called it rape</u>. New York, NY: Harper and Row.
- Weis, K., and Borges, S. S. (1973). Victimology and rape: The case of the legitimate victim. <u>Issues in Criminology</u>, 8, 71-115.
- Weld, H. P., and Danzig, E. R. (1968). A study of the way in which a verdict is reached by a jury. In R. J. Simon (Ed.), The sociology of law (pp. 83-93). San Francisco, CA: Chandler.
- Wigmore, J. H. (1935). A students' textbook of the law of evidence. Brooklyn, NY: The Foundation Press.
- Williams, J. E. (1979). Sex role stereotypes, women's liberation and rape: A cross-cultural analysis of attitudes. <u>Sociological Symposium</u>, <u>25</u>, 61-97.
  - Williams, K. M. (1976). The effects of victim characteristics on the disposition of violent crimes. In W. F. McDonald (Ed.), <u>Criminal</u> justice and the victim (pp. 177-213). Beverly Hills, CA: Sage.
  - Williams, L. S. (1984). The classic rape: When do victims report? <u>Social Problems</u>, <u>31</u>, 460-467.
  - Wilson, K., Faison, R., and Britton, G. M. (1983). Cultural aspects of male sex aggression. <u>Deviant Behavior</u>, 4, 241-255.
  - Wilson, P. R. (1978). The other side of rape. St. Lucia, Queensland, Australia: University of Queensland.
  - Wilson, W., and Durrenberger, R. (1882). Comparison of rape and attempted rape victims. <u>Psychological Reports</u>, <u>50</u>, 198.
  - Wishman, S. (1986). Anatomy of a jury. New York, NY: Times Books.
  - Wong, P. T. P., and Weiner, B. (1981). When people ask "why" questions, and the heuristics of attributional search. <u>Journal of Personality and Social Psychology</u>, 40, 650-663.
  - Wood, P. L. (1973). The victim in a forcible rape case: A feminist view. The American Criminal Law Review, 11, 335-354.
  - Wrightsman, L. S., Kassin, S. M., and Willis, C. E. (1987). <u>In the jury box: Controversies in the courtroom</u>. Newbury Park, CA: Sage.
  - <u>Yale Law Journal</u>. (1972). The rape corroboration requirement: Repeal not reform. <u>81</u>, 1365-1391.
  - Yegidis, B. (1986). Date rape and other forced sexual encounters among college students. <u>Journal of Sex Education and Therapy</u>, <u>12</u>, 51-54.

